Genetic studies of disease resistance and bolting time based on genomic analysis in Japanese bunching onion (*Allium fistulosum* L.)

ネギのゲノム解析に基づく病害抵抗性
および抽苔性の遺伝学的研究
ション

Tadayuki Wako

2016

Genetic studies of disease resistance and bolting time based on genomic analysis in Japanese bunching onion (*Allium fistulosum* L.)



by Tadayuki Wako

A thesis submitted as part fulfillment for a PhD in AGRICULTURE

The United Graduate School of Agricultural Sciences Tottori University JAPAN

2016

# **TABLE OF CONTENTS**

	Page
Chapter I: General introduction	1
Chapter II: Screening and incorporation of rust resistance from	<i>Allium cepa</i> into
bunching onion (Allium fistulosum) via alien chromo	some addition
Introduction	
Materials and methods	6
Results	9
Discussion	13
Chapter III: Mapping of quantitative trait loci for rust resistance	in bunching onion
Introduction	19
Materials and methods	
Results	22
Discussion	24
Chapter IV: Mapping of quantitative trait loci for bolting time in	bunching onion
Introduction	
Materials and methods	
Results	
Discussion	51
Chapter V: Construction of an Allium cepa linkage map using	g doubled haploid
technology	
Introduction	57
Materials and methods	
Results	70
Discussion	73
Chapter VI: General discussion	
Acknowledgements	
Literature cited	
Summary	103
Japanese summary	106
List of papers related to the thesis	

#### **Chapter I: General introduction**

Cultivated and wild species of alliums belong to the genus *Allium* in the Allioideae subfamily of the Amaryllidaceae (The Angiosperm Phylogeny Group, 2009). The genus comprises more than 800 species and includes the cultivated onion, garlic, scallions, shallots, and leeks, as well as chives and hundreds of other wild species (Li et al., 2010). *Allium cepa* L. (2n = 2x = 16), which includes bulb onion (*A. cepa* L. Common onion group, genomes CC) and shallot (*A. cepa* L. Aggregatum group, genomes AA), is among the most important cultivated species in the world. The bulb onion ranks second in value after tomatoes on the list of cultivated vegetable crops worldwide and the annual production of bulb onions is around 85 million tons (FAO, 2013). Although only 1.2% of the world bulb onion crop is produced in Japan, the annual value of this crop is more than a hundred billion yen.

The bunching onion (A. fistulosum L., 2n = 2x = 16, genome FF), also referred to as the Japanese bunching onion or Welsh onion, originated in northwestern China and is mainly cultivated in East Asian countries, in particular in Japan, China, and Korea (Kumazawa and Katsumata, 1965; Ford-Lloyd and Armstrong, 1993). Annual production of bunching onion is one of the highest among fruit and vegetables grown in Japan, following production of tomato, strawberry, and cucumber (Ministry of Agriculture, Forestry and Fisheries, 2014). Many local varieties are adapted to various climatic conditions, and are classified into four groups: 'Kaga', 'Senju', 'Kujo', and 'Yagura-Negi', according to morphological and agronomic adaptation traits (Inden and Asahira, 1990). Traits that are major targets for improvement in bunching onion are disease resistance, yield, late bolting, improved consumption qualities (e.g., low pungency or high sugar content), and suitability for mechanized farming, although the genetic system of these traits is unclear. The breeding methods for bunching onions are relatively unsophisticated in comparison to those used for other major crops. Bunching onion is a typical allogamous crop that arose through protandry, resulting in a high rate of outcrossing (Ford-Lloyd and Armstrong, 1993). It is self-compatible, but suffers from severe in breeding depression when self-pollinated. Open-pollinated varieties of bunching

onion exhibit very high degrees of genetic heterogeneity (Haishima et al., 1993) and even parental lines of F<sub>1</sub> hybrids maintain a certain degree of genetic heterogeneity (Tsukazaki et al., 2006).

Many hybrid cultivars of bunching onion adapted to different climatic conditions and growing seasons have been developed by private breeding companies. However, there are sometimes serious impediments to their cultivation, such as various diseases and insect pests. For example, rust caused by *Puccinia allii* (DC.) F. Rudolphi is a fungal foliar disease in bunching onions that causes substantial deterioration of quality and yield. Uredospores released from uredinia by the pathogen are the main inocula for secondary infections throughout the year. Epiphytotics of this disease occur frequently in relatively cool and moist seasons (mainly in the autumn and spring) (Takeuchi, 1990). Because no commercial bunching onion cultivar possesses sufficient rust resistance, large amounts of fungicides are used to control this disease. It is difficult to apply fungicides at the time of disease outbreak, because the growing period of bunching onion is very long. Also, the overuse of fungicides may have adverse effects on the environment and human health, in addition to a financial burden on farmers. Thus, breeding for rust resistance would be an effective, economical, and ecologically compatible measure to prevent epiphytotics.

*A. cepa* was reported to be less susceptible to *P. allii* (Morinaka, 1985; Jennings et al., 1990a). Although *A. fistulosum* and *A. cepa* are in the section *Cepa* of the genus *Allium* (Hanelt, 1990), the genetic basis of resistance to rust in *A. cepa* remains unknown. Introgression of resistance genes from related species can be effective for development of novel cultivars. Monosomic alien addition lines (MAALs) are invaluable materials for localizing genes that control traits of interest, as an entire donor genome is subdivided into individual extra chromosomes added into the genetic background of a recipient species. A complete set of bunching onion – shallot MAALs (2n = 2x + 1 = 17, FF+1A through FF+8A) were developed (Shigyo et al., 1996) for mapping in *A. cepa* (Shigyo et al., 1997; Masuzaki et al., 2006b; 2006c; Yaguchi et al., 2009; Vu et al., 2012) and will help to determine which shallot chromosomes carry rust resistance.

A recurrent selection program for rust resistance in bunching onion was previously conducted, and a rust-resistant parental line 'Negi Chuukanbohon Nou 1' was developed (Yamashita et al., 2005; Wako et al., 2012). This line shows a high level of field resistance to rust, so molecular markers linked to genes controlling rust resistance are needed for efficient selection. Because many agricultural traits are probably controlled by quantitative trait loci (QTLs), breeders seek DNA markers closely linked to QTLs to use for marker-assisted selection. Approximately 2000 genomic simple sequence repeats (gSSRs) have been identified in the bunching onion genome (Song et al., 2004; Tsukazaki et al., 2007) and genetic linkage maps have been constructed using these SSR markers (Ohara et al., 2005a; Tsukazaki et al., 2008; 2011; 2012). A major QTL for pungency was identified using the population SaT03, which was derived from a cross between a rust-resistant line and a susceptible line (Tsukazaki et al., 2012). This mapping population will be useful for investigating QTL for rust resistance.

Bolting, the premature formation of flowering stems before harvest, has a severe impact on the yield and quality of bunching onion products during spring and early summer because the flower stalks are so tough that they are inedible. Therefore, the late-bolting trait is indispensable in cultivars for spring and summer production. Bunching onion, which is biennial like bulb onion, vernalizes for flower induction in response to low temperature and short photoperiod (Brewster, 2008). Genetic variation in bolting time in bunching onion is based on seedling age and low-temperature requirements (Inden and Asahira, 1990). However, other genetic studies of flowering are very limited in bunching onion, so it will be important to investigate QTL for bolting time as a first step toward understanding the genetics of flower initiation.

In *Allium* vegetable crops, molecular genetic studies are less advanced than in other major crops. RAPD (random amplified polymorphic DNA) and AFLP (amplified fragment length polymorphism) analyses have also been performed in *Allium* (Wilkie et al., 1993; Bradeen and Havey, 1995; van Heusden et al., 2000a; 200b; Ohara et al., 2005a; 2005b; Ipek et al., 2005). King et al. (1998) reported a linkage map of bulb onion based on RAPDs and RFLPs (restriction fragment length polymorphisms). Fischer and Bachmann (2000) reported the first development of 30 SSR markers in bulb onion. In contrast, hundreds of expressed sequence tag (EST)-derived SSRs, EST-derived single nucleotide polymorphisms (SNPs), and insertion–deletion (InDel) markers have been

developed from medium-scale sequencing of ESTs in bulb onion (Kuhl et al., 2004; Martin et al., 2005). Some of these markers have been effectively used to construct linkage maps not only in bulb onion (Martin et al., 2005; McCallum et al., 2006; 2007; 2012), but also in bunching onion (Tsukazaki et al., 2008; 2011). Recently, transcriptome sequencing using next-generation sequencing (NGS) technology has been conducted and used to detect single nucleotide polymorphisms (SNPs) and insertion–deletions (InDels) between cultivars or individual plant lines, thereby facilitating the construction of highresolution genetic maps (Baldwin et al., 2012; Duangjit et al., 2013). As for bunching onion, more than 50,000 unigenes were recently obtained from transcriptome shotgun assembly of next-generation sequencing data, and numerous DNA markers based on SSRs, SNPs and InDels have been developed (Tsukazaki et al., 2015). However, comparative mapping between *A. fistulosum* and *A. cepa* has not yet been conducted because the number of markers common to both species is limited. Development of a linkage map with markers common to both bunching onion and bulb onion will facilitate comparative mapping and analysis of orthologous genes in both species.

The present studies were conducted with the following objectives: 1) to investigate the susceptibility of *A. fistulosum* and *A. cepa* cultivars to rust and identify the *A. cepa* chromosome(s) related to rust resistance, 2) to identify QTL for rust resistance in *A. fistulosum*, 3) to identify QTL for bolting time in *A. fistulosum*, and 4) to construct an *A. cepa* linkage map using a DH population.

This dissertation is composed of six chapters. Chapter 2 addresses Objective 1). Chapter 3 discusses Objective 2). Chapter 4 refers to Objective 3). Chapter 5 focuses on Objective 4). And finally, Chapter 6 provides a general discussion. This dissertation compiles the results of studies conducted by the author at the Laboratory of Vegetable Crop Science, Division of Agrobiology, Department of Biological and Environmental Science, Faculty of Agriculture, Yamaguchi University and at the Vegetable Breeding and Genome Division, Institute of Vegetable and Tea Science, National Agriculture and Food Research Organization, Japan, with the above-mentioned objectives from 2007 to 2016 (Wako et al., 2015; 2016).

# Chapter II: Screening and incorporation of rust resistance from *Allium cepa* into bunching onion (*Allium fistulosum*) via alien chromosome addition

#### Introduction

Rust caused by *Puccinia allii* (DC.) F. Rudolphi is a serious foliar disease in bunching onions and causes substantial deterioration of quality and yield. Because no current commercial bunching onion cultivar possesses sufficient rust resistance, large amounts of fungicides are used annually to prevent and control this disease. Recurrent selection for rust resistance in bunching onion has been previously conducted and a parental line with rust resistance has been successfully developed (Yamashita et al., 2005; Wako et al., 2012). Although this line shows a moderate level of field resistance to rust, it does not completely control the disease. Therefore, it would be desirable to identify stronger sources of rust resistance that could be incorporated into cultivars.

Also, information regarding the pathogenicity of a range of *P. allii* isolates on *Allium* germplasm is limited. Jennings et al. (1990a) reported on screening experiments in the UK in which *A. cepa* showed higher resistance than *A. fistulosum* or *A. ampeloprasum* ssp. *porrum* (leek) after inoculation with *P. allii* isolates from leek. *A. cepa* is less susceptible to *P. allii* isolated from *A. fistulosum* in Japan (Morinaka, 1985). *A. fistulosum* and *A. cepa* have been classified into the section *Cepa* in the genus *Allium* (Hanelt, 1990), but the genetic basis of resistance to rust in *A. cepa* remains unknown. Introgression of resistance genes from related species can be effective for the development of novel resistant cultivars.

MAALs contain an extra chromosome from a related species. A complete set of MAALs is invaluable material for facilitating localization of genes responsible for traits of interest, as an entire donor genome is subdivided into individual extra chromosomes added to the genetic background of a recipient species. Shigyo et al. (1996) developed a complete set of bunching onion – shallot (*A. cepa* Aggregatum group) MAALs (2n = 2x + 1 = 17, FF+1A through FF+8A). Each MAAL thus displays distinct phenotypic and physiological characteristics (Shigyo et al., 1997), and can be used to identify which shallot chromosomes are responsible for its rust resistance. Using MAALs, Vu et al.

(2012) demonstrated that shallot chromosome 2A carried genes for *Fusarium* wilt resistance as well as others encoding enzymes that produce antifungal metabolites against *F. oxysporum*. These complete addition lines have also been effectively used to determine the chromosomal locations of genes involved in flavonoid biosynthesis (Masuzaki et al., 2006b; 2006c), sulfur assimilation (McCallum et al., 2007), and sucrose metabolism (Yaguchi et al., 2008).

The objectives of the present study were to investigate the susceptibility of *A*. *fistulosum* and *A. cepa* cultivars to rust isolates from bunching onion, and to identify which *A. cepa* chromosome is responsible for rust resistance using alien chromosome addition lines.

## Materials and methods

#### Plant materials and P. allii isolate

Ten cultivars of bunching onion, four cultivars of bulb onion (described in Table 1), shallot 'Chiang Mai' from Thailand, and selfed progenies of a complete set of bunching onion - shallot MAALs (Shigyo et al., 1996) were used for seedling tests. Seedlings from each MAAL were screened for shallot-derived chromosomes using isozyme or Sequenced Characterized Amplified Region (SCAR) markers. Chromosomes 1A, 2A, and 6A were identified using the Lap-1, Got-1, and Got-2 isozymes, respectively (Shigyo et al., 1994; 1995). Chromosomes 3A, 4A, 5A, 7A, and 8A were identified using the SCAR markers API40, Allinase, AJK265, 5S rDNA, and API73-2, respectively (van Heusden et al., 2000b). The addition of chromosomes in these lines was confirmed using microscopic observation or flow cytometry (Shigyo et al., 2003). One hundred seeds from selfed progenies of each MAAL were sown, and more than eight plants carring each monosomic addition were identified. For tests using adult plants, the MAALs, multichromosome addition lines, and hypoallotriploids (2n = 2x + 2 - 7 = 18 - 23) (Hang et al., 2004; Yaguchi et al., 2008) were grown in pots for over one year. Uredospore inoculum of P. allii was collected from a field in Seiro, Niigata Prefecture, Japan, multiplied on bunching onion plants, and then stored at -30 °C until just before use. Inoculum for use

Cultivar	Species	Origin	No. of leaf le	urec	linia pe *	r cm
Senshu-chuko-ki	А. сера	Takii Seed Co. Ltd., Kyoto, Japan	0.10	±	0.02	а
Kaiduka-wase	А. сера	Sakata Seed Co., Yokohama, Japan	0.12	±	0.05	а
Imai-wase	А. сера	Sakata Seed Co., Yokohama, Japan	0.21	±	0.04	а
Shonan-red	А. сера	Sakata Seed Co., Yokohama, Japan	0.24	±	0.10	а
Shimonita	A. fistulosum	Kaneko Seed Co., Maebashi, Japan	1.01	±	0.15	b
Hikawa	A. fistulosum	Nihon Norin Seed Co., Tokyo, Japan	1.16	±	0.15	bc
Kasho-ipponfuto	A. fistulosum	Sakata Seed Co., Yokohama, Japan	1.24	±	0.16	bc
Shimotae	A. fistulosum	Mikado Kyowa Seed Co. Ltd., Chiba, Japan	1.27	±	0.15	bc
Choetsu	A. fistulosum	Mikado Kyowa Seed Co. Ltd., Chiba, Japan	1.27	±	0.16	bc
Choju	A. fistulosum	Mikado Kyowa Seed Co. Ltd., Chiba, Japan	1.47	±	0.13	bc
Kincho	A. fistulosum	Mikado Kyowa Seed Co. Ltd., Chiba, Japan	1.52	±	0.17	bc
Yoshikura	A. fistulosum	Musashino Seed Co. Ltd., Tokyo, Japan	1.53	±	0.21	bc
Kujo-futo	A. fistulosum	Takii Seed Co. Ltd., Kyoto, Japan	1.80	±	0.23	c
Ishikura-nebuka-fuvufuto	A. fistulosum	Kaneko Seed Co., Maebashi, Japan	1.82	±	0.21	с

**Table 1.** Number of uredinia on leaves of *Allium cepa* and *A. fistulosum* cultivars at 14 days after inoculation with *Puccinia allii*.

Instructure-rugurutoA. JistutosumKaneko Seed Co., Maebashi, Japan $1.82 \pm 0.21$ c\*Mean  $\pm$  SE (n = 20), Significant differencedetected using Tukey-Kramer HSD test (P = 0.05). Values followed by<br/>the same letter are not significantly different.

on multi-chromosome addition lines and hypoallotriploids was obtained from uredinia on bunching onion leaves collected from a field located in Sodegaura, Chiba Prefecture, Japan.

## **Inoculation of seedlings**

Seedling tests were conducted to compare rust resistance among cultivars of A. fistulosum, A. cepa and the MAALs. Seeds for bunching onion, bulb onion, shallot, and MAALs were sown into plastic pots (6 cm in diameter) and grown in a greenhouse for approximately 90 days until leaves were 30 cm in length. In our preliminary tests, seedlings less than 60 days old were not suitable because their leaves senesced easily after inoculation. To inoculate each plant, a mixture of 1 mL of  $5 \times 10^4$  uredospore suspension with 0.1 % polyoxyethylene sorbitan monolaurate (Tween<sup>®</sup> 20, Nacalai Tesque, Inc., Japan) and 0.5 % talc (Nacalai Tesque, Inc., Japan) was sprayed onto the leaves. Inoculated plants were placed in a growth chamber (KPSH-30, Ozawa Seisakusho Co. Ltd., Japan) maintaining a temperature of 18 °C and a 12 h light/dark cycle. To maintain damp conditions for the germination of uredospores, the atmosphere was kept saturated with water using an ultrasonic humidifier (FT-10N-14, UCAN Co. Ltd., Japan) for 24 h after inoculation. After incubation for 14 days, the number of uredinia produced on the youngest full-grown leaf of each plant was counted. For inoculated seedlings from the MAALs and inoculated adult plants from the multi-chromosome addition lines and hypoallotriploids, the quantities of uredinia were scored using a 0-3 scale as follows: 0, no uredinia pustule; 1, 1–20 uredinia per leaf (slight symptoms); 2, 21–100 uredinia per leaf (moderate symptoms); 3, more than 101 uredinia per leaf (severe symptoms).

#### **Inoculation of adult MAAL plants**

Inoculation tests were conducted in the field to examine the rust resistance of fullgrown shallot and MAAL plants. Plants of approximately 30 cm in height that had been grown in plastic pots were transplanted into plots in a plastic greenhouse in a randomized block design with three replicates and were grown for 2wo months. Before inoculation, 0.006% polyoxyethylene nonylphenyl ether or KUMITEN (Kumiai Chemical Industry Co., Ltd., Japan) was sprayed onto plants to stimulate germination of the subsequently inoculated spores. Each plant was sprayed with a mixture of 1 mL  $5 \times 10^3$  uredospore suspension with 0.01% Tween 20 and 0.5 % talc. Beginning 14 days after inoculation, the severity of rust symptoms was scored on full-grown leaves of each plant at 2-week intervals. Plant responses to inoculation were ranked using a 0–5 scale as follows: 0, no uredinia; 1, 1–3 uredinia per leaf; 2, 4–10 uredinia per leaf; 3, more than 11 uredinia per leaf; 4, uredinia distributed over the entire leaf; and 5, uredinia distributed densely over the entire leaf. To evaluate disease intensity, the area under the disease progress curve (AUDPC) was calculated according to Shaner and Finney (1977), using the following equation:

AUDPC = 
$$\sum_{i=1}^{Ni-1} (t_{(i+1)} - t_i) (DS_{(i+1)} + DS_i)/2$$

where Ni = number of observations;  $t_i$  = days at the i<sup>th</sup> observation ( $t_0$  = 0); and  $DS_i$  = rust severity at the i<sup>th</sup> observation ( $DS_0$  = 0). After inoculation, the temperature inside the plastic greenhouse ranged from 10 to 23 °C.

# Inoculation of adult plants from multi-chromosome addition lines and hypoallotriploid lines

A total of 34 individuals from multi-chromosome addition lines and hypoallotriploids possessing between 1 and 7 shallot chromosomes were used. A solution of 0.1% Tween 20 was sprayed onto the plants to completely remove wax layer of leaves at 1 day before inoculation. Each plant was sprayed with a mixture of 5 mL of  $1 \times 10^5$  uredospore suspension with 0.01% Tween 20 and placed in a moist chamber under high humidity for 48 h. Inoculated plants were then incubated in a greenhouse regulated at 18–20°C, and rust severity was scored 14 days after inoculation. The scales for rust severity were the same as those used for seedling tests. Inoculation experiments were replicated twice in different years.

#### Results

## Comparison of resistance between A. fistulosum and A. cepa cultivars

In both *A. fistulosum* and *A. cepa*, small white spots first appeared on each leaf 8–9 days after inoculation. After 2–3 days, the spots developed into orange pustules (i.e., uredinia). The mean number of uredinia per cm of leaf length at 14 days after inoculation differed between *A. fistulosum* and *A. cepa*. Numbers of uredinia on plants of all of the tested *A. fistulosum* cultivars were significantly greater than in *A. cepa* (Table 1). These data shows that *A. cepa* is highly resistant to rust isolated from *A. fistulosum*. Significant differences in uredinia numbers among *A. fistulosum* cultivars were observed. The cultivars 'Kujo-futo' and 'Ishikura-nebuka-fuyufuto' were highly susceptible to rust.

#### Rust resistance of shallot and bunching onion – shallot MAALs in seedling tests

Similar to observations on bunching onions and bulb onions, small white spots first appeared on the leaves of shallot and MAAL plants at 8–9 days after inoculation. The number of white spots on shallot leaves was greater than that on bulb onion leaves. Most of the spots remained white and few developed into orange uredinia (Fig. 1). By 14 days after inoculation, no uredinia were produced (scored as 0 on the rust symptom scale) on nearly 40% of shallot plants and the number of uredinia per leaf was fewer than 20 (scored as 1) on 90% of plants (Table 2). In contrast, more than 101 uredinia developed per leaf in 70% of *A. fistulosum* 'Kujo-hoso' plants (scored as 3). Within a complete set of *A. fistulosum* - shallot MAALs, half of the FF+1A plants were scored as 1 on the rust symptom scale and the remaining half of the plants were scored as 2 or 3. The mean rust symptom severity for FF+1A (1.6) was significantly lower than that of FF (2.7). In FF+1A plants, either white spots did not develop into uredinia, or pale green haloes appeared around the few uredinia that did develop. All plants from the lines FF+2A through FF+8A were scored as 2 or 3 on the rust symptom severity scale, with rust severities similar to those of FF.

# Evaluation of adult plant resistance of bunching onion – shallot MAALs in the field test

In *A. fistulosum* 'Yoshikura', a few uredinia were observed 14 days after inoculation, and rust symptoms increased with days after inoculation. Uredospores released from



**Fig. 1.** Symptoms on leaves of *Allium fistulosum*, shallot, and FF+1A seedlings 14 days after inoculation with *Puccinia allii*. A, *A. fistulosum* 'Kujo-hoso'; B, shallot 'Chiang Mai'; C, FF+1A. Scale bar = 1 cm.

Cultiver or line	No. of	Frequency distribution of rust severity <sup>a</sup>					
	plants	0	1	2	3	Mean	
FF (Kujo-hoso)	30			9	21	2.7	
AA (Chiang Mai)	47	18	24	5		0.7	**
FF+1A	20		10	8	2	1.6	**
FF+2A	8			5	3	2.4	
FF+3A	27			5	22	2.8	
FF+4A	12			7	5	2.4	
FF+5A	8			2	6	2.8	
FF+6A	10			2	8	2.8	
FF+7A	19			1	18	2.9	
FF+8A	14			3	11	2.8	

**Table 2.** Rust severity scores on *Allium fistulosum* (FF), shallot (AA), and eight *A. fistulosum* – shallot MAALs (FF+nA) at 14 days after inoculation with *Puccinia allii*.

<sup>a</sup> Rust severities were ranked as follows: 0, no uredinia; 1, 1–20 uredinia per leaf; 2, 21–100 uredinia per leaf; 3,more than 101 uredinia per leaf.

\*\* Significant differences from FF were detected using Wilcoxon rank-sum test (P = 0.01)

uredinia appeared to infect newly expanded leaves, so rust severity reached scores 4 to 5 at 6 weeks after inoculation. Significantly fewer uredinia were observed developing on shallots (Fig. 2). The AUDPCs of FF+1A, FF+4A, FF+5A, FF+6A, FF+7A, and FF+8A were similar to that of *A. fistulosum*, suggesting these MAALs were all susceptible to rust. The AUDPCs of FF+2A and FF+3A were relatively lower than those of *A. fistulosum* and other MAALs, however the differences between them were not significant.

# Evaluation of adult plant rust resistance in bunching onion – shallot multichromosome addition lines

Shallot was scored as 1 on the rust severity scale (< 20 uredinia per leaf) in each replicate (Table 3). The rust severity scores of plants from multi-chromosome addition lines and hypoallotriploids were generally higher than those of shallot, regardless of the particular additional chromosomes. In the first experiment, nine plants with shallot chromosome addition were scored as 1 on the rust severity scale. However, in the second experiment, only one plant possessing all of the shallot chromosomes except for 4A, and 2 plants possessing 1A and 5A were scored as 1 on the disease severity scale. Although two FF+1A+5A individuals showed low disease severity comparable to that of shallot in both experiments, some individuals from addition lines containing shallot chromosomes 1A and 5A exhibited high disease severity.

#### Discussion

In the present study, *A. cepa* cultivars including bulb onions and shallots showed high resistance at different ages to a *P. allii* isolate derived from *A. fistulosum*. These results were consistent with those of previous studies (Morinaka, 1985; Jennings et al., 1990a). Morinaka (1985) investigated the pathogenicity to *Allium* crops of five isolates of *P. allii* collected from *A. fistulosum* grown in different regions in Japan. All of the isolates showed no pathogenicity to Chinese chive (*A. ramosum*) or Japanese scallion (*A. chinense*). These isolates produced white flecks on leaves only on bulb onion, leeks, and chives (*A. schoenoprasum*), and some isolates did not form sporulating uredinia. Jennings et al. (1990a) reported that *P. allii* isolates derived from *A. porrum* (leeks) produced pale



**Fig. 2.** Area under the disease progress curves (AUDPCs) for *Allium fistulosum* (FF), shallot (AA), and eight *A. fistulosum* – shallot MAALs (FF + nA) 6 weeks after inoculation with *Puccinia allii.* \*\* indicates significant difference from *A. fistulosum* detected by Student's *t* test (*P* = 0.01). Bars indicate ± SE (*n* = 3).

No. of	Chromosomal composition						Rust severity score <sup>a</sup>				
chromosomes				losoin		iposit	1011			1 <sup>st</sup> ex.	2 <sup>nd</sup> ex.
16	FF (Ku	jo-hoso	o)							1-2 <sup>b</sup>	2 <sup>b</sup>
16	AA (Ch	iang N	1ai)							1	1
23	FF +	1A	2A	3A		5A	6A	7A	8A	$1 - 2^{b}$	1-2 <sup>b</sup>
23	FF +	1A	2A	3A	4A	5A	6A		8A	3	3
22	FF +	1A	2A		4A		6A	7A	8A	1	2
22	FF +	1A	2A	3A	4A		6A	7A		1	2
22	FF +	1A		3A		5A	6A	7A	8A	1	2
22	FF +	1A	2A	3A		5A	6A	7A		2	2
22	FF +	1A	2A		4A	5A	6A	7A		3	2
22	FF +	1A	2A	3A	4A	5A	6A			3	2
22	FF +	1A	2A	3A		5A	6A		8A	3	3
22	FF +	1A	2A	3A		5A	6A	7A		3	3
21	FF +	1A	2A		4A	5A	6A			2	3
21	FF +	1A	2A	3A	4A	5A				2	3
21	FF +	1A	2A	3A				7A	8A	2	3
21	FF +	1A	2A	3A		5A	6A			2	2
21	FF +	1A	2A	3A		5A			8A	2	3
21	FF +	1A	2A			5A	6A		8A	3	$NT^d$
20	FF +	1A	2A			5A		7A		1	2
20	FF +	1A			4A		6A		8A	2	2
20	FF +	1A		3A		5A	6A			2	2
20	FF +	1A	2A	3A		5A				3	2
20	FF +	1A		3A	4A		6A			3	$NT^d$
19	FF +	1A				5A			8A	1-2 <sup>b</sup>	2b
19	FF +	1A		3A				7A		3	2
18	FF +	1A				5A				1-2°	1 <sup>b</sup>
18	FF +		2A		4A					2	3
18	FF +	1A			4A					3	3
17	FF +					5A				1	2
17	FF +			3A						1 <sup>b</sup>	2 <sup>b</sup>
17	FF +								8A	2	2

**Table 3.** Rust severity on *Allium fistulosum*, shallot, and *A. fistulosum* – shallot hypoallotriploids at 14 days after inoculation with *Puccinia allii*.

<sup>a</sup> The scales for rust severity scores are noted in the footnote for Table 2. <sup>b</sup> Two plants were tested. <sup>c</sup> Three plants were tested. <sup>d</sup> Not tested.

green flecks and small pustules with haloes on leaves of *A. cepa*, and the quantity of pustules on plants of this species was lower than that on *A. fistulosum*. In our results, white flecks were found on leaves of *A. cepa* after inoculation, and few uredinia formed. Thus, while *P. allii* can presumably infect and colonize *A. cepa*, it does not appear able to form sporulating uredinia (Jennings et al., 1990a). Although the quantitative differences in uredinia formation between *A. cepa* and *A. fistulosum* were significant, variation in uredinia formation within each species was low. In *A. fistulosum*, there were significant differences between cultivars in field resistance of full-grown plants, which indicated that recurrent selection for rust resistance was successful (Yamashita et al., 2005). In seedling tests in the present study, no significant differences were observed among cultivars except for 'Shimonita', 'Kujo-futo', and 'Ishikura-nebuka-fuyufuto'.

Allium ramosum, A. chinense, and A. schoenoprasum have been less useful for breeding rust resistance in bunching onion because they belong to subgenera and sections that are genetically distant from A. fistulosum. However, A. cepa and A. fistulosum, which have been classified into a single section Cepa, are able to cross-pollinate (Emsweller and Jones, 1935). Therefore, A. cepa should be focused as an important genetic resource for breeding in A. fistulosum. No other rust-resistant species within the section Cepa has been reported. Rust resistance genes from A. cepa will likely be introduced into A. fistulosum via introgressive hybridization. A. cepa and A. fistulosum have the same chromosome numbers (2n = 2x = 16) and similar karyotypes (Albini and Jones, 1988). However the DNA content of the A. cepa genome is 28% greater than that of A. fistulosum (Labani and Elkington, 1987), and A. cepa chromosomes are an average of 12% larger at somatic metaphase than those of A. fistulosum (Jones and Rees, 1968). To date, attempts to introgress genes from A. fistulosum into A. cepa by crossing A. cepa and A. fistulosum followed by backcrossing to A. cepa have not been successful, except in only one report (Peffley and Hou, 2000). Sterility in backcrossed generations is thought to be due to an imbalance between the nuclear and cytoplasmic genomes (Ulloa et al., 1995). On the other hand, van der Meer and de Vries (1990) showed that A. roylei (2n = 2x = 16) crosses readily with either A. cepa or A. fistulosum. Khrustaleva and Kik (1998) showed that the three parental genomes in the first generation bridge cross A. cepa  $\times$  (A. fistulosum  $\times$  A.

*roylei*) could be distinguished from each other by means of genomic *in situ* hybridization. Recombination between the three genomes was frequently observed in meiotic anaphase 1 and prophase 2 chromosomes of the first-generation bridge cross and in mitotic chromosomes of the second-generation bridge cross (Khrustaleva and Kik, 2000).

Within a set of A. fistulosum - shallot MAALs, only FF+1A showed significantly lower rust disease symptoms during the seedling stage (3 months after sowing). This result suggests that the gene(s) related to rust resistance are located on chromosome 1A of shallot and could perform in the genetic background of A. fistulosum. Because the resistance level of FF+1A was comparatively lower than that of shallot, other genes involved in rust resistance might exist on chromosomes other than 1A. It is also possible that alleles on A. fistulosum chromosomes could counteract or weaken the function of resistance gene(s) from shallot. The degree of rust resistance exhibited by MAALs, multichromosome addition lines, and hypoallotriploids was unclear in full-grown plants under both controlled-environment and field conditions, even though shallot showed distinct resistance. Some individuals of FF+1A+5A repeatedly showed low severity of symptoms comparable to that of shallot. However, other individuals with chromosome additions did not show resistance regardless of whether they carried chromosomes 1A and 5A from shallot. Thus, consistent results regarding the relationship between rust resistance and shallot chromosomes have not been obtained. These results implicate possible interactions between genes from shallot and A. fistulosum. In field tests, plants were continuously attacked by rust for a long period of time. As the growth and differentiation of the pathogen is retarded in A. cepa (Jennings et al., 1990a), uredinia formation likely varies depending on host and environmental conditions. Results in the present study indicate that A. cepa possesses resistance to rust at the seedling stage, but not at the adult plant stage. Jennings et al. (1990b) discussed the effects of plant age, leaf position, and leaf segment on infection of leek by rust. They proposed that evaluation of resistance should be carried out on several leaves per plant on replicate adult plants and that several components of the disease, including the latent period, pustule density, and pustule length, should be measured. Resistance to rust and other diseases often changes as plants mature (de Jong, 1995).

Vu et al. (2012) reported that shallot expresses chemical compounds that show antifungal effects against *F. oxysporum*. In the present study, FF+2A showed the highest level of resistance to *Fusarium* wilt among the eight *A. fistulosum* – shallot MAALs. The FF+2A line also showed a specific saponin band derived from shallot. Lanzotti et al. (2012) reported the inhibition of some phytopathogenic fungus by saponins extracted from white onion. However, the addition of chromosome 2A had no effect on the rust resistance of lines carrying it in the present study.

A. fistulosum – shallot MAALs are highly fertile (Shigyo et al., 1999b) and the extra chromosomes can be transmitted via both male and female gametes (Shigyo et al., 1999a). Recombination between A. fistulosum and A. cepa can be induced during homoeologous chromosome pairing and crossing over during meiosis. Many similar studies using MAALs in other crops have been described previously. For example, Savitsky (1978) developed *Beta vulgaris – B. procumbens* MAALs for nematode (*Heterodera schachtii*) resistance and selected resistant diploid plants from among their progenies that were assumed to have translocations between homoeologous chromosomes. Resistance genes for the foliar disease Cercospora beticola and the soil-borne fungus Polymyxa betae, the vector of Beet Necrotic Yellow Vein virus, were found in *Beta vulgaris – B. procumbens* MAALs and B. vulgaris – B. patellaris MAALs (Paul et al., 1992, Mesbah et al., 1997). Kaneko et al. (1996) identified a Turnip Mosaic Virus resistance gene in kale - radish MAALs. Akaba et al. (2009) reported club root resistance originating from a radish chromosome in a Brassica napus - Raphanus sativus MAAL. Peterka et al. (2004) reported a monosomic chromosome addition for transferring resistance to beet cyst nematode from radish (R. sativus) to rape (B. napus). Thus MAALs will be useful materials for stable introgression of beneficial genes from extra chromosomes into recipient chromosomes.

In conclusion, the present study reported resistance of *A. cepa* cultivars including bulb onion and shallot against *P. allii* isolated from *A. fistulosum* in controlled environment and field tests. The gene(s) for rust resistance located on chromosome 1A act mainly during the seedling stage. The information presented here will be helpful in breeding programs for the development of rust-resistant bunching onion varieties.

# Chapter III: Mapping of quantitative trait loci for rust resistance in bunching onion

#### Introduction

Bulb onion (A. cepa) is highly resistant to rust (Wako et al., 2015), therefore, so resistance genes from A. cepa will be introduced into A. fistulosum via introgressive hybridization using alien chromosome addition lines. On the other hand, considerable variation has been observed in the degree of disease severity, namely the area under the disease progress curve (AUDPC) (Wako et al., 2012). Bunching onion cultivar types such as 'Senju-Aigara', 'Kaga', and 'Kujo' exhibit high susceptibility to the disease, whereas cultivars like 'Senju-Aiguro' and 'Senju-Kurogara' were found to be more resistant than the others. This indicated that the rust resistance present in bunching onion was likely a quantitative trait. Among the 133 cultivars tested, 'Seito Ippon', 'Iwai 2', 'Choju', 'Senami', 'Fuyuogi-Ippon', and 'Toyokawa-Futo' showed the lowest values of AUDPC (Yamashita et al., 2005). Using these relatively resistant cultivars, a recurrent selection program was conducted to achieve a high level of field resistance to rust, because this breeding method has been successful in improving quantitatively inherited traits in other outcrossing crops. As a results of two cycles of recurrent selection for rust resistance followed by several generations of continuous selection within selfed-lines, a parental line 'Negi Chuukanbohon Nou 1' with the highest rust resistance found so far in bunching onion was developed (Wako et al., 2012).

Quantitative trait loci (QTL) analysis based on a genetic linkage map can effectively reveal the mode of inheritance of agronomic traits. However, despite the economic importance of bunching onion, its genetic characteristics have been poorly studied and molecular approaches were needed to help clarify the genetic control of many traits of interest. Approximately 2000 gSSRs have been identified from bunching onion (Song et al., 2004; Tsukazaki et al., 2007) and used to construct genetic linkage maps (Ohara et al., 2005a; Tsukazaki et al., 2008; 2011; 2012). In a previous study, a major QTL for pseudostem pungency was identified using the SaT03 population derived from a cross between a rust-resistant line Sa03 and a susceptible line T03 (Tsukazaki et al., 2012).

Sa03 is a selfed line derived from 'Negi Chuukanbohon Nou 1'.

In the present study, QTL analyses of the rust resistance of 'Negi Chuukanbohon Nou 1' were conducted in order to determine its genetic basis and to develop DNA markers for efficient selection of rust resistance in bunching onion.

#### **Materials and Methods**

#### Plant materials and pathogen

The mapping population SaT03 (119 individuals) was used in the present study. SaT03 is the segregating F<sub>2</sub> derived from a cross between Sa48-10s-7ic-3s-2ic-3s-1s-9s (Sa03) and T26-4s-2s-2s (T03). Sa03 is an inbred line derived from the cultivar 'Chuukanbohon Nou 1' (Wako et al., 2012) and is highly resistant to rust under field conditions. T03, a susceptible type, is an inbred line derived from the cultivar 'Fuyuwarabe' (Wako et al., 2010).

The *P. allii* isolate used in the present study was same as that described in Chapter II. Because *P. allii* is an obligate biotrophic pathogen, it requires live host tissue to reproduce. Therefore, *P. allii* uredospores were collected from bunching onion leaves and stored at -30 °C until just before use as inoculum.

# **Evaluation of rust resistance**

Triple inoculation tests were conducted in the greenhouse at the NARO Institute of Vegetable and Tea Science to examine rust resistance in the mapping population SaT03 (Table 4). Seeds of F<sub>2</sub>-derived F<sub>3</sub> (F<sub>2:3</sub>) lines of the SaT03 population were sown in 200-cell plug trays and transplanted into the plastic greenhouse in a randomized design with five replications and 10 plants per replication. The plants were grown for several months until reaching the adult plant stage. Inoculation was performed by spraying with a mixture of 1 mL  $5 \times 10^3$  uredospore suspension with 0.01% Tween 20 and 0.5% talc as described in detail in Chapter II. The disease index (DI) was calculated according to the severity of rust symptoms scored from one to four times after inoculation. DIs were defined using a 0–5 scale as follows: 0, no uredinia; 1, 1–3 uredinia per leaf; 2, 4–10 uredinia per leaf; 3, more than 11 uredinia per leaf; 4, uredinia distributed over the entire leaf; and 5, uredinia

Trial	Number of F <sub>2:3</sub> lines	Sowing date	Transplanting date	Inoculation date	Scoring date
2009	100	15 July 2008	3 Sep. 2008	6 Mar. 2009	17 Apr. 2009
2013	80	26 June 2012	28 Aug. 2012	31 Jan. 2013	28 Feb., 6, 13 and 27 Mar. 2013
2014	88	5 July 2013	3 Sep. 2013	3 Apr. 2014	21 Apr., 6 May 2014

**Table 4.** Sowing, planting, inoculation and scoring dates for evaluation of rust resistance in F<sub>2:3</sub> population.

distributed densely over the entire leaf. AUDPCs were calculated according to methods described in Chapter II.

# QTL analysis

QTL analysis was performed using the composite interval mapping method with the computer program QTL Cartographer version 2.5 (Wang et al. 2007). The log likelihood (LOD) threshold was determined with 1000 permutations at P = 0.05. Statistical analysis of the associations between the genotypes of plants in the F<sub>2</sub> generation and the average DI or AUDPC of F<sub>2:3</sub> plants of the same genotype was conducted at each QTL-linked SSR locus in each trial using the Tukey–Kramer Honestly Significant Difference (HSD) test (P = 0.05) with JMP version 4.0 (SAS Institute Inc., NC, USA).

#### Results

# Variation in rust resistance in the SaT03 population

In the 2009 trial, DI was determined only one time at 42 days after inoculation (Table 4). The average DI of Sa03 was 1.4, while the DI of T03 was 3.6 (Fig. 3), and the DI of F<sub>1</sub> hybrids was 3.3. DIs of the F<sub>2:3</sub> lines ranged continuously from 1.3 to 3.9. In the 2013 trial, DIs were scored four times at 31 to 59 days after inoculation. In the 2014 trials, DIs were scored two times, at 18 to 33 days after inoculation. AUDPCs were used for evaluation of rust resistance in both trials. AUDPCs of Sa03 were 84 in the 2013 trial and 27 in the 2014 trial, while those of T03 were 172 and 52, respectively (Fig. 4). AUDPCs of the F<sub>2:3</sub> lines ranged continuously from 68 to 162 in the 2013 trial and from 28 to 59 in the 2014 trial. The correlation coefficient between DIs of F<sub>2:3</sub> lines in the 2009 trial and AUDPCs in the trial 2013 was 0.62 (Fig. 3). The correlation coefficient between AUDPCs of F<sub>2:3</sub> lines in the 2013 trial and the 2014 trial was also 0.62 (Fig. 4). Transgressive segregation for resistance was observed in the 2013 trial and also for susceptibility in the 2014 trial.

# QTL analysis for rust resistance

In the 2009 trial, two QTLs were detected on the linkage groups 1a on chromosome 1



**Fig. 3.** Distribution and correlation coefficient between disease indices (DIs) in 2009 trial and AUDPCs in 2013 trial with 82 F<sub>2:3</sub> lines of SaT03 population.



**Fig. 4**. Distribution and correlation coefficient between AUDPCs in 2013 trial and 2014 trial with 77 F<sub>2:3</sub> lines of SaT03 population.

(Chr. 1a) and 8a on chromosome 8 (Chr. 8a) and were designated as qRst1a and qRst8a, respectively (Fig. 5). The LOD peak for qRst8a (5.7) was higher than that for qRst1a (4.9), and the proportion of phenotypic variation explained by qRst8a (20.4%) was also higher than that explained by qRst1a (14.2%) (Table 5). To examine the effects of these two QTLs, the F<sub>2</sub> progeny were classified according to the genotype at each marker and the bolting time of F<sub>2:3</sub> lines was correlated with the nine observed genotypes (Table 6). F<sub>2</sub> progenies homozygous and heterozygous for the linked AFA02F07 and CF436630 alleles derived from the Sa03 parent showed significant differences in rust severity from the progeny carrying the homozygous genotypes derived from the T03 parent (Table 6).

A QTL was detected in a region on linkage group Chr. 3a in the 2013 trial (qRst3a). The maximum LOD score (3.6) and the observed variation explained (12.7%) for qRst3a were both lower than for qRst1a and qRst8a detected in the 2009 trial (Table 5). When the F<sub>2:3</sub> lines were categorized according to the genotype of their preceding F<sub>2</sub> generation at the linked locus ACE320, F<sub>2</sub> progenies homozygous and heterozygous for the Sa03 allele showed significant differences in AUDPC from the progeny carrying the homozygous genotypes derived from the T03 parent (Table 7). No significant QTL was detected in the 2014 trial.

#### Discussion

There have been several linkage mapping studies for traits in *Allium* crops (Baldwin et al., 2014; McCallum et al., 2006; 2007; Tsukazaki et al., 2012). However, no approach had yet been reported for disease resistance in *A. fistulosum*. Susceptibility to rust in the F<sub>2:3</sub> individuals showed a continuous distribution from susceptible to resistant, suggesting that rust resistance is controlled by quantitative trait loci. In the present study, three QTLs related to rust resistance were identified in two inoculation trials in the greenhouse. Despite the considerably high correlation of rust severity between trial years, different QTLs for rust resistance were observed. Variances explained by those QTLs were relatively low, suggesting that there are other loci involved in rust resistance in Sa03. In addition, improvement of the current linkage map will make it more informative for future genetic studies in bunching onion.



**Fig. 5.** Linkage map and QTLs detected in SaT03 population in 2009 trials and 2013 trials. QTLs are indicated by the white boxs (2009 trials) and the black box (2013 trials) to the left sides of chromosomes and the positions of LOD peaks are shown by arrows.

Trial	QTL	Linkage group	Closest marker of peak LOD score	LOD peak	Additive effect <sup>a</sup>	Dominant effect <sup>a</sup>	$R^2$ (%) <sup>b</sup>
2009	qRst1a	Chr. 1a	AFA02F07	4.9	0.3	-0.1	14.2
2009	qRst8a	Chr. 8a	CF436630	5.7	0.3	-0.1	20.4
2013	qRst3a	Chr. 3a	ACE320	3.6	5.0	-4.0	12.7

**Table 5.** QTLs for rust resistance in SaT03 F2:3 population.

<sup>a</sup>Additive or dominant effect of 'Sa03' allele.

<sup>b</sup>Percentage of variance explained at the peak of QTL.

	Marker g	enotype <sup>a</sup>	Number		Significant		
Population	AFA02F07	CF436630	of F <sub>2:3</sub>	Rust severity <sup>b</sup>	diffe	difference <sup>c</sup>	
	(qRst1a)	(qRst8a)	lines	Sevency .	AA	BB	
F <sub>2</sub> (Sa03 x T03)	А	А	7	2.81		*	
	А	Н	7	3.05			
	А	В	8	3.11			
	Н	А	9	2.51		*	
	Н	Н	32	3.00		*	
	Н	В	8	3.32			
	В	А	4	3.28			
	В	Н	9	3.26			
	В	В	7	3.72	*		
Sa03	А	Α		1.43			
F1 (Sa03 x T03)	Н	Н		3.27			
Т03	В	В		3.63			

**Table 6.** Rust severity in  $F_{2:3}$  individuals categorized by the  $F_2$  genotypes at AFA02F07 and CF436630 in SaT03 population (Trial 2009).

<sup>a</sup> A, genotypes of homozygous for Sa03 allele (rust resistant); H, heterozygous; B, homozygous for T03 allele (rust susceptible).

<sup>b</sup> Average number of rust severities of F<sub>2:3</sub> lines.

<sup>c</sup> When significant difference between each genotype and AA or BB is detected by Tukey-Kramer HSD test (P = 0.05), asterisk is denoted.

Population	Genotype	Number of F <sub>2:3</sub> lines	
F <sub>2</sub> (Sa03 x T03)	T03 homo	18	128 a
	Hetero	39	110 b
	Sa03 homo	14	107 b
Т03	T03 homo		179
F1 (Sa03 x T03)	Hetero		185
Sa	Sa03 homo		84

**Table 7.** Rust severity in  $F_{2:3}$  individuals categorized by the  $F_2$  genotype at ACE320 (Trial 2013).

<sup>a</sup> AUDPC of  $F_{2:3}$  lines. Different letters indicate significance in Tukey-Kramer HSD test (P = 0.05).

Although no significant differences were observed among cultivars of *A. fistulosum* in seedling tests, there were significant differences in field resistance to rust in full-grown plants (Wako et al., 2015). The resistance of Sa03 is derived from the cultivar 'Chuukanbohon Nou 1' (Wako et al., 2012), which was developed by recurrent selection. This parental cultivar likely possesses multiple genes relating to rust resistance.

In a previous study, it was reported that *A. cepa* showed rust resistance and possess genes relating to resistance on chromosome 1A and 5A (Wako et al., 2015). Resistance genes from *A. cepa* will likely be introgressed into *A. fistulosum* using MAALs. However, no recombinant containing a crossover between the genomes of two species both carrying rust resistance has been obtained thus far. In contrast, recurrent selection would help accumulate a number of rust resistance genes in *A. fistulosum*, therefore, this method would be a practical and reliable breeding approach. For example, in vegetable crops, the fruit yield of cucumber has been improved by 10 cycles of recurrent selection (Wehner and Cramer, 1996). Populations with improved resistance to *Verticillium dahliae* were developed by two cycles of recurrent selection in pepper (*Capsicum annuum*) (Palloix et al., 1990).

Rust resistance genes have been studied in other plant species. In particular, extensive analyses have been performed in wheat (Dedryver et al., 1996; Huang and Gill, 2001; Mago et al., 2002; Raupp et al., 1983; Stein et al., 2000) and barley (Dreiseitl and Steffenson, 2000; Feuerstein et al., 1990; Jagathpriya et al., 2003). To date, a number of qualitative and quantitative rust resistance alleles have been found in wheat and barley. The wheat leaf rust resistance genes, Lr10 (Feuillet et al., 2003) and Lr21 (Huang et al., 2003), and the barley stem rust resistance gene, Rpg1 (Brueggeman et al., 2002), were successfully cloned by map-based isolation strategies. In bunching onion, however, rust resistance genes have not yet been identified. The rust resistance developed in this study is not true resistance, but rather what is known as field resistance. This type of resistance is under polygenic control, and is therefore probably more effective against a broad range of pathogenic races and more durable than monogenic resistance, which may break down under severe epiphytotics (Brewbaker, 1983).

The QTLs identified here were identified using STS (sequence-tagged site) markers,

most of which are co-dominant and near the top of the LOD peaks for each QTL. Therefore, they would be useful for selection of bunching onions with a high level of rust resistance.

# Chapter IV: Mapping of quantitative trait loci for bolting time in bunching onion

#### Introduction

Because bunching onions cannot be stored for long periods, adaptation to different growing locations throughout the country and year-round production are very important in Japan. The late-bolting trait is essential in bunching onion cultivars for spring and summer production. Genetic variation in bolting time in bunching onion is based on seedling age and low-temperature requirements (Inden and Asahira, 1990). Low temperature (3–15 °C) is required for flower formation, and the optimal temperatures and exposure periods required for flowering differ according to cultivar (Yamasaki et al., 2000b; Abe and Nakazumi, 2004; Dong et al., 2013). Additionally, a short photoperiod also promotes flower bud initiation and a long photoperiod promotes elongation of the flower stalk after flower initiation (Yamasaki et al., 2000a). However, genetic and molecular studies of bolting time in bunching onion have not been reported thus far.

In *Brassica rapa*, several quantitative trait loci (QTL) controlling flowering time were identified and found to colocalize with the ortholog of *FLOWERING LOCUS C* (*FLC*) from *Arabidopsis* (Schranz et al., 2002; Kakizaki et al., 2011; Kitamoto et al., 2014). Several candidate *BrFLC* genes have also been discussed in terms of their functions in flowering time and the vernalization response in *B. rapa* (Kim et al., 2007; Kitamoto et al., 2014).

In bulb onion, the *FLOWERING LOCUS T* (*FT*)-like genes have been investigated for functional characterization. Up-regulation of the expression of the *AcFT2* gene appears to be involved in the vernalization-dependent initiation of flowering, whereas the expression of two other *FT* genes, *AcFT1* and *AcFT4*, has been correlated to bulb formation (Lee et al., 2013). Additionally, a QTL designated as *AcBlt1* that conditions bolting has been mapped to chromosome 1 (Baldwin et al., 2014), but the association between this QTL and the flowering candidate genes is unclear. Genetic studies of flowering are very limited in bunching onion. Therefore, it is important to investigate the QTL for bolting time as a first step toward understanding the genetics of flower initiation in bunching onion.

In the current study, bolting time was evaluated in two bunching onion populations under different environmental conditions. Major QTLs detected on the same linkage groups in the two populations were compared using the same set of markers. The Taiwanese ever-flowering cultivar 'Bei-cong' was used as the genetic source of the early-bolting trait. Most Taiwanese cultivars have minimal requirements for low-temperature to induce flower initiation (Inden and Asahina, 1990). Yamasaki et al. (2012) reported that 'Bei-cong' showed seed vernalization-like response for flower formation. A linkage map was constructed using a population derived from a cross between the ever-flowering line Ki and the late-bolting line C, and QTL analysis for bolting time was conducted under vernalized and unvernalized conditions. Additionally, QTL for bolting time was identified using another SaT03 map constructed in a previous study, and the resulting maps for these two populations were compared.

#### Materials and methods

#### **Plant materials**

Two mapping populations, KiC (134 individuals) and SaT03 (119 individuals) were used in the present study. KiC is the segregating F<sub>2</sub> derived from a cross between two bunching onion inbred lines, K1s-5s-2s-2s-4s-5 (Ki) and Cho1s-1s-2s-2s-8s (C). Ki is an inbred line derived from the Taiwanese ever-flowering cultivar 'Bei-cong' (named 'Kitanegi' in Japanese, JP138785). C is an inbred line from the late-bolting cultivar 'Cho-etsu' (JP133892). SaT03 is the segregating F<sub>2</sub> derived from a cross between Sa48-10s-7ic-3s-2ic-3s-1s-9s (Sa03) and T26-4s-2s-2s (T03). Sa03, an early-bolting type, is an inbred line derived from the cultivar 'Chuukanbohon Nou 1' (Wako et al., 2010), and T03, a late-bolting type, is an inbred line derived from the cultivar 'Fuyuwarabe' (Wako et al., 2012). The mapping population SaT03 is identical to that previously used for QTL analysis of pseudostem pungency (Tsukazaki et al., 2012).

## Construction of a linkage map

Mainly genomic SSRs from bunching onion and EST-SSRs from bulb onion were used in the present study (Tsukazaki et al., 2011). Polymerase chain reaction (PCR) conditions and genotyping of SSR and EST markers were performed either as described in Tsukazaki et al. (2011) or according to the protocols published at the Vegmarks database (http://vegmarks.nivot.affrc.go.jp). Linkage analysis was performed using the computer program MAPMAKER/EXP 3.0b (Lander et al., 1987), using the Kosambi function to obtain cM values (Kosambi, 1944). A log of the odds (LODs) threshold of 4.0 was used to group and order the markers. Each linkage group was assigned to chromosomes by comparison to the previously constructed maps (Tsukazaki et al., 2008, 2011, 2015).

#### **Evaluation of bolting time**

Bolting time was evaluated during field trials in two years for the populations KiC and SaT03. Additionally, bolting time for the KiC population was also evaluated under heated greenhouse conditions in the winter of 2007–2008 (the 2008 greenhouse trial, Table 8).

Firstly, the KiC population was evaluated for bolting time under non-vernalized conditions. A total of 121 F2-derived F3 (F2:3) lines from the KiC population were sown in 200-cell plug trays (two seeds per plug) on 18 July 2007, then 20 plants per line were transplanted into plastic boxes (65 cm  $\times$  18 cm  $\times$  22 cm) on 29 August 2007 and grown in the greenhouse kept at a minimum temperature of 10 °C (2008 trial). The KiC population was then grown in the open field at the NARO Institute of Vegetable and Tea Science (34.8°N, 136.4°E) through the winter in 2008–2009 and 2009–2010 (2009 field trial and 2010 field trial) to measure bolting times (Table 8). Bolting time of the SaT03 population was measured in the 2013 field trial and the 2014 field trial (Table 8). Standard crop management for bunching onion was used and the bolting date of each plant was recorded. Bolting dates of F2 individuals were used to estimate the average date of bolting of all plants in the  $F_{2:3}$  lines. A plant was scored as bolting when the top of the flower scape bud emerged. The bolting time of the KiC population was defined as the number of days between the average bolting date of Ki and the bolting date of each F<sub>2:3</sub> line. In the same manner, the bolting time of the SaT03 population was defined as the number of days between the average bolting date of Sa03 and the bolting date of each F<sub>2:3</sub> line.

Population	Trial	Number of F <sub>2:3</sub> lines	Number of plants in each lines	Condition	Sowing date	Transplanting date
KiC	2008	121	20	Heated greenhouse <sup>a</sup>	18 July 2007	30 Aug. 2007
KiC	2009	108	60	Open field	11 July 2008	8 Sep. 2008
KiC	2010	108	45	Open field	4 June 2009	10 Aug. 2009
SaT03	2013	80	42	Open field	26 June 2012	28 Aug. 2012
SaT03	2014	87	42	Open field	5 July 2013	3 Sep. 2013

**Table 8.** Sowing and planting dates for evaluating bolting time in  $F_{2:3}$  populations.

<sup>a</sup> Minimum temperature in the greenhouse was 10 °C during winter.
## QTL analysis

QTL analysis was performed by the composite interval mapping method using the computer program QTL Cartographer version 2.5 (Wang et al., 2007). The log likelihood (LOD) threshold was determined with 1000 permutations at P = 0.05. Statistical analysis of the associations between the genotypes of plants in the F<sub>2</sub> generation and the average bolting date of F<sub>2:3</sub> plants of the same genotype was conducted at each QTL-linked SSR locus in each trial using the Tukey–Kramer Honestly Significant Difference (HSD) test (P = 0.05) with JMP version 4.0 (SAS Institute Inc., NC, USA).

#### Results

## **Evaluation of bolting time**

In the 2008 greenhouse trial, the bolting date of Ki was 12 January 2008, 136 days after transplanting, while bolting in C was not observed before 30 June 2008. The bolting date of the F<sub>2:3</sub> lines ranged continuously from 49 to 180 days later than that of Ki, and 10 of the F<sub>2:3</sub> lines did not bolt.

In the field trials in 2009 and 2010, the bolting dates of Ki were 19 November 2008 and 20 November 2009, respectively (Fig. 6). The differences in the bolting dates of C relative to Ki were 159 days in the 2009 field trial and 170 days in the 2010 field trial. The bolting dates of the F<sub>2:3</sub> lines ranged from 64 to 153 days and from 25 to 152 days later than that of Ki in the 2009 and 2010 field trials, respectively. The bolting times of the F<sub>2:3</sub> lines in the field trials in 2009 and 2010 were highly correlated (r = 0.93). Transgressive segregation was not observed in these three trials.

The differences in the bolting dates of the  $F_{2:3}$  lines of the SaT03 population were smaller than those of the  $F_{2:3}$  lines of the KiC population. The bolting dates for Sa03 were 26 March 2013 in the 2013 field trial and 28 March 2014 in the 2014 field trial (Fig. 7). The differences in the bolting dates between T03 and Sa03 were 49 days in the 2013 field trial and 51 days in the 2014 field trial. The bolting dates of the  $F_{2:3}$  lines were 19 to 60 days later than that of Sa03 in the 2013 field trial and 17 to 65 days later than that of Sa03 in the 2014 field trial. The bolting times of the  $F_{2:3}$  lines in field trials in 2013 and 2014 were also highly correlated (r = 0.81). Transgressive segregation for late bolting was



**Fig. 6.** Distribution of the bolting date in 105 F<sub>2:3</sub> lines of KiC population and correlation coefficient between trials.



**Fig. 7.** Distribution of the bolting date in 76 F<sub>2:3</sub> lines of SaT03 population and correlation coefficient between trials.

observed in both the 2013 and 2014 field trials of the SaT03 population.

# Construction of a linkage map in the KiC population

A linkage map consisting of 16 linkage groups was constructed with a total of 266 markers, including 249 SSRs and one gene from bunching onion, and with two SSRs, four InDels (insertion-deletion), and 10 CAPS designed from bulb onion ESTs (Table 9 and Table 10). The total map length was 2802 cM and the average marker interval was 10.5 cM. All of the linkage groups were assigned to eight chromosomes on the previously constructed map using the same markers.

# QTL analysis of bolting time in the KiC population

In the 2008 greenhouse trial, two QTLs were detected on the linkage groups 2a on chromosome 2 (Chr. 2a) and 6a on chromosome 6 (Chr. 6a) and were designated as qBlt2a and qBlt6a, respectively. The LOD peak for qBlt2a was higher than that for qBlt6a, and the proportion of phenotypic variation explained by qBlt2a was also higher than that explained by qBlt6a (Table 11).

QTLs were detected on the linkage groups 1a on chromosome 1 (Chr. 1a; qBlt1a) and 2a on Chr. 2a (qBlt2a) (Fig. 8 and Table 11) using data from the field trials conducted in 2009 and 2010. The LOD peaks for these two QTLs were located in close proximity in both trials. The LOD peaks for qBlt2a in the 2009 and 2010 field trials were located close to the QTL detected in the 2008 greenhouse trial. The maximum LOD scores for qBlt1a and qBlt2a were 10.2 and 8.8 in the 2009 field trial, and 10.3 and 10.7 in the 2010 field trial, respectively (Table 11). The QTL qBlt1a and qBlt2a accounted for 15.4% and 13.1% of the observed variation, respectively, in the 2009 field trial, and 14.7% and 16.3% of the observed variation, respectively, in the 2010 field trial. Two additional QTLs were found on the linkage groups 1b on chromosome 1 (Chr. 1b; qBlt1b) and 3b on chromosome 3 (Chr. 3b; qBlt3b) in the 2010 field trial (Fig. 8), with effects lower than those of qBlt1a or qBlt2a. The maximum LOD scores for qBlt1b and 4.7, and the proportion of variance explained by these QTLs were 12.0% and 6.1%, respectively (Table 11).

					KiCı	uan	Fragn	nent	
Marker	GenBank	Core reneats	Forward primer	Reverse primer		dur	sizes (t	ni (qo	Renorted in
name	Accn.	COLUTION	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	Ki	С	
Bulb onion St	SR								
ACE112	CF444011	(TCA)6	ecett et gett t gtt tatt ett cacae	acgagtttt gttt gtgagct ctt gg	Chr.5b	152.0	127	139	Tsukazaki et al. 2011
ACM006	BQ580184	(CTC)7	gcagt tct cccttt gt aaaat ca	gt gat ggat gagt ggat gga	Chr.3b	0.0	199	193	Kuhl et al. 2004
Bunching oni	on SSR								
AFA01C06	AB564753	(TG)7tata(TG)6	t at gt at a ga a a t t gt a c ggggt t gg	aacattaaagggggggggggggaggt caatagtattt	Chr.6a	332.6	244	242	Tsukazaki et al. 2007
AFA01C07	AB499342	(TG)14	catt ccagacgt tt gtt gt gacag	tggtctccaatacgagcat caaaa	Chr.1b	86.8	284	280	Tsukazaki et al. 2007
AFA01E06	AB499345	(AT)3gaggctgctagatttccata aca(AC)14(AT)5	caatt ct ggt caaacc gagact ca	tctgttccatctattcatgttcctcc	Chr.4a	102.6	251	266	Tsukazaki et al. 2008
AFA01E09	AB499346	(AT)3ctatatgt(AC)13	cccaacctactt gagggatt gcta	gt tet gt gt at geaggeaat tt gg	Chr.2a	130.0	166	178	Tsukazaki et al. 2008
AFA01F04	AB795037	(TG)16	gc ggt aggcagtt tt ct gtt gt gt	t gagt a gtt gaac att ggt ggc gt	Chr.1b	0.0	212	196	Tsukazaki et al. 2015
AFA01F12	AB499348	(AC)12(AT)3	t gaagggggacaaaat aagaa gca	tetececacttaaaaagaattteg	Chr.5a	60.4	164	160	Tsukazaki et al. 2015
AFA01G11	AB564758	(AC)15	t cacact gct ggt agc caggt tag	ggac ctt taa at gt gt agat ac gggt g	Chr.4a	34.4	178	180	Tsukazaki et al. 2011
AFA01H08	AB795037	(AC)9at(AC)5atac(AT)5	cacgt tt catt aggt t ggggaaaa	cacaaaaat ccacgagt cgaact g	Chr.4a	45.7	257	263	Tsukazaki et al. 2015
AFA02B05	AB795039	(TG)3aaca(TA)8(TG)17aagtgaaacca	t gcaa cggt agagagat gat ggag	t gaaat cccct gcga gatt gt agt	Chr.5a	30.5	249	257	Tsukazaki et al. 2015
		gagg(caa)3							
AFA02B06	AB564763	(AC)12(AT)4atatgtgtatgg(AT)3	at ggggagct caaagtt gt gaatc	gggaaggagt to cat aaacgat co	Chr.7b	13.0	279	281	Tsukazaki et al. 2011
AFA02C10	AB795040	(TG)3tatgttctaga(AG)3(TG)5ta(TG)3 ta(TG)7	gggtaggtttcccttgtttggaag	tgagggat gcattt gctattttgag	Chr.2a	45.9	252	244	this study
AFA02D02	AB795041	(TA )4aacgttag(TG)13	acgtacatatcgaaccattt cccg	gcct cct ccccatt caagaact at	Chr.5b	116.9	142	140	this study
AFA02D03	AB795042	(AC)16a(TA)8	ggatgcatgcttggagtcttgtta	t gaaact gcct caacacaaaggaa	Chr.2a	238.3	220	224	this study
<b>AFA02D05</b>	AB499354	(TA)3cta(AT)3(AC)11	ttaggcgt cacataccaaagct acaa	tggaattt caaaaagacaccgagtaa	Chr.5b	12.8	232	234	Tsukazaki et al. 2008
AFA02F09	AB499356	(at)3(ac)14a(ta)3tg(ta)3	cctaggtt cat taggggact at agga	ccaagcaccagtat ct gcct tt ct	Chr.8a	61.0	296	290	Tsukazaki et al. 2007
AFA02H08	AB499357	(TA)3cat(AC)4aa(AC)12	a gatett ggat agtt at ta aggt agtt ceagt aga	gggctgaaatattatgtgggtttg	Chr.6a	180.9	181	187	Tsukazaki et al. 2007
AFA03B12	AB564772	(AC)12	cat catt aggcgct ac a agggt tc	tccagt gggttgtaacaaggaagg	Chr.4a	107.4	220	228	Tsukazaki et al. 2011
AFA03D07	AB499363	(TG)3(CA)18	t gt cct cgaaaat gagat gct t ca	tgaaat gtaacccgact gt ccaaa	Chr.6a	378.0	301	282	Tsukazaki et al. 2008
AFA03F08	AB499365	(TA)4(TG)7ta(TG)7	ttaggtaaagggacgaaacgacca	tgcct ccaggact gaacaat acaa	Chr. la	110.4	302	286	Tsukazaki et al. 2008
AFA04D08	AB564782	(AC)4gc(AT)3gg(AC)4tcatacaaa(C A)12(TA)6g(AC)4	t t caat cgcact ggt ggt at gaac	tgcat gtt ccat ct ct ac ctaa cgtg	Chr.8a	0.0	260	262	Tsukazaki et al. 2011
AFA04E02	AB564784	(GC)3(AC)3ttat(AC)5aa(AC)15	ggaaccagtt gaggat gt tt aaggg	gagcactttttaggtgcctgtgct	Chr.6a	0.0	234	224	Tsukazaki et al. 2011
AFA04E06	AB564786	(TA)7(TG)18	ccctacacattttcatccgctctt	agctt ctt gaaggt ggat gt gct c	Chr.4b	133.0	234	232	Tsukazaki et al. 2011
AFA04H06	AB564789	(AC)6at(AC)8(AT)3	cagaggagcaat gt cat gcct a ga	ggcat gat aaaat ga gt gaaggt ga	Chr.3b	142.2	241	252	Tsukazaki et al. 2011
AFA05A06	AB564791	(AC)12tac(TA)3t(TA)3	c gcat acggt ct cat c cgct at aa	taggat cggcagt gt tggaat tt t	Chr. la	0.0	271	280	Tsukazaki et al. 2011
AFA05D05	AB564793	(TA)6cactac(CA)3tacat(AC)4ag(AC	ctt aat gtccccaat gat gt ga ctt	cgaaagtgtgacggaaattcgttt	Chr.3a	393.9	250	264	Tsukazaki et al. 2011
		C(1 Y)71(							

Table 9. Primer sequences of 2 bulb onion EST-derived SSRs and 249 bunching onion SSR markers located on KiC map.

Marker hameGenBank AcensCore repeats sequence (5-3)Reverse primer sequence (5-3)AFA05E01AB564794(AC)13(AT)7sequence (5-3)sequence (5-3)AFA05E01AB564794(AC)12ag(AT)5sequence (5-3)sequence (5-3)AFA05F04AB564794(AC)12ag(AT)5tegaggaat accept compare tecture gag at grant comparecetting ang gat transcriptingAFA05F01AB564799(AC)12ag(AT)5tegaggeat accept comparecetting ang gat transcriptingAFA05F01AB99374(CT)9(G)7(T(T)12ttttccggttgcgtacacettcettgaacattetting agg and gat cogAFA05F03AB99376(CA)3(TA)8(TO)14tegaggeat accept comparetetting agg and gat cogAFA05F04AB99377(CT)9(G)7(T(T)12ttttccggttgcgtacacettcettgaacattetting agg and gat cogAFA06E05AB99376(CA)3(TA)8(TO)14gag comparetetting agg and gat cogAFA06E05AB99376(CA)3(TA)8(TO)14gat comparetetting agg and gat cogAFA06E05AB99376(CA)3(TA)8(TO)14tetting agg and codtetting agg and codAFA06E05AB99386(TO)3(TA)8(TO)14tetting agg and codtetting agg and codAFA06E05AB99380(TO)3(AT)13tetting agg and codtetting agg and codAFA06F05AB99386(TO)3(TA)70AB99389tetting agg and codAFA07A08AB99380(TO)141tttrattgctcccug tragg agg and cod codacacat accat accatAFA07A08AB99380(TO)141tttrattgctcccug tragg agg and codtetting agg and cod<	orward primer Re squence (5'-3') se gitaaggcattacctatatccacc co argitacgcattacctatatccacc co argitactcogctt ggcaca to co cagagagcattaget at to accticctt ggcacaagca to to accticctt ggcaaagca at the gcaaagg gaggt at gt ggcaaagg gaggt at gt ggcaaagg gaggt at gt ggcaaagg gaggt at gt	verse primer quence (5'-3') ttt ggttt gggad at gtotog atttt gat ggga at gtotog atttt gat gga at tt gtaactagat gt tt citcact gora gocatat tatca it cagagaaggat acagaaggga it cagagaagge ttotott gaggt segaaaggott otott gaggt segaaaggott otott gaggt segaaaggott otott gaggt caaaat taacaact titt cat g	KiC Linkage group Chr.8a Chr.1b Chr.1b Chr.1b Chr.1b Chr.1b Chr.2a Chr.2a Chr.4a	map position (CM) 50.4 101.9 92.3 108.3 108.3 138.6 15.4 17.1	Fragn   sizes (t   Ki   Ki   239   239   239   230	ent p) in C 237 211 211 228 228	Reported in
nameAccn.Core repeatssequence (5-3)sequence (5-3)sequence (5-3)AFA05F01AB564794(AC)7aa(AC)11(AT)4catatcct[TA)3gegatageattactatatccaccctittigntggattgaactagragtatAFA05F01AB564794(AC)7aa(AC)11(AT)4catatcct[TA)3gerggacaaccagtacataagcacctittigntggattgaactagragtatAFA05F01AB99374(GT)9(TG)7ttt(TG)12ttttccgtgtgcgtacsecttoctigacattcctittigntggattgaactagragtatAFA05F01AB499374(GT)9(TG)7ttt(TG)12ttttccgtgtgcgtacsecttoctigacatagcatcctittigntggattgaactaggagtatAFA05F01AB499376(CA)6ia(TACA)2(AC)10atatac(AT)cccaggaggggagattacggaggagtattcttccggagagggttccttggggAFA06F05AB499376(CA)6ia(TACA)2(AC)10atatac(AT)cccagggggggggggggggggggggggggggggggggg	cquence (5'-3') se gitaaggcattacctatatccacc col argitatctogictt ggcaca too cagrgacaaccagtacaagca too cagrgacaactatgctta tot accttoctt gcacatttgctta tot tot fgccaggagggattt ggtta tot tgccaggttoctt ggctaaa tg tggcataggt gaggt at gt ggataggt gagacatccaaa aa	quence (5'-3') ttt ggt tt gggag at grotog atttt gat ggg attt gtaactagat gittt atttt gat gga attt gtaactagat gittt att cagngaagat acagaaggga tt cogngagga ggg aagge att caga seg accaccat ttt t cat g seg cost to ggg gat tt caaaattaacaact to geg att	Linkage group Chr.8a Chr.8a Chr.7b Chr.7b Chr.7b Chr.7b Chr.2a Chr.2a Chr.4a	position (cM) 50.4 101.9 92.3 108.3 108.3 138.6 15.4 17.1	Ki 239 203 318 230 230 230 230 222	C 237 211 211 228	Keponed in
AFA05E01AB564794(AC)13(AT)7gentagegata accesscell tigging grangegata gentAFA05E04AB564799(AC)12a(AC)11(AT)4catatoc(TA)3gentagegata accessteclt tigging grangegata gentAFA05E05AB564799(AC)12a(AC)11(AT)4catatoc(TA)3gentgigacaacagta canageateclt tigging grangegata gentAFA05E01AB499374(GT)9(G7)ttt(TG)12ttttccgfgtgcgtacacettect general gentteclt tigging general generalAFA05E01AB499376(CA)6ta(TACA)2t(AC)10atatac(AT)cenaggenerangeateclt tigging generalAFA06A08AB499377(CA)5ta(TACA)2t(AC)10atatac(AT)cenaggenerangeatetl tigging generalAFA06A08AB499377(CA)5ta(TACA)2t(AC)10atatac(AT)cenaggenerangeatetl tigging accessAFA06A08AB499377(CA)3ta(TO)10cenaggenerangeatetl tigging accessAFA06B07AB564806(C3)3ta(TO)10cenaggenerancestigging accessAFA06B05AB574806(C3)3ta(AC)10cenaggenerancestigging accessAFA06B05AB574806(C3)3ta(AC)10cenaggenerancesaccast generantAFA06B05AB574806(C3)3ta(AC)10cenaggenerancestigging accessAFA06B07AB574806(C3)3ta(AC)10cenaggenerancesaccanatatacast tigging accessAFA06B07AB574806(C3)3ta(TG)5t	gttaaggcattacctatatccacc col atgtatctcgtcttggcaca too cagtgacaaccagtacaaagca too acottocttgcacatttgctta too too too ggagagggtattttggtt ctt tgccaggttotttggcttaaa tg tgccaggttgaggtatgt tt tgcaaaggtgaggcatccaaa aaa	ttt ggt tt gggngt at gictog citting at ggr at tt graactagat gitt citt cact gorat gocat at tt at ca tt cagngaaaggt acagnagggg gggaaaggott ct cittig gaggt cage accaccat tt tt catig segt cott to gggg at tt caaaat taac aact to got accaga	Chr.8a Chr.1b Chr.1b Chr.7b Chr.7b Chr.7b Chr.2a Chr.2a Chr.2a	50.4 101.9 92.3 108.3 138.6 15.4 17.1	239 203 318 230 230 188 188 290 222	237 211 296 228	True 1 at al 2011
AFA05F04AB564797(AC)7aa(AC)11(AT)4catatcct(TA)3grangtatccgatatoctttagragattgatactgatagtAFA05F01AB564799(AC)12a(AT)5(GT)9(TG)7ttt(TG)12ttttccgtgttgcgtaccettcctgaraattgatatocttcactgatgcgatAFA05A08AB499376(GT)9(TG)7ttt(TG)12ttttccgtgttgcgtaccettcctgaraattgatatocttcactgatgcgatAFA06A08AB499376(CA)5ta(TACA)2t(AC)10ttatac(AT)octcaggragggatattgattottcagggaagtacaggagattacaggagattttggtAFA06A08AB499377(CA)5ta(TG)12gttgcagggattttggttottcagggaaggatacaggagagattacaggagagattacaggagattacaggagattaggAFA06A10AB499377(CA)5gag(AC)5tatactttaggtgttgcaggacacacattgatgAFA06B07AB564804(TG)3tata(TG)10ccttaal graaacatag ggggacatagtgcgagacacacatttttaatgAFA06B07AB564804(TG)3tata(TG)10ccttaal graaacatag ggggacatagtgggacacacaggattAFA06B07AB564804(TG)3tata(TG)10ccttaal graaacatag ggggacatagtgggacacacaggattAFA06B07AB564806(CA)5gag(AC)5tatactttaagtccttaal graaacatag ggggacattaccacagga accatagagacataggagattAFA06B07AB564806(TG)7010ccttaal graaacatag ggggacattaccataggagattAFA06F05AB795644(TG)7010ccttaal graaacatag ggggacacacaataccatagaacataggagattAFA06F05AB7956480(GT)14tttatgcfC516(TG)5acgatcggagagagacataggacataggacataggacataggagattAFA07A10AB99380(GT)14tttatgcfC516(TG)5acgatcggagagatccaaggatgattaccactgggagagatccaaggatgatggagattAFA07A10AB999388(TG)4TA9tgtafCTA)7acfacggagagatga	at grat ctc grett gge aca troc cagt ga caaccagt acaaage a too acct to ctt ge acatt t get ta too acct to ctt ge acatt t ggt t t cagg gag ggg at tt tug t ge cagg gag act a a a a a a a a a a a a a a a a a a	citti gai ggi atti gtaaciagai gitti citcaci geat gecalattai ca it cagagaaagat aca gaaggga gegaaaggett etti gaggi cia geacaccatti tteat gi segteette ggga att caaaattaacaaci tegetaett	Chr. Ib Chr. 7b Chr. 7b Chr. 7b Chr. 7b Chr. 2a Chr. 2a Chr. 4a	101.9 92.3 108.3 138.6 15.4 17.1	203 318 230 188 188 290 222	211 296 228	I SUKAZAKI ET AL. 2011
AFA05G05AB564790(AC)12ag(AT)5LgagggacaaccagacaaageatecttcactgcatgcatattacaAFA05H01AB499374(GT)9(TG)7ttt(TG)12ttttccgtgttgcgtacaccttccttgcacattgtatecttcactgcatgcagaAFA05H0AB499376(CA)6ia(TACA)2t(AC)10attac(AT)excergagagagattatettcarggaaaggacttcttggggAFA06H07AB499376(CA)6ia(TACA)2t(AC)10attac(AT)excergagagaggaattatettcarggaaaggacttcttggggAFA06H07AB564804(TG)3iata(TG)10ettaatgcaaactatgatggggaatttgstagcaccacattttcargtAFA06H07AB564804(TG)3iata(TG)10ettaatgcaaactatgatggggaatttgstagcaccacattttcargtAFA06H07AB564804(TO)3iata(TG)10ettaatgcaaactatgatggggaattaccaaattaaccacttggggaattAFA06H07AB564804(TO)3iata(TG)5ecgatatgggggaattaccaaattaaccacttggggaattAFA06H07AB564804(TA)97(G)7iata(TG)5ia(TG)5ia(TG)5iaeggagaattaccaattgarggggagattAFA07A08AB499381(TA)97(G)7iata(TG)5iaeggaggagagagagagagagagagagagagagagagag	cagt gacaaccagt acaaagca to acottoctt geacatt t gotta tot de aggag gag att tt ggt tot t gocaggt tot tt ggott a aa t gocaggt tot gag gag at gt tt aat geraaact at gat gaggt at gt tt g	citt cact geat gecatatitat ca it cagagaaagat acagaagggga gggaaagget ti ctett gaggt stageaccaccat titt cat gf segteet te gggg att caaaattaacaact tegetaccaga	Chr.7b Chr.1b Chr.7b Chr.7b Chr.2a Chr.1b Chr.4a	92.3 108.3 138.6 15.4 17.1	318 230 188 290 222	296 228	Tsukazaki et al. 2011
AFA05H01AB499374(GT)9(TG)7ttt(TG)12ttttccgtgttgcgtaccecttcttgagatttcttcaggaaaggrtacgaaaggrtacgaaggtacAFA06A08AB499376(CA)6ta(TACA)2t(AC)10tatata(AT)ctcaggaaaggrtacgacttgggaaaggrtacttrgggAFA06A10AB499377(CA)6ta(TACA)2t(AC)10tatata(AT)ctcaggaaggrtacgatggtaggaaggrtacgaAFA06B07AB499377(CA)5ga(AC)100cttaatgcaaggtttttggtatggtaggacaccacattttcafgAFA06B07AB564804(TG)3taaT(G)10cttaatgcaaggtgaggtggtaggacaccacattttcafgAFA06B07AB564804(TG)5faat(TG)10cttaatgcaagggaggagtggtaggacaccacattttcafgAFA06B07AB564804(TG)5faat(TG)5taattttaagtcggtargaggaatgAFA06B07AB564804(TG)5faat(TG)5taatttaagtcggcaattgggggaggattAFA06B07AB790504(TJ)9(TG)7taa(TG)5taa(TG)5tagtcgtcacataccaaattaaccactaggggggattAFA06B07AB499381(TG)14ttattgtcccatgtag(TA)4tggcaatggagggttagggggggattaccaaatgaacgaggaggattAFA07A08AB499381(TG)14ttattgtcccatgtag(TA)7acaatggagggggttaggggggaggacattaggaggaggattaccaatgaacgaggaggaggattAFA07A08AB499381(TG)14ttattgtcccatgtag(TA)7tacatgggagggggggggggggggggggggggggggggg	acctitecti geacatiti getta ted iteagagagagga attiti ggit ett i gecaggittectiti ggettaaa tig ittaal geaaactai gal gaggi at gi igeal aggi gagacatecaaa aa	tt cagagaaagat acagaagga gegaaaggett et et t gagg stageaceaceattttt cat gt segteett eg gggf att caaaattaacaacttegetaceaga	Chr.1b Chr.7b Chr.2a Chr.1b Chr.4a	108.3 138.6 15.4 17.1	230 188 290 222	228	Tsukazaki et al. 2011
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	dcaggagagggaatttt ggt ctt tgecaggttttcttt ggettaaa tg taatgeaaactatgatgaggaatgt ttt ggeataggdgaggeatcecaaa aa	:gegaaaggett et et t gagg clageaceaceatttt teat gf yegteette giggatt caaaattaacaaettegetaceaga	Chr.7b Chr.2a Chr.1b Chr.4a	138.6 15.4 17.1	188 290 222		Tsukazaki et al. 2008
AFA06A10AB499377(CAA)3(TA)8(TG)14gt gecaggtttt ggettaaAFA06B07AB564804(TG)3tata(TG)10ccttaatgeaactatgtt gggt atgAFA06B05AB564806(CA)5gag(AC)5tatctttaagtt certaatgeaactatgt ggga atgtgetagecaccattttteatgAFA06F05AB574806(CA)5gag(AC)5tatctttaagtt certaatgeaggagaactacaaatgetagecaccattttteatgAFA06F05AB795044(TA)9(TG)7tata(TG)5ta(TG)5ta(TG)5 acgat agg ggga actecataaccaaaattaacaacttegtageAFA06F12AB499380(GT)14tttattgtctccatgtatg(TA)4tgeratgeggggagttaccaaaattaacaaggaggaccaAFA07A10AB499381(TG)8tt(TG)5acgattggggtgttagegeacaatagaacaggaggacgaAFA07A10AB499381(TG)8tt(TG)5teggaaggtttagegeacaatagaacaggacgaAFA07A10AB499381(TG)8tt(TG)5teggaaggattagegetttagggtttagegeacaatagaacaggacgaAFA07A10AB499382(AC)11(AT)4tgtatg(TA)7teggaaggetttagggettt aggettatgegegtgeacaatagaacaggacatagegggggggggggggggggg	tgecaggttictti ggettaaa tg titaatgezaaactatgatgagtatgt tt <u>g</u> geataggtgagacateceaaa aa	ztageaceaceattttt cat gt segteettegtgatt eaaattaaeaaettegetaceaga	Chr.2a Chr.1b Chr.4a	15.4 17.1	290 222	190	Tsukazaki et al. 2007
AFA06B07AB564804(TG)3tata(TG)10ccttaatgeaactatgatgageattAFA06E05AB564806(CA)5gag(AC)5tatactttaagt(ctaagtactaatgeaactatgatgageattAFA06F05AB564806(CA)5gag(AC)5tatactttaagt(CTG)3AFA06F12AB795044(TA)9(TG)7tata(TG)5ta(TG)5AFA06F12AB499380(GT)14ttattgfctccatgatg(TA)4AFA07A08AB499381(TG)8tt(TG)5AFA07A10AB499381(TG)8tt(TG)5AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7AFA07A10AB499383(TG)4tTA)8tctc(TG)9AFA07A10AB499383(TG)4tTA)8tctc(TG)9AFA07A11AB499383(TG)4tTA)8tctc(TG)9AFA07A11AB564813(TG)4tTA)8tctc(TG)9AFA07A11AB564813(TG)4tTA)8tctc(TG)9AFA07A11AB564813(TG)4tTA)8tctc(TG)9AFA07A10AB995047(TO)10AFA07A10AB995047(TO)10AFA07A10AB995047(TO)10AFA07A10AB995047(TO)10AFA07A10AB995043(TO)4tTA)8tctc(TG)9AFA07A104AB75048(AC)11 <td>ttt aat geaaactat gat gaggt at gt geat aggt gagacat ceaaa aa</td> <td>segreettegreggratt caaaartaacategeraccaga</td> <td>Chr.1b Chr.4a</td> <td>17.1</td> <td>222</td> <td>278</td> <td>Tsukazaki et al. 2007</td>	ttt aat geaaactat gat gaggt at gt geat aggt gagacat ceaaa aa	segreettegreggratt caaaartaacategeraccaga	Chr.1b Chr.4a	17.1	222	278	Tsukazaki et al. 2007
AFA00E00AB504300(CA)>Bag(AU)>tatactitaagticagto gegenerget gegenerget gegenerget gegenerget gegenerget ceta(AC)5ta(AC)10actactoctactigageactactactoctagt acgenerget gegenerget gegenerget gegenerget gegenerget gegenerget gegenerget gegenerget at gegent angevenerget actactgegenerget gegenerget geg	द्धित्वा वष्ट्रधा प्रवह्नतत्वा एए ववव	caaaaiiaacaaciicgciaccaga	Chr.4a	5	110	224	Tsukazaki et al. 2011
AFA06F05AB795044(TA)9(TG)7tata(TG)5ta(TG)5ta(TG)5 acgarctscctacttgageacgggaagntccttagggacgarctagggacgarctagggacgarctagggacgarctagggacgarctagggacgarctagggacgarctagggacgarctaggargarctagggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargarctaggargargarggargargarggargarggargarggargar				01./	CC7	507	I Sukazaki et al. 2011
AFA06H12AB499380(GT)14ttattgtctccatgtatg(TA)4tggacattgactttggatgtagggacattggattagcAFA07A08AB499381(TG)8tt(TG)5tctggaggcgtttaggcattgacacatagaacagaacatagcgargAFA07A10AB499382(AC)11(AT)4tgtatg(TA)7tctggaggcgtttaggcattgacacatagaacagaacatagcgargAFA07A10AB499382(AC)11(AT)4tgtatg(TA)7actactgggargaatgccaaggagacacatagaacagaacatagcgargAFA07B06AB795046(AT)3tca(TA)3gcacaa(AC)11atatacattactgggargaatgccaaggataccagtfgcacatgccatggargAFA07C08AB499383(TG)4(TA)8tctc(TG)9tggcttatagcgattaggtggttcgggcattttcgggAFA07E07AB564811(TG)10gaggtgcccaaggatcattggtggttcgggcatattggAFA07E07AB564813(TG)4(TA)8tctc(TG)9tgggtaacaggacattggtggttcgggcattttgggAFA07E07AB564813(TG)4(TA)8tctc(TG)9gaggtgcccattttggcacttggttcgggcattttgggAFA07E07AB564813(TG)4(TA)8tctc(TG)9gaggtgcccattttggcacttggttcggccattttgggAFA07E07AB564813(TG)4(TA)8tctc(TG)9gaggtgccccattttggcacttggttcggccattttgggAFA07E04AB795047(AC)11atcccataacccggfggccccatggggggcccatttggatgcttcaaaggacgggggcccatttggAFA07E04AB795048(AC)8atcccataacgcccatggggggggggggggggggggggg	egat et ge eert actt gagg	gggaagat cct tagt gt acgca	Chr.1a	54.0	238	240	this study
AFA07A08AB499381(TG)8tt(TG)5tctggaagcggttaaggcattgcatccatatccatatccatatccatatAFA07A10AB499382(AC)11(AT)4tgtatg(TA)7actactgggatgaatgcccaaagaatccatftcaaagtAFA07B06AB795046(AT)3tca(TA)3gcacaa(AC)11atatacatggcttatagacgattaaggcccaaagaatccatftcgcatggattugAFA07B07AB499383(TG)4(TA)8tctc(TG)9tggcttatagacgattuggggtggttcggcatttttccttaggtAFA07E07AB564811(TG)10gaagtgcctccattttggacactggttcggaaaatccacggAFA07E01AB564813(TG)4(TG)13atcccatatccaaacatggttaaggcggtactattgggAFA07E01AB564813(TG)10gaagtgcctccattttggcacacggtcaaagggaaatccacggfAFA07E01AB564813(TG)10gaggtgcctccattttggcacacggtcaaagggaaatccacggfAFA07E01AB564813(TG)10gaggtgccccattttggcacacggtcaaagggaaatccacggfAFA07E01AB564813(TG)10gagatgcctccattttggcacacggtcaaaggaaatccacggfAFA07E01AB564813(TG)10gagatgcctccattttggcacacggtcaaaggaaatccacgfAFA07E01AB564813(TG)10gagatgcctccatttggcacacggtcaaaggaaatccacgfAFA07E02AB707H04AB705048(AC)11atcccataacggtggrgcctccatttggAFA07E03AB707H04AB705048(AC)13atcccataacggtgggggggggggggggggggggggggg	gacatt gact tt tggat gt tt agc	acacat agaaacagaacatagcgaaga	Chr.1b	88.5	274	266	Tsukazaki et al. 2007
AFA07A10AB499382(AC)11(AT)4tgtatg(TA)7actactgggagagaggcccaaagaatccagtgcacatgacttttgAFA07B06AB795046(AT)3tca(TA)3gcacaa(AC)11atatacatggcttatagacgcaatggtggttcggcacttttccttaggtAFA07C08AB499383(TG)4(TA)8tctc(TG)9tgggttaaggcggattttcgtgtggttcggcatttttccttaggtAFA07E07AB564811(TG)10gaggtgcctccattttggacatccggtcggaaagggggacaacgggggacaacggggggacaacgggggg	t ggaaggeggt tt aagget at t	at ccat at ccat at ccacgat	Chr.5b	158.9	251	243	Tsukazaki et al. 2007
AFA07B06AB795046(AT)3tca(TA)3gcacaa(AC)11atatacatgecttatagacgratttcctgtggtcgrgcanttttccttaggtAFA07C08AB499383(TG)4(TA)8tctc(TG)9tg getaaagccaggtacattgggtggtcaaaggacaattgggAFA07E07AB564811(TG)10gaagtgectcaattttggacatccggtcaatagggaaaatccaacgcAFA07E07AB564813(TG)10gaggtgectcaattttggacatccggtcaatagggagacaAFA07F11AB564813(TG)10gaggtgectcaatttgggttgatgtgggcaatttgggAFA07F04AB795047(AC)11atctccataacggttccaaaacaatctccataacggtgggcgactcattggAFA07H04AB795048(AC)8atctccataacggttccaaaacagttcaagtagggggggggggggggggggggggggggggg	stact gggat gaat gcccaaa ga	scagt tgcacat gact tt tg	Chr.4a	174.7	228+230	228	Tsukazaki et al. 2007
AFA07C08AB499383(TG)4(TA)8tctc(TG)9tggtaaagcaggataattgagtgfttaaaggaagacaAFA07E07AB564811(TG)10gaagtgctccatttggacattcggtcaataggaaaatccaacgtcAFA07F11AB564813(TA)4(TG)13gagggcccattttggacattcggtcaatagggaaaatccaacgtcAFA07F11AB564813(TA)4(TG)13gtggggaacaattttggacattcggtcaataggggaaaatccaacgtcAFA07F11AB564813(TA)4(TG)13gtggggaacaattttggacattgtcaataggggggcccatttggAFA07F04AB795047(AC)11atctccataaccgcttccaaaacaacccctagggggggcccatttggAFA07H04AB795048(AC)8gcaaaatcaaggatggccagcgttcaagtcatggaAFA07H04AB795048(AC)8gcaaaatcaaggatggccagcgttcaagtcatggaAFA08H07AB499391(TG)13atcccatggcatcattggacggttcaagtcattgga	ggettatagaegeattteegtg tg	gttcgagcattttttccttaggt	Chr.7b	0.0	275	273	Tsukazaki et al. 2015
AFA07E07AB564811(TG)10gaagngcotccatttigacatecggcaatagegaaaatccaacgtAFA07F11AB564813(TA)4(TG)13gtggggaacaatttitgaatgggttigatgtggccactttiggAFA07G04AB795047(AC)11atctccataaccgttccaaaacaacccctaggggggrctattggaAFA07H04AB795048(AC)8gcaaaaatcaagaatgggfccaagggtttcaagacgggggrctattggaAFA07H04AB795048(AC)8gcaaaaatcaaggatgggccaagggtttcaagtcatggaAFA07H04AB795048(AC)8atcccataggatgggccaagggtttcaagtcatggaAFA07H04AB79391(TG)13ataccatggcatccattgaactccctatccatttcaagatggc	t gct aaagccaggat catt gag tg	ctttcaaaagtaaaggggacca	Chr.2a	109.2	244	242	Tsukazaki et al. 2008
AFA07F11AB564813(TA)4(TG)13grg ggg aacaattittgaatggtitgatgt ggccacttitgAFA07G04AB795047(AC)11atciccataaccgcticcaaaacaacccctaggggggccacttiggaAFA07H04AB795048(AC)8gcaaaaatcaaggatggg caagggtitcaagcat ggraAFA07H04AB79391(TG)13ataccatgggatgcrattgatgtitcaagtat ggraAFA08E07AB499391(TG)13ataccatggcatccattgaactccctatccattcaagatgg	agat goot coat tt t gacat co	t caat agegaaaat ceaae gte	Chr.3a	196.2	269	265	Tsukazaki et al. 2011
AFA07G04AB795047(AC)11atcccataaccgcttccataacaacccctagggggctcattggaAFA07H04AB795048(AC)8gcaaaaatcaaggatgggcaagggtttcaagf cat ggcgg gggAFA08E07AB499391(TG)13ataccatggcattccatt gaactccctatccatttcaagatggg	gf gggf aacaatt tt t gaat ggg	gatgtt gtgccactttt gg	Chr.1b	90.4	216	239	Tsukazaki et al. 2011
AFA07H04 AB795048 (AC)8 gcaaaaatcaagaat ggd caago gtttcaagt cat gcacg gtttc   AFA08E07 AB499391 (TG)13 ataccatggcattccatt gaact ccctatccactttcaagatgacg	ct ccat aacc gctt ccaaaaca aco	ccctagggaggactcatt ggta	Chr.7b	54.9	271	265	this study
AFA08E07 AB499391 (TG)13 ataccatgcattccattgaact ccctatccacttccagatgacg	aaaaat caa gaat gggt caagc	tt caagt cat gcacgt gt tt cc	Chr.7a	86.9	127	127	this study
	accatggcattccatt gaacct cc	cctatccactttcaagatgacg	Chr.3b	124.9	223	233	Tsukazaki et al. 2008
AFA08E10 AB499392 (TA)3atgc(TA)3c(GT)11 gtacgatgegecaacetacteta aggggeaacgetaaataacceat	acgatgegeecaacetaeteta	gaggcaac gctaaat aacccat	Chr.3a	88.3	277	281	Tsukazaki et al. 2008
AFA08F05 AB795049 (GT)7ataca(TA)5(TG)9 gattcaaaegcaacttccagatcc cacagacatgcaccaatcttg	itt caaac gcaact t ccagat cc ca	cagacat gcacaccaat ctttg	Chr.6a	412.9	278	284	Tsukazaki et al. 2015
AFA09A11 AB499396 (AC)3(AT)3acat(AC)17at agaactctctgttcgctcattcca gcgagacttatatcgcctcgctta (AC)3g(TACA)2(TA)3	gact ct ct gt tc gct catt cca gc;	gagact tat at cgcct cgct ta	Chr.6a	254.1	180	184	Tsukazaki et al. 2008
AFA09C08 AB499397 (TA)4c(AT)4(GT)10a(TG)4c(GT)6at gaaaccgataatgacttaattgcttgaaaa ccttatcttaatctgcttccagtctccaa a(TG)9	aaccgataatgacttaattgcttgaaaa ccl	ttatettaatet getteeagteteteaa	Chr.5b	202.8	190	194	Tsukazaki et al. 2007
AFA 09F08 AB499400 (AC)3g(CA)12t(AC)3aa(AC)8 gggaacat gg caat aagacattg aat aat gg caacaaaaat taa cg fatta	gaacat gt gcaat aagacat t g	taat gtgcaacaaaaattaacggtatta	Chr.3a	360.9	248	258	Tsukazaki et al. 2008
AFA10B05 AB795050 (CG)5(TG)12 ggaaaaagccgtcgaaaataaagc tagatggtcgtcctatcctgtc	aaaaagccgt cgaaaataaagc tag	gat ggtcgctcctatcctgctc	Chr.3a	217.7	259	261	this study
AFA10B06 AB499402 (AC)15(TC)6 ccgcaaactggactaactgtacaaa ggcagaatagcccaagtgtttca	gcaaact ggactaact gtacaaa	cagaat agcccaagt gt ttt ca	Chr.la	281.5	281	302	Tsukazaki et al. 2007

(continued)
9.
Table

Marker	GenBank		Forward primer	Reverse primer	KiC 1	nap	Fragments Fragme	ient p) in	
name	Accn.	Core repeats	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	Ki	С	ui najoten III
AFA10C07	AB499403	(AC)22(AT)7	caacat t gcaaacat gt cacaca	gcact gagt tt gcgtt gagct a	Chr.5b	27.5	255	247	Tsukazaki et al. 2007
AFA10G05	AB795051	(AC)12(AT)9	c cagact aaaa ggaacc caact gc	tggtttgaatgtggtcagtggtct	Chr.5a	121.5	186	194	this study
AFA11B05	AB499408	(TA)6(TG)16ttgtggattggc(AT)3	ctcggtccaactttggagttatcg	tatggatat aacggggact ggct g	Chr.3a	234.4	250	254	Tsukazaki et al. 2008
AFA11C02	AB795052	(TC)3cttctcaataataagctatctgctta(CA )3(TG)6(GT)7	aataggcacttgtccaggaagcag	gcatt caat catgtgctt gtct ca	Chr.2a	172.3	225	223	this study
AFA11C09	AB564822	(TG)11tttgcgtgcgcg(TG)3catgcgtgc c(TA)3ca(TA)3	cagaacaacgitti ggi gaci gga	gct gtt agat ct gcct tct ccct g	Chr.7a	73.8	283+291	285	Tsukazaki et al. 2011
AFA11D02	AB564823	(A T)4gcatgtatg(TA)6cacatatc(TG)9 tt(TG)4tcc(GT)3	c ggcaaacct gtt aaat cc	tttatacataaacacaaacacacacg	Chr.1b	84.5	165	169	Tsukazaki et al. 2007
AFA11E06	AB564825	(A T)3t(AC)6at(AC)4ttataccctatttcc ttga(AC)13	gaaaaggtttttcaaaggagatttttca	gacttccacgggcaaggat	Chr. la	59.3	246	242	Tsukazaki et al. 2011
AFA11E12	AB499410	(AC)5tcttc(AC)12	gctggacggactt ctgtat gcttt	cgacctt aagt cat aaacgt ggt aa	Chr.8a	130.6	220	208	Tsukazaki et al. 2008
AFA11G05	AB795053	(TA)5(TG)11	ggt ca ggag ca a c c ca a a c a t a c	tectttaccaccatacctgattttg	Chr.5a	0.0	237	239	this study
AFA13E04	AB795054	(TA)6tgtggg(TG)12	agccaggt caaaat gtcggataac	gggcaaact tet gt tgt gt aacce	Chr.3b	20.7	220	208	this study
AFA13G12	AB795055	(AC)12	ggattt cctt aggcgatttgatga	agt cgt tacaact ccat c gct gt g	Chr.3a	285.4	259	261	Tsukazaki et al. 2015
AFA14A02	AB499412	(TG)12(TA)3	aaagt ggacgt aggcacat tot gg	tagccgt gggttt gaagagaaaaa	Chr.1b	69.7	244	246	Tsukazaki et al. 2007
AFA14A11	AB795056	(AC)10	ct cga cggggt gct agt aaaga ga	tt gact cca cat ccacgt acga at	Chr.2a	167.6	145	143	this study
AFA14B03	AB564831	(AT)3(AC)17	actgcatctttccaggtttgctc	tacatcgcacgtgccaaaatactc	Chr.6a	305.6	227	215	Tsukazaki et al. 2011
AFA14E04	AB795057	(AC)16	gaattttcgccatatcaaccatc	cgt at gagagcat actt gtt gt att t ga	Chr.8a	102.8	208	218	this study
AFA14E07	AB795058	(TG)7ta(TG)7(TA)6	ga agcaac acaact t caaaac gga	gacgaaatttccttcttcatccga	Chr.8a	103.4	126	123	this study
AFA15D05	AB564835	(TG)10	gcattt ggagt agt gggct taa agg	acagt cagaat cgcat cgat aaca	Chr.7b	65.0	213	217	Tsukazaki et al. 2011
AFA15E08	AB499418	(GT)8attg(TA)3	t gagaagt gt gt aaggcaa ggc	gccccaaagt cat act gct ggt ag	Chr.7b	152.2	241	245	Tsukazaki et al. 2008
AFA15G01	AB795060	(TG)13	ctagt gagaccat gccgcttagtg	aaccagggat ggaaaatccaagat	Chr.6a	272.8	202	210	Tsukazaki et al. 2015
AFA15H06	AB564837	(TG)12	ccaat actgatt aagcaact t gt gt tat gaa	t cgt cat ca agt agg cat tt ct ca aa	Chr.3a	176.5	173	170	Tsukazaki et al. 2011
AFA16B11	AB795061	(TG)13	ctttgtttggaatggtggtgatga	gccgt gat tt t ct cc tat caat cg	Chr.1b	133.9	293	286	Tsukazaki et al. 2015
AFA16C06	AB499420	(TG)18	ggagct ct ggaacgt t cgaact aa	at tgt caaaat caagccac ggaag	Chr.3b	73.9	254	256	Tsukazaki et al. 2008
AFA16D05	AB499421	(AC)12(AT)5	aatt cacccataactt ccgctacg	gt gcaacactactgacct cgcatt	Chr.7a	60.1	204	188	Tsukazaki et al. 2008
AFA16E07	AB564840	(GT)3(ATGT)2a(TG)6tatg(TATATG )2(TG)4tc(TA)4tccg(TG)3caga(TG)4	t at gegt agacat gagggeacaaat	cacaaacat gagaaata cgtt ggca	Chr.3a	343.9	215	213	Tsukazaki et al. 2007
AFA16E11	AB564840	(TG)3tatgtatgta(TG)6tatgtatatgtata( TG)5tc(TA)4tccg(TG)3caga(TG)4	t at gogt agacat gagggcacaat	cacaaacat gagaaata cgtt ggca	Chr.5b	0.99	203	205	Tsukazaki et al. 2007
AFA17A05	AB499422	(TG)22	ttccat caatt acact gc cagcac	tgctgaacttggtatgttgatactcca	Chr.8b	40.3	181	161	Tsukazaki et al. 2007
AFA17A12	AB564843	(GT)3atatacatg(GTTT)2gtgttgca(TA )7(TG)9(CG)3tgcg(TG)9tt(TG)4	ggcagcagt gt t ct ggggt tat ac	agoot gagoacat caaat acaoga	Chr.8a	19.0	264	268	Tsukazaki et al. 2011

Table 9. (continued)

Table 9. (co	ntinued)									I
Marker	GenBank	Poero exercición	Forward primer	Reverse primer	KiC	map	Fragn sizes (1	nent op) in	Downwood	
name	Accn.	Core repeats	sequence (5'-3')	sequence $(5-3)$	Linkage group	position (cM)	Ki.	С	III naundau	
AFA17B01	AB564844	(TG)3ccatgg(T)6gtatggtgggggttaat(A C)6ag(AC)14tc(AC)3	t gt at ggt ccaggt tt cat t gt gc	caa gcact gcaga gagagt tcagc	Chr.7a	105.3	175	177	Tsukazaki et al. 2011	I
AFA18A02	AB564850	(AC)10(AT)3	a agget acgea a gaa at cac caa	acctgcacct gtttgctt gagtta	Chr.7b	79.6	186	190	Tsukazaki et al. 2011	
AFA18A08	AB795062	(AC)9	ggtt gt att ggcaa att gcact ga	ttgaatttctaccaacctgtatccc	Chr.2b	51.0	206	204	this study	
AFA18C12	AB564851	(AC)15	gaacgattccaaaggcacgataag	cat ggct gagcactt gagaggt ta	Chr.2b	0.0	222	214	Tsukazaki et al. 2011	
AFA18F04	AB564852	(TA)5(TG)12	atccaaggttt gtgaccaattt	acaccaaccgagcat at aat caa	Chr.5b	188.0	249	237	Tsukazaki et al. 2011	
AFA18G02	AB564853	(TA)3gtgcatatcaaagtgcagacatgaatat (AC)11	t gaaacaaca gtt gcaaaccagg	tittt ccct ctat gact t cggcac	Chr.5a	28.4	292	294	Tsukazaki et al. 2011	
AFA18H08	AB795063	(TG)13	t cgt gt ct gt at ct gt gt ccgt gt	gt tt at cctt ct ggt gaaat gccc	Chr.2a	0.0	290	286	this study	
AFA23A08	AB564855	(TG)11	a gaat cagagt ct t ggtt ggt gcc	at tct aaaccatggtct gt cggga	Chr.7b	94.7	252	248	Tsukazaki et al. 2011	
AFA23C12	AB499427	(AC)20	aaact gatcaaaat gt gccccact	gagaatccgattaaaaacgagcaag	Chr.1a	97.9	154	134	Tsukazaki et al. 2008	
AFA23D11	AB564858	(TA)6(TG)13	caatcacacagcattgagcatc	ct gat tt gtt cagt tgat gatt gcg	Chr.5b	175.3	291	279	Tsukazaki et al. 2011	
AFA23G01	AB564861	(TA)8(TG)14cgc(AT)3gtagatcatatgt atg(TA)3tggatgtttatt(TG)4	catcatttccatcacctccatgtc	aagggctaaggtttctcttcctaca	Chr.1b	33.8	254	244	Tsukazaki et al. 2011	
AFA23H11	AB564862	(TA)3catatgtatgtacgtgta(TG)3tatgta a(TA)6(TG)17	calgaccacett1gtcgcatt1t	t ctcgtt ctgaatt caaccccaat	Chr.2b	6.99	237	231	Tsukazaki et al. 2007	
AFAA00B01	AB795064	(AC)13	at ggt gt aagggcatt tt gt tt gg	gat gggaaaagat gt caagaa gca	Chr.3a	161.8	172	174	this study	
AFAA01A08	AB795065	(CA)7(TA)5cacaa(AC)4	catt gcaacttt gt tt at at gt ct t gt gc	cccctt gtccacct ctcaat gt	Chr.2b	39.4	260	256	this study	
AFAA01G08	AB795066	(TG)4tctatg(CA)7t(AC)6aaa(TA)8t( AC)7	ttgtgtcaccggttttgttttatg	agccaaccccgtattcgtaatgt	Chr.4b	87.5	238	226	Tsukazaki et al. 2015	
AFAA02D08	AB499432	(TG)10(TA)6tgtatgtatgcatgta(TG)7	ctat ggccaagt tccatttt gagc	t att tt tac agcaac ggagca gca	Chr.1a	86.7	181	169	Tsukazaki et al. 2007	
AFAA03B07	AB564866	(ATT)3gttgtt(TA)7(TG)12	a a a ga a ga a ga a a a t g ct a c a a c c g t a	ttgaatttcccaacagattaatatcattacaa	Chr.6a	162.0	186	204	Tsukazaki et al. 2011	
AFAA03C05	AB564867	(AG)12	ttgggaaatttgggaattagggtc	caccataaatcgttttgcttacacc	Chr.4b	31.5	201	189	Tsukazaki et al. 2011	
AFAA04D06	AB499435	(AC)4(A)5tattaatacaaacc(AC)4agat(	ttattagccatgatcaacacacacaaaaa	agatt caagcagagt caaggt at t caatg	Chr.3a	65.9	289	275	Tsukazaki et al. 2008	
A E A 01E01		AC)4gcatgc(AC)4tcat(AC)14(AT)11	t concret of consold south on cons	t among as a tt of t tt as a access	-17 10	000		020	بالمنام منابر	
AFAA04F04	AB/9000	(AC)10(A1)9	i ggyrii yi gaagi aagi i gagca	rggggargaanten aagggg	CIII.40	0.0	707	017	T15 Study	
AFAA04FU3 AFAA04F12	AB5048/0 AB795068	(A U) (A U) (A U) (A U) (A U) (A T) 3 gattat (A C) 5 gagaga (A C) 7 cctc (T	er georga a a a a can construction of the second	agi gaagan nguch gan ngg t gact teacaaat gacae gaat gt g	Chr.3a	09.4 263.2	280 280	284 284	1 Sukazaki et al. 2011 this study	
		G)8tttgta(TG)3attgtgca(TG)5cgtgtt( TG)5attgtgcgtatgcgta(TG)4							'n	
AFAA04H07	AB564872	(TG)20	t gcacctct acaagtcaact gggt	at gtt gt ccaagcactt tgatgga	Chr.2a	182.8	269	261	Tsukazaki et al. 2011	
AFAA05B02	AB499436	(AC)10(AT)11(AC)8(AT)3acacaa(A C)9(AT)9acg(TA)5agag(TG)4	cgtaaagttat ccatacaatgccca	ggaccagttatccatacgtacaca	Chr.4b	73.9	296	288	Tsukazaki et al. 2008	

	GenBank		Forward primer	Reverse primer	<b>KiC</b>	nap	Fragn sizes (h	nent bp) in	
	Accn.	Core repeats	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	Ki	С	reported III
11	AB564874	(A T)3gcatacaaat(AC)4aa(A C)9ttata tgta(TG)3	gcaaacccaacat at acat gaacac	tttactatgcatggccgattaccc	Chr.4a	39.6	178	210	Tsukazaki et al. 2011
202	AB795069	(A T)3gta(TG)7tt(TA)3aa(TG)5(TA) 5tgtatatgtata(TG)5	aataaatcatccaccttccatgcg	tttctgcttcggtgctaactgttt	Chr.3a	127.0	242	244	this study
312	AB564877	(A T)4(A C)12	caacgttt ctt gt at gt ca gct catt c	gt gacat cgcctt agact t gat cg	Chr.5b	42.8	216	218	Tsukazaki et al. 2011
H03	AB564878	(TA)3c(AT)3gta(TG)10(TA)8(TG)4t a(TG)3ta(TG)6(TA)7	gaacacatect at tgaacet geat e	gt gatt gt tt tagcaat ggt aaggg	Chr.3b	74.5	225	243	Tsukazaki et al. 2011
H11 A01	AB564879 AB795070	(TA)3c(GT)4tagtatgc(AT)9(AC)12 (AT)3gtgtatatct(AC)13(AT)10tgaga	gaaacgttgtggcctctttagcat ttcataacataacgtgataacatattgaaa	tcatgttctcccttcatttcgtga taaacatgggcatataattagcatt	Chr.5b Chr.2a	107.9 214.8	238 225	236 229	Tsukazaki et al. 2011 Tsukazaki et al. 2015
		tttatatctat(TG)3tatc(TA)3							
C01	AB564881	(AC)5aaaacatacat(AC)11aa(AC)3( AT)3	t ggt act cact gatt t aat ggcat aat	cttt aat gatgaatct gttgcat acg	Chr.5a	18.8	276	280	Tsukazaki et al. 2011
B01	AB564883	(AC)11	gggcccaact gaacaga gt catag	cccaact cact ccagaa cact a caa	Chr.3a	292.4	224	232	Tsukazaki et al. 2011
B04	AB564884	(TC)6(CA)6at(AC)4attatacaccc(AC) 7(AT)7(AC)8at(AC)3atgcgtgca(TA) 4gatgtatatgtgta(TG)3tatgt(AC)3atat ana/TA13caccca(AC)4	caacttgaacatgatttggaagtattta	ម្លេជ្ញ ជា យុទ្ធា អ្ន ពួល ខេស្តា ព	Chr.4b	6.66	253	255	Tsukazaki et al. 2011
C04	AB564885	(CA)9aaacacagaagacacag(AC)3aaa cacagaag(AC)3ag(AC)5atgta(TG)7	ttttgcgagattctgcttcttt	ct gaaaagigi tai gaacti gaggiai t	Chr.3a	331.6	245	237	Tsukazaki et al. 2011
E06	AB564887	(TG)6tt(TG)13	at gt at catt gact gt att t gcacct t ct aa	at caacat cagt cttt ta a attgcat cagta	Chr.6a	352.0	216	239	Tsukazaki et al. 2011
H05	AB795071	(TA)3tgtgtctcagaatt(TA)3tgtgggtga a(TG)7tt(TG)3	tggcagfgtttagfgttgggttga	gcacgtaa gaaaaa ccacac ccat	Chr.6a	316.6	174	172	this study
H07	AB795072	(AC)14	cccacaacatacacacacat ctt ca	t ctt cct cct cct gctt at cct cc	Chr.2b	43.9	180	182	this study
305	AB499440	(A T)3ctatgctaaccttctctaaagtacaac(T A)6tttcgtgtatc(TG)12	tgt ccact att cacggct tact ca	gat gggt ctgcctcctcctt att t	Chr.2a	117.8	202	206	Tsukazaki et al. 2007
307	AB795073	(AC)7gc(AC)6(ATAC)4ct(AC)9a(T A)8	cctatact gt cat ccaatagt gaacc	attagtaatggacgggcctaggga	Chr.2a	197.8	260	252	Tsukazaki et al. 2015
B08	AB795074	(TG)6	t gt t ggt at gt gt t cat gcaaggt t	gaca ccca cca ca ca ca ca ca ca ca ca ca	Chr.8b	0.0	252	250	this study
H10	AB564890	(GT)8(T)7aaaatgagtga(GT)9(A T)4c( TG)3(TA)3tgta(TGTGTA)2(TG)6	agcacacaaat actaagagaaaaat cagca	agaaaatttcatcttttagcaacactttga	Chr.2a	35.1	255	257	Tsukazaki et al. 2011
-03	AB499443	(AC)11at(AG)9aatt(AC)3gcg(CA)3t gcatgcacg(CA)3att(AC)6	t gacatagacccttttgt aggaggaaa	tgcat acat acat acat acgcatat acat acaa	Chr.8b	26.7	210	200	Tsukazaki et al. 2008
405	AB564893	(AC)4aa(AC)7atatgcttatattg(GT)3	t agca gaagcett gtagegagatag	gaat gcaataaaatgacacat caagca	Chr.2a	139.3	154	150	Tsukazaki et al. 2007
<b>A</b> 03	AB564893	(AC)4aa(AC)7atatgcttatattg(GT)3	tagcagaagcctt gtagcgaggatag	gaat gcaataaaatgacacat caagca	Chr.4b	116.1	238	236	Tsukazaki et al. 2007
311	AB795393	(AC)8atctacacgtgtatgtatgt(AC)7	gcatcgggt cagacgctagtattt	caagaagacaaaaggat gactgaag	Chr.3b	64.2	221	217	Tsukazaki et al. 2007

(continued)
9.
Table

Marbar	GanBank		Forward primar	Pavarca nrimar	KiC	map	Fragn sizes (h	ient in in		
name	Accn.	Core repeats	sequence (5 <sup>-3</sup> )	sequence (5'-3')	Linkage group	position (cM)	Ki	C	Reported in	
AFAT02C04	AB696898	(A C)8ttc(T)5aagc(TG)3tattatttcttcat gtgaccat(TC)3	ctt gocttt tctt cgttttt	tt gcat gatt ggctaa agg	Chr.4a	20.8	187	184	Tsukazaki et al. 2012	
AFAT02C08	AB564894	(TC)13	ctaacaaccaatcaaaactaagataaca	acat aat geget gaett gga	Chr.1a	30.0	225	239	Tsukazaki et al. 2011	
AFAT02F09	AB795075	(TG)4cata(TG)11(TA)4ga(TA)3gata TG)7	(tgaaacatgaagcettecaggata	ggttgtatgtcaggaattgtctggg	Chr.1a	247.5	135	137	this study	
AFAT02H05	AB499445	(AC)12tg(AT)4	tctcgtaccagtttctcctttttcc	gattc ggatcaat acagggaat gc	Chr.4a	95.0	292	294	Tsukazaki et al. 2008	
AFAT03F10	AB795077	(TG)3tatctatgtatgta(TG)9	t ct caaat t ccct cct agat ccca	cat ggaggagt agcagatcgaaca	Chr.2b	45.9	227	229	this study	
AFAT03H10	AB564899	(AT)4ttatgagagtctt(AC)5atgcat(AC 4atgcgta(TG)3tatg(TA)3cag(AC)11( AT)7	) gcacttcctacttgattcacacaagc	tt gattt cat ggt gtataaact ggt aat ctt	Chr. la	118.5	194	170	Tsukazaki et al. 2007	
AFAT04B03	AB499448	(TA)7(TG)22	tttttcatttcagcataaagtaattttcta	at aacaacggtaaccaacgctctaa	Chr.2a	147.3	245	231	Tsukazaki et al. 2008	
AFAT04D05	AB564903	(TG)15atattgcattactgatatgcaaa(TA) 3agttctc(TA)3	gtt gct gattt cttt tct cattt tcc agt	tggtgcaaattctaaattttcctttgat	Chr.3a	313.6	195	193	Tsukazaki et al. 2011	
AFAT04F03	AB564904	(CA)8tattat(AC)5gc(AC)3t(CA)3tgc g(CA)3tacatt(CATA)3gat(AG)3actg ttgccgaactaacaatt(TC)4	caaaaggi gagaaagggctacatga	gagagagaatt gtt agt teggeaaca	Chr.1a	80.2	161	165	Tsukazaki et al. 2007	
AFAT04H04	AB564906	(ACII0(AT)5	cagaat t ga gccacggaagt tt	ct ctt gct agggt caaactaacctct	Chr.4b	55.2	193	185	Tsukazaki et al. 2011	
AFAT05B10	AB795078	(AC)3caatgttgatctatctactcaaat(AC)1 0at(AC)12	cctcattttgfcctattgctcattc	gegeccaact cattagagatagg	Chr.3b	35.2	258	256	Tsukazaki et al. 2015	
AFAT05H05	AB499458	(A C)11(A T)6(A G)3aaatga(T)6aaaat atag(A T)3	ttgttccgcaaatagttaaactcaa	tgttcgaatcttacgtaacgtgatt	Chr.4a	103.4	117+131	131	Tsukazaki et al. 2008	
AFAT06C11	AB564912	(AC)23(AT)4	ccacaaactacagcct gaaat gga	gat acaaa gacgt gaga cgt ggga	Chr.5b	143.1	138	146	Tsukazaki et al. 2011	
AFAT06C12	AB564913	(TA)6(TG)15tt(TG)6	gc gct taagcaccat tacacaa ga	cccatgat cttccaaaat ctagaac	Chr.3a	377.9	260	266	Tsukazaki et al. 2011	
AFAT09B02	AB564918	(AC)8gtgtaca(TA)3	caaagggcaat ccaat ggt agaga	gcgatagttcttcgttattttgggc	Chr.7a	78.1	268	272	Tsukazaki et al. 2011	
AFAT09B05	AB564919	(GT)3acacata(TG)3acta(AG)13gcgtt1 taaacc(AT)3	t aggeggggttatttgactacatga	aat cacaataggaaaccaaccgga	Chr.5b	132.1	214	210	Tsukazaki et al. 2011	
AFAT09B10	A B564920	(AT)7(AC)10	gctaaagtatagggcgcgactcaa	tcccattagactt cat caacatt gc	Chr 5a	24.9	292	304	Tsukazaki et al 2011	
AFAT09E05	AB499461	(TG)11	aactgegagtaatttctcaacaactge	ggaggaaaat ct ctactcccagcatc	Chr.4b	108.4	255	257	Tsukazaki et al. 2008	
AFAT09F07	AB795079	(AC)8	gcacaaatttcctcatccct gttc	t gt gc ag caa g c gt aat ct a g g	Chr.3a	358.2	267	265	this study	
AFAT09G02	AB564921	(A G)22tgtt(TA )4tgtatgcatacgtgc(A T)3gtatgga(TG)4	cgccattttcttcttcttctccctgta	aaaatt ggt ccaagccaaggaact	Chr.3a	34.0	260	266	Tsukazaki et al. 2011	
AFAT09G03	AB795080	(CA)7taaatacat(AC)8at(AC)7(AT)3	ctgccatcctctttctctttcaa	cttac caat ctat gcca gcaag cc	Chr.6a	88.2	247	249	this study	
AFAT10D10	AB564924	(GT)7attta(TG)3	agtgatt gaaagteet agggaggt	tt cat tt acaaa ccat gccaca cg	Chr.1a	162.4	220	218	Tsukazaki et al. 2011	
AFAT11A02	AB564928	(TA)3tc(TG)3tatgc(AT)3(GC)6(AC)	gc gcgcacacacagagat ac	tccgct ctttt at ccttat atttctaattggtt	Chr.6a	59.4	165	158	Tsukazaki et al. 2011	
		4agagatacgtatgttgacatgtg(AT)4cta( TG)4tt(TG)8								

Table 9. (continued)

1 able 9. (co	ontinued)									
Marker	GenBank		Forward nrimer	Reverse primer	KiC	nap	Fragn sizes (b	nent DD) in		
name	Accn.	Core repeats	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	K	C	Reported in	
AFAT11A03	AB499464	(AT)3gtatgtaaacttgttatt(TA)3aatatc aaaaactagtaaaat(TA)4(TG)9tcttacat eraaaaatatcagcestif(TA)5	cttgcctatatgtattgaatgttgccaaa	cggaatattttggfgfgcatgtgf	Chr.3a	302.2	216	218	Tsukazaki et al. 2008	
AFAT11B03	AB564930	(AC)3agat(AC)9	t aat get egat gget geaaa	ttt gagat ggfgatttaataaaatatgi gac	Chr.6a	281.2	220	216	Tsukazaki et al. 2011	
AFAT13C12	AB795081	(AG)3taagagtattatctgtatgttaactca(T A)6tgcata(TG)7ggtgcttgctcta(TG)3t aatottfaaaaaacr(TG)5	ggggf at caaaccet t catt caaa	tgatttcccatacacatatgcactg	Chr.3a	0.0	283	285	this study	
AFB01B03	AB499467	(AG)11	gatt gcattcgatttcgtt gagt g	cagcacacttcgtttggatttcag	Chr.1a	185.5	194	220	Tsukazaki et al. 2007	
AFB01E09	AB795082	(TC)12	gc gggt aggt gaacgacaagt aag	tt et gt gaaggt et eaceact t eg	Chr.2a	177.0	275	295	Tsukazaki et al. 2015	
AFB02D05	AB564937	(CA)3(AG)3aaaac(GA)14	t at gtttgaatcacgctacgcaagt	tgacttttatgtatatcgttgggttagca	Chr.7a	45.9	251	253	Tsukazaki et al. 2011	
AFB03B05	AB564941	(TC)10cc(TC)11ggtata(ATTT)2ataat gttt(AAC)3	. getgatat ggat gaaageaaga gg	ct gctt cgagat gatt gcaaaaag	Chr.7a	0.0	269	275	Tsukazaki et al. 2011	
AFB03E02	AB795083	(A G)4at(A G)5gg(A G)5aaagagaa(A G )6	, ggotcaagaaagttttcacaggga	tcaacttgtcgtgctttcgtttgt	Chr.3a	276.1	278	276	this study	
A FR04A 09	A R499475	(GA )3aagaaga(A G)13	actagagt catt gega gae gae go	tatttt ett gage gegaggt cagea	Chr 3a	368 7	757	268	Tsukazaki et al 2007	
AFB04B12	AB499477	(AC)6(TC)15	ctagactt gaggggaggat gggtt	ggagaggggaacat gcatt tt ag	Chr.3b	101.4	235	233	Tsukazaki et al. 2007	
A FB04F07	AB795395	(TC)10	agcaat at gaaggagacacat gg	cccaaact aatt tagggt tt acaat tc	Chr.2a	164.1	152	154	Tsukazaki et al. 2007	
AFB04H04	AB499479	(TC)3ctcc(CT)3t(TC)7	cgcaacacaaaagataacgtctcg	agactt cgacgt aga gt cc gat gg	Chr.7a	93.9	293	289	Tsukazaki et al. 2008	
AFB05D11	AB499480	(TC)3cttatcccttctcaacgctgctt(TC)3ct	actitit gat a accca ccca gt cca	ggt tcc agcgat gac caat tt tt	Chr.6a	200.7	294	290	Tsukazaki et al. 2008	
		tettacatgacgtcgcc(TC)13ttetcacatgt								
	100400	atgctttg(C1)3		tt met and to be the second second second	د ج	1 100		0.00		
AFB0H09	AB499485	(AU)/ ((UA)	ge gagegi ai eecei aagaagaai	।। घुष्ट्रा एएए। बघुबा ।। एबए। घुटबटब	Chr.3a	585.4	7/0	708	I sukazaki et al. 2007	
AFB06C08	AB499484	(A G)3gg(A G)18	c gcacaaaacat ata gggaga ggg	tetec gtaa gaagae et geaet ca	Chr.1a	151.7	170	144	Tsukazaki et al. 2007	
AFB06E05	AB499485	(TC)13	ttt gaatt ggcagagacaat ggt g	t gt at ggaggaaggget gaa gag	Chr.1b	59.1	149	127	Tsukazaki et al. 2007	
AFB07E06	AB564953	(GA G)3aaagag(GA)3	t gcatt gaagaa gaggaa gaggaga	ggcaaatatcacgcttagaccagg	Chr.4a	0.0	267	247	Tsukazaki et al. 2011	
AFB07F07	AB795084	(AG)8ggatgatgctctca(AG)5	atcgtgacgtggagattgagagg	tt cat ct ct gaatat gagcgcagc	Chr.4a	131.2	137	132	this study	
AFB08D06	AB564956	(A G)18	gggt gt at gt gt acggct at gggt	ct ccgt ctcctacctcct ccaaat	Chr.6a	245.7	242	246	Tsukazaki et al. 2011	
AFB08G07	AB564958	(TC)7tt(TC)5	ctcttcctttt gacttcaacccca	attettegggat gaaat ggaggat	Chr.4a	81.4	230	232	Tsukazaki et al. 2011	
AFB09D04	AB696900	(A G)8	t gagat ggt gt ggt tt ggt a ggaa	aat ctt gattgccaacgctgat ct	Chr.2a	41.0	169	171	Tsukazaki et al. 2007	
A FB09G09	AB564960	(CT)13tatatgcatc(TAT)3	ccaagacaaaagatt cct atacaccc	aat cagacat cgaggat gcaaaca	Chr.6a	212.5	205	203	Tsukazaki et al. 2007	
A FB09H04	AB564961	(CA)8taa(TC)12	at at cacaagt a cgcact ccct ca	agt tatat tgggcggcagt gt t gt	Chr.7a	58.5	198	208	Tsukazaki et al. 2011	
AFB10D09	AB795396	(TC)9	ggtt cct tt t ggtt ct gt gt tt gc	ccat acagatt t cctt aagagacga	Chr.5b	122.8	301	303	Tsukazaki et al. 2007	
AFB11E02	AB564965	(AG)10	catt gagggaaaggt gcgaatt ac	tcatgcgtcatcccacttctgtat	Chr.6a	291.5	129	149	Tsukazaki et al. 2011	
AFB11E05	AB564966	(TC)3caactca(TA)3tcgatccc(TC)7gg	a a at a c c g a c c c t t c c t t c c t	ggac aggtt t ggagat gacgaagt	Chr.1a	92.3	301	297	Tsukazaki et al. 2011	
		cactataccgaaccc(TC)3caagtca(TA)3								
		ttgatccc(TC)7aacactacactgacc(TC)3								

(continued)
9.
Table

Marker	GenBank		Forward primer	Reverse primer	KiC	map	Fragn sizes (h	aent pp) in	
name	Accn.	COLORED	sequence (5'-3')	sequence (5-3')	Linkage group	position (cM)	Ki	С	
AFB11H08	AB564968	(TC)8cc(TC)19	gttgatcaaacacccatgataacag	t gtcttt aagt gat tcgtcggcaa	Chr.5b	94.5	261	257	Tsukazaki et al. 2011
AFB12A08	AB564970	(A G)15	caacactct aat tegaacaaaaat gaga	cat ccgaaat aaaat tgatagt ggaca	Chr.6b	0.0	154	174	Tsukazaki et al. 2011
AFB12H10	AB564974	(AC)5(GA)10gcacattcgt(TA)3	gatagggtaagagttgtcgccttaga	cgt gt gcaaagt gactt at at at tactt aaact	Chr.4b	55.2	151	143	Tsukazaki et al. 2011
AFB13G10	AB499488	(AG)17	aaaaggt aaaggaggc gt gt aggc	at ctgccgggatt gattt cctact	Chr.2b	59.2	240	236	Tsukazaki et al. 2008
AFB13H10	AB795085	(TC)9	catgttcattcacactttcgttcg	aagaaggca ggagt tggaaggagt	Chr.2a	24.3	200	194	Tsukazaki et al. 2015
AFB14D09	AB795086	(TC)8	ccggt cgt catt agt ta agc ct t g	at tet aact ggaget get ge gaet	Chr.3a	319.2	281	283	this study
AFB15D03	AB564977	(A G)8	ggtcctgaactctctccaaatcca	ct ccaat tccctt tcagt ccat tg	Chr.4a	74.0	226	222	Tsukazaki et al. 2011
AFB15G10	AB795089	(TC)3aaacgacaacttctttcctagccgtca(	aagat gtet gat egee gtt gt gt a	aggatct ccccaacacagtat gga	Chr.3a	256.0	197	199	Tsukazaki et al. 2015
		TC)7							
AFB17B09	AB564980	(A G)8	ctagt ggaa gaagt cgca ggcac	caagtaagcot of gocat of too caagtaag	Chr. la	73.8	127	125	Tsukazaki et al. 2007
AFB17G01	AB795397	(TC)7	gatt caagt gcccact tt t caa ca	gcagaagaagaat gaat gt ga gcc	Chr.7b	114.9	255	251	Tsukazaki et al. 2007
AFB18A09	AB795090	(A G)8	gctt aagcact gcat at cat ccgc	t ccacct tt t cagat gt ct ccct c	Chr.3b	79.2	291	299	this study
AFB18H05	AB795091	(A G)6	gttaaattggagagttggcgttcg	at ctt gct tgcct ccct ct at tcc	Chr.2b	21.1	227	229	this study
AFB19A01	AB564982	(A G)6(G)5(A G)7gtgtatagttaactgta(A	A at gt gt gt gt gt gt ac gaggt t	cccttgtttt aat tt acccacccc	Chr.8a	37.4	257	261	Tsukazaki et al. 2011
		G)4							
AFB19D01	AB564983	(GA) bagagaacgaaaggagaggagattatg (GA)3	g ttcaaggtaagcgctgcatcaata	cttttcctttccaaatttccagcc	Chr.4a	55.9	260	262	Tsukazaki et al. 2011
AFB19H03	AB696901	(TC)3ttt(CTC)3c(TC)11	caaggat cat gaccatt t ggaagg	gccgt gacagat acaaggct cttt	Chr.2a	50.2	285	287	Tsukazaki et al. 2012
A FB20E09	AB795092	(TC)14	aacacataccggaggcaacctaga	at gtat gctt cggatt gct gat ga	Chr.2a	84.2	280	260	Tsukazaki et al. 2015
AFB20F12	AB564984	(AG)11(TG)3	agtat ccac ccattt ggt tgt tgg	at gtttcaatacatccactccgc	Chr.7b	46.9	228	226	Tsukazaki et al. 2007
AFB20G03	AB696902	(A G)25(TG)5	t gct tctt gt cact ggt taggt at gc	cca gaagt cagat t cacaaacaga	Chr.2a	56.4	202	219	Tsukazaki et al. 2012
AFB21D05	AB564987	(A G)7	ttccattcaaatccatcaagca	ct cccaccaccaccatt	Chr.8a	46.2	183	187	Tsukazaki et al. 2011
AFB228	AB564988	(CT)17	cgaat ctaaccccaaacgag	acgacgat ggat ggaagg	Chr.6a	194.0	206	208	Tsukazaki et al. 2011
AFB22B04	AB795093	(A G)6	t ct tcgaggaga gt cat ggt aggg	aaatt gt gc cacgt cct caaaa gt	Chr.2b	34.0	269	271	Tsukazaki et al. 2015
AFB22C11	AB564990	(A G)7	t gaggtaaa ggat ggat gct gaca	t ccaacatttt ctttct cct ccca	Chr.4a	65.2	251	247	Tsukazaki et al. 2011
AFB22C12	AB795094	(TA)3caa(AT)3tatatgcaatc(AC)3cag	g cgcagaagctcaaaagtgtgaaa	cct ct ccaca agt t cat cggat ct	Chr.6a	261.2	227	241	this study
		catatecettactatettat(1C)14							
AFB237	AB499495	(ca)3tgt(ga)16	aggcgt agcat aagggt g	gt gt gc gt ga gt gact gt g	Chr.7a	25.8	144	120	Tsukazaki et al. 2008
AFB23C03	AB795095	(TC)3ctgcatctatcc(TC)10	aagat ggat agcggctt aaacaag	t caagcaaat at cgc gt ct gaa ac	Chr.6a	116.0	131	143	this study
AFB23G02	AB499496	(A G)13aa(AC)3	aagcaaagatcatagcccgagatg	gt tgcaaaggtaat cggt aacaaag	Chr.8a	42.5	219	225	Tsukazaki et al. 2008
AFB25D02	AB795096	(AG)15ggac(AG)4	ttcactactaaaaataaaataaaataaaaatccgaatag	a gccgcagctacaaacctatctc	Chr.2a	160.0	230	240	this study
AFB25G01	AB564995	(TC)20	t gagat gt t gagt ggagagt cgct	t gcaat gt caagt ccat gtt cact	Chr.8a	72.5	219	209	Tsukazaki et al. 2007
AFB277	AB499498	(CT)15	aggtggatgggtt at gt cc	t cctt tt cacact ctt t ct t c c	Chr.4b	117.5	103	109	Tsukazaki et al. 2008
AFC01E09	AB499499	(TCT)7	cccttttgttcgacaagatttcca	caagacagcgatgacaaggaagaa	Chr.7a	53.7	140	143	Tsukazaki et al. 2007
AFC01F02	AB795097	(AG)3g(GAA)8	atggaaatgatccatgaggagaa	ttggagggat cgcct cagt tacta	Chr.3a	206.7	249	239	this study
AFC02H05	AB564998	(GAT)4ttggatttg(GAT)3(GAA)8	gt cct ct at ca aggat t gggagca	gct ct tt ct caata gcat caaat gcc	Chr.5a	37.8	108	111	Tsukazaki et al. 2011

Table 9. (continued)

Marker	GenBank		Forward primer	Reverse primer	<b>KiC</b>	map	Fragr sizes (l	nent bp) in	-
name	Accn.	Core repeats	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	Ki	C	Reported in
AFC02H08	AB564999	(TC)3(TTC)3c(TCT)3ttcatctgt(TCT) 6ctc	ctt cgt agt ca cacct cct t cggt	gt gaaat toogagagaat gegat g	Chr.1a	67.0	154	160	Tsukazaki et al. 2011
AFC03A04	AB795098	(GAA)7ga(AG)3	aataaaccaaacatgtccaagccg	tttcgctgcgattgtgagattcta	Chr.7b	88.5	145	141	this study
AFC03B02	AB795099	(TCT)6atttt(CTC)3	ategt ggt caccatet etteette	gat gagggagt ggaaat gaaaacg	Chr.3b	87.0	130	136	this study
AFC04D10	AB565002	(GA) 3aggaaaagaggagataagggaagta aaa(GAT) 3(GAA) 4at(AGA) 4	gat gcaaacgat cct ggaagaaag	gact cal ct ccct ccgaactagca	Chr.3a	42.0	165	162	Tsukazaki et al. 2011
AFC08G05	AB499503	(GA)3a(GAA)11aa(AGA)3	gttaaaggcccattgggtatgaca	gaagtt cgat gcctt catgt ct ca	Chr.2b	85.2	293	290	Tsukazaki et al. 2007
AFC09E09	AB795100	(TCT)4tcatcc(TCT)3	caaaagttcctttttggttcgacg	gagaagcaaagcagt gtgcaaaaa	Chr.2a	165.8	246	243	this study
AFC10G12	AB795101	(GAA)7	acacgttgctagtgaagtgcaagc	ccat aagactcgtacacccttggc	Chr.4a	50.9	190	200	this study
AFC12	AB499505	(TTC)8	t gcgatt gt ga gatt tt gt	ccagc gtttggatagttg	Chr.1a	137.8	239	236	Tsukazaki et al. 2007
AFD01C08	AB565006	(GTT)3(GCT)8	t gagactaggat agct tggagctt	tcctttaccttcgagatcgacaca	Chr.1b	41.9	251	257	Tsukazaki et al. 2011
AFHA00C17	AB795102	(AC)4(TC)4t(AC)5gtgca(TG)3	t at t ga a gt t t t ga at g c c c g c g	t gt at gagg cag cat t c t a ct gt gg	Chr.5b	0.0	190	195	this study
AFHT02B03	AB565010	(TG)3gatggatcgcttga(GT)9	ggatatt gatggatcggtggtgat	aacc ggt c c ct a c ct t ag c a c a at	Chr.8a	24.4	201	199	Tsukazaki et al. 2011
AFHT05F12	AB565012	(TG)3(TA)8	t cagt ttt gggaaggt ct aggggt	gat tggtaggagat ggaaa cacgc	Chr.6a	232.2	237	245	Tsukazaki et al. 2011
AFRA00E11	AB565014	(TA)8	c ccat acacacacacat ct agge	cct cat at at ct gcacat gcgt cg	Chr.6a	341.2	268	282	Tsukazaki et al. 2011
AFRA04B10	AB499515	(TG)5cagttgta(TG)8cgcgtgcat(GCG	gt gagt gt gcagtttt gt gagggt	gt gegeaggtecatet gt aaaaa	Chr.6b	10.1	296	298	Tsukazaki et al. 2007
		T)2gtgcatgc(GT)3atgtgctac(TG)3ca( TG)4cagttgtgta(TG)5catagttgta(TG) 7							
AFRA04D06	AB795103	(TG)12tct(AC)3	atcagtgat cggacat agctgctg	caat cacctct ccatttt aacgcc	Chr.5b	102.1	175	179	this study
AFRA04D09	AB499516	(TG)3tatgtgtatctaca(TG)3ca (TG)7tatatttatatgtatgtgtA)3	ctage ggggaatt tat eet ggt te	cgat acgcccaact t act cgact t	Chr.4b	44.0	147	149	Tsukazaki et al. 2007
AFRA04F03	AB499517	(A C)4(GC)3(A C)3atacat(AC)7(AT)6	accgact ttt cacat tt gct ccat	ccat ta cgga ag caat a a a ga a cc	Chr.1a	126.8	230	243	Tsukazaki et al. 2008
AFRA05F04	AB795104	(TG)15	cccttacacattt ccaattgctc	t caaat cat ggct gt ct t cct caa	Chr.7a	68.8	249	255	this study
AFRA05G10	AB565023	(AC)10(TC)5t(TA)4	ttaggcact aat gggaggtt ccct	caaccgaccatgagctgtgaaata	Chr.5b	82.9	207	209	Tsukazaki et al. 2011
AFRA07E10	AB499519	(TC)9	cagaactagacggtttgggggaa	aget cat at tttgaagete geagg	Chr.4a	99.5	170	172	Tsukazaki et al. 2007
AFRA11F11	AB499520	(CT)3atggttttaagcatcatatacat(AC)5 (AT)4(AC)4ata(TACA)2(TA)5t(AC) o	ccgccattagtgagagactacgtg	gggcaaacggttacatatagccaa	Chr.2a	154.5	213	217	Tsukazaki et al. 2008
AFRA12C02	AB565027	(TG)3cgtatgca(TG)8(CG)3tgcgtgcgc (GT)5	aatagagacccctgcacttgtcc	tgcaaaaacaaaggaaagggacc	Chr. 1a	103.8	295	297	Tsukazaki et al. 2011
AFRA12C11	AB565028	(AC)11	catctttggtccagtgcataacga	cgatt cct gt cact act aggt tt ggc	Chr.7a	79.5	256	254	Tsukazaki et al. 2011
AFRA12E05	AB499522	(TA)6(TG)15	aaaagt caat gccaagcccat ct a	gattt ccttt gtgatattcgacgtg	Chr.3b	69.8	303	305	Tsukazaki et al. 2008
AFRA12G10	AB795105	(AC)18	gcaggggggggggggggggggggggggggggggggggg	gct cgct gact ct t ggt aggaat c	Chr.5b	112.4	247	245	this study
AFRA13A04	AB795106	(TG)3atgattggttgttatttaaag(TG)9ttta tg(TA)3(TGT)3	agget gtt gaegat geaaat gata	act cgt ggct gt ct at cagct cct	Chr.3a	247.3	178	176	this study

(continued)
9.
Table

Marker	GenBank		Forward primer	Reverse primer	KiC1	nap	Fragi sizes (l	nent bp) in	Darrowed
name	Accn.	COIE IEPEaus	sequence (5'-3')	sequence (5'-3')	Linkage group	position (cM)	Ki	С	ur patied III
vFRA13E10	AB795107	(GA)3agttctaagaaaattatgataa(AC)7a a(AC)7	agtcacggccataaagtgatggag	ध्रुक्षद्वार द्वारार ध्रुवरा घट्टा १९	Chr.3a	416.3	277	273	this study
FRT01B06	A R795108	TGR	aaatatcagaattgagccacggga	aaaaacaabbcepptceacee	Chr 1a	1152			this study
EDTOTIO1			accordant at the even event of of	and associated of anal vota or	Chr. 42	2.011			
FK103H01	401C6/AA	8(JA)26(UJ)	acceatatti geageacteri gi	ಡಶ್ರೆಗೆ ಚಲ್ಲಲ್ಲಿ ಶ್ರೇಲ್ ಶ್ರೇ ಶ್ರೇ ಕ್ಷೇತ್ರ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್ತ ಸಂಗತ್	Chr.4a	6.60	677	177	this study
.FRT07F04	AB565034	(CT)3atagatacattttat(AC)4atatacaat( AC)4atacaatatgcatgcat(AC)4atgcat	cacacat at acaat acacacacat acaatat gc	ccgacaccaaaaat gaagaagtagc	Chr.8b	18.9	220	254	Tsukazaki et al. 2011
		at(AC)4atgcatat(AC)4gcatat(AC)5g							
		cacat(AC)5gcacatacat(AC)8atatacat							
		(AC)4atatacat(AC)3(AT)3(AC)3gca catatacatat(AC)4tca(TA)3							
FRT08B03	AB565035	(GT)3ataggtattgcgtgcatatagatac(AT) )4(GT)8	taggggtetgeatactecacagt	aataacttcacacgcgcttgcata	Chr.8a	14.1	271	269	Tsukazaki et al. 2011
FS006	AB499314	(AC)13(AT)6	gt gacctt at gtaggggt taggatt	tcgctccatt caa att aaaaa	Chr.3b	50.9	282	294	Tsukazaki et al. 2008
FS008	AB499315	(AC)6at(AC)6	ggtt at gaatactagcggggatag	at cagaacaacat tetget gact a	Chr.8a	62.6	254	256	Ohara et al. 2005
FS017	AB499318	(TC)5catctctttac(TC)15	t gaactttt attt atgctt ctt cct c	at ggaggacgaaggtgggtt	Chr.8a	118.4	237	235	Ohara et al. 2005
FS039	AB499319	(AT)8	c gggt aataacggat at cataaaca	cagttgttacatgtggtatcagagc	Chr.6a	347.8	286	290	Ohara et al. 2005
FS058	AB499322	(AC)8(AT)6	t ggact gaggt gaaaggat gg	acctttt ccatt gcgtttgg	Chr.7b	106.0	199	203	Tsukazaki et al. 2008
FS072	AB795110	(TG)8	ctttgttttggggggggaaactcg	ccact caa cagaccaat cat ca ca	Chr.8b	11.8	148	160	this study
FS088	AB499323	(TG)10	t at ctt cgagcacggt tctt ctt gt	at ggct tcgat gat ggat agt t gta	Chr.8a	87.1	174	160	Song et al. 2004
FS099	AB499325	(AC)13(AT)8cacacttata(AT)3	t gcccctcatt aat aacaacat gac	tt aat cgcatt gacaaagtt tat tt	Chr.1b	81.2	235	245	Song et al. 2004
FS103	AB696903	(TA)9(TG)9(TA)3tatatactt(AT)3	ttttacctagatattttcgaatttca	cat cttt cttttt cactagctt cct g	Chr.1b	15.9	255	227	Ohara et al. 2005
FS109	AB499329	(TG)3tt(TG)7	cctat gt ctt tacctat ccaaccaaca	ccgaatttcaagtgtgtcaagtttt	Chr.5b	134.5	193	197	Song et al. 2004
FS111	AB499330	(AT)3gtgtctcta(TG)8	t gt tt aat ggact tt caat gcct gt	gcat taaaat gaagaaat ccgaag	Chr.2b	89.8	230	228	Song et al. 2004
FS131	AB499333	(AC)8	caacaaatcagagagaaacagat ga	act gt at at tt at gt at cact cc at gt aaa	Chr.5b	137.5	159	161	Ohara et al. 2005
FS142	AB499335	(AC)3aa(AC)11(AT)5gt(TA)8	t gaga gaat taatat tatt ggaggcctat	at aaaat gacaaccaac ccat gtta	Chr.1a	153.7	248	252	Ohara et al. 2005
FS149	AB499337	(CA)12(TA)7tcacttgtac(AT)4	aaccaattgatt acct ct catctgc	tgcggacctt ccat agt ct gt at aa	Chr.4a	97.6	199	193	Tsukazaki et al. 2006

		)			,				
	GenBank	Forward primer sequence	Reverse nrimer sequence	KiC 1	map	Fragment sizes	i (bp) in	Restruction	
Marker name	Accn.	(5-3)	(5'-3')	Linkage group	position (cM)	Ki	C	enzyme	Reported in
ACFLS2_Msel	A Y647262	ttaaggacgaccactggtt	ccacgacatccgtgact	Chr.4a	115.8	197+105+46	125+105+72+ 46	Mse I	this study
API27_MspI	AA451547	gacaagaccatcactaagctct	agtactcctcagtcctgcct	Chr.2b	81.7	420+98	518	Msp I	this study
API32_TaqI	AA451548	aaaaccaaatcgtttgtgcc	aaaccaaagcaggcaattaac	Chr.7b	126.5	376	312+64	Taq I	this study
CF450061_InDe1	CF450061	tgaagtcagatgaaaattggaaaa	tgcgaggtttttccatttg	Chr.5a	39.4	219	220	InDel	Tsukazaki et al. 2011
CF441057_MseI	CF441057	atgcaaatttaaccatccagaact	tgatatggtggaagcacaactg	Chr.1a	297.8	200	150+50	Mse I	Tsukazaki et al. 2011
CF451262_InDel	CF451262	gattgcactgacagctcacttc	cactcctcttatcggtttgtgg	Chr.2a	30.4	150	153	InDel	Tsukazaki et al. 2011
TC0009_Hsp92II	TC3870/TC5647	taaagagaccgatgccttcat	agaaccggcacgataacagata	Chr.1b	92.4	139+105+94+ 78+54+24+9	149+105+84+ 78+54+24+9	Hsp 92II	this study
TC0024_Hsp92II	TC3945	ccattaaatacctccattatcatttccc	caccaagtacacccacttatcatcca	Chr.8b	31.8	182+116+106 +54	222+182+106 +54	Hsp 92II	this study
TC0408_Hsp92II	TC6957	gatacatatgcgagggacaagg	ctgttaatggttggttcagcact	Chr.1a	210.5	90+70	160	Hsp 9211	this study
TC0411_Hsp92II	TC411	ctgggtttacggaagaagcggaa	agcaccattgacttcctttccttg	Chr.5b	129.9	228+163+141 +31	258+163+141 +31+30	Hsp 92II	this study
TC1753_InDel	TC6173	tggcaaaggaattttcttacaa	tttgagctttgagaatgcaaca	Chr.7b	31.9	104	102	InDel	Tsukazaki et al. 2011
TC2062_Tsp5091	TC2062	tectecgtettetecatetactacea	atticacccatticctccttcgtc	Chr.4b	122.9	435	250+185	<i>Tsp</i> 5091	this study
TC2678_Nsil	TC2671	caacgtggcatcaccaatg	ccttgcaatggtgaaagactg	Chr.3a	406.3	400+16	211+189+16	Nsi I	this study
AFGA2ox1_InDel	ı	cacttccttatgtccctaacata	cctgaagtgcaatgtatagtc	Chr.6a	71.0	600	1200	InDel	Shiraiwa et al., personal communication

Table 10. Primer sequences of 13 bulb onion EST and 1 bunching onion-derived non-SSR markers located on KiC map.

Trial	QTL	Condition	Linkage group	Closest marker of peak LOD score	LOD peak	Additive effect <sup>a</sup>	Dominant effect <sup>a</sup>	$R^2$ (%) <sup>b</sup>
2008	qBlt2a	Greenhouse	Chr. 2a	AFA01E09	9.2	-19.5	-2.0	30.5
2008	qBlt6a	Greenhouse	Chr. 6a	AFB23C03	4.3	-13.8	-9.4	22.1
2009	qBlt1a	Open field	Chr. 1a	AFB11E05	10.2	-13.7	-1.7	15.4
2009	qBlt2a	Open field	Chr. 2a	AFAT04B03	8.8	-14.1	-1.6	13.1
2010	qBlt1a	Open field	Chr. 1a	AFB11E05	10.3	-18.7	-5.7	14.7
2010	qBlt1b	Open field	Chr. 1b	AFA05F04	8.8	-16.3	-5.8	12.0
2010	qBlt2a	Open field	Chr. 2a	AFAT01H05	10.7	-20.2	-4.1	16.3
2010	qBlt3a	Open field	Chr. 3b	AFAT02B11	4.7	-11.3	0.2	6.1

Table 11. QTLs for bolting time in KiC  $\mathrm{F}_{2:3}$  population.

<sup>a</sup> Additive or dominant effect of Ki allele.

<sup>b</sup> Percentage of variance explained at the peak of the QTL.



**Fig. 8.** Linkage map and QTLs detected in KiC population in different trials. QTLs are indicated by the boxes to the left sides of chromosomes and the positions of LOD peaks are shown by arrows.

The LOD peak for *qBlt1a* was located in close proximity to the marker AFB11E05 in the field trials in 2009 and 2010 (Fig. 8). The LOD peaks for *qBlt2a* were located close to the marker AFAT04B03 in the 2009 field trial and were located near the marker AFAT01H05 in the 2010 field trial. To examine the effects of the two QTLs, the F<sub>2</sub> progenies were classified according to their genotypes at each marker, the bolting time of  $F_{2:3}$  lines were correlated with the nine observed genotypes (Table 12). The F<sub>2</sub> progeny homozygous for the linked AFB11E05 and AFAT01H05 alleles derived from the Ki parent showed significant differences in bolting time from the progeny carrying the homozygous genotypes derived from the C parent (Table 12).

# QTL analysis of bolting time in the SaT03 population and comparative mapping of *qBlt2a*

A major QTL was detected in a consistent region of linkage group Chr. 2a in both field trials in 2013 and 2014 (Fig. 9 and Table 13). The maximum LOD scores for this individual QTL were 10.2 and 19.7 in the 2013 field trial and the 2014 field trial, respectively. The phenotypic variances explained by this QTL were 33.9 % in the 2013 field trial and 46.8 % in the 2014 field trial. When the  $F_{2:3}$  lines were categorized according to the genotype of their preceding  $F_2$  generation at the linked locus AFAT10E12, there were significant differences in bolting time between the genotype categories (Table 14).

Comparative mapping of Chr. 2a between the KiC and SaT03 populations was conducted using the same 10 markers. Two of the markers, AFA01E09 and AFAT04B03, which were closely linked to the LOD peaks for *qBlt2a* in the KiC population, were assigned to Chr. 2a in the SaT03 population. These markers were also linked to QTLs detected in the SaT03 population and were adjacent to AFAT10E12 (7.6 cM), which is close to the LOD peak for these QTLs. Therefore, these QTLs are considered identical.

## Discussion

To obtain information for marker-assisted selection of the late-bolting trait in bunching onion, QTLs for bolting time were identified in two populations derived from crosses

				<b>2009</b> 1	field tria	al	2010	) field t	rial
	Marker ge	enotype <sup>a</sup>	Number		Signit	ficant		Sign	ificant
Population	AFAT01H05	AFB11E05	of F <sub>2:3</sub>	Bolting time <sup>b</sup>	differ	ence <sup>c</sup>	Bolting time <sup>b</sup>	diffe	rence <sup>c</sup>
	(qBlt2a)	(qBlt1a)	lines	time	AA	BB	time	AA	BB
$F_2(Ki x C)$	А	А	5	74		*	77		*
	А	Н	12	72		*	70		*
	А	В	2	99		*	96		
	Н	А	20	72		*	59		*
	Н	Н	23	94	*	*	95		*
	Н	В	11	113	*		119	*	
	В	А	7	95	*	*	98		*
	В	Н	21	113	*		121	*	
	В	В	6	131	*		142	*	
Ki	А	А		0			0		
F <sub>1</sub> (Ki x C)	Н	Н		90			109		
С	В	В		159			170		

**Table 12.** Bolting time in  $F_{2:3}$  individuals categorized by the  $F_2$  genotypes at AFB11E05 and AFAT01H05 in the KiC population in the field trials in 2009 and 2010.

<sup>a</sup> A, genotypes homozygous for Ki allele (early-bolting); H, heterozygous; B, genotypes homozygous for C allele (late-bolting).

<sup>b</sup> Average number of days between bolting dates of Ki and  $F_{2:3}$  lines.

<sup>c</sup> Asterisk indicates significant difference between each genotype and AA or BB detected by Tukey-Kramer HSD test (P = 0.05).



**Fig. 9.** Linkage map comparison between KiC and SaT03 populations at linkage group Chr. 2a. Common markers are connected with lines.

Trial	QTL	Linkage group	Marker closest to peak LOD score	LOD peak	Additive effect <sup>a</sup>	Dominant effect <sup>a</sup>	$R^2$ (%) <sup>b</sup>
2013	qBlt2a	Chr. 2a	AFAT10E12	10.2	-10.7	-6.9	33.9
2014	qBlt2a	Chr. 2a	AFAT10E12	19.7	-12.3	-10.1	46.8

**Table 13.** QTL for bolting time in the SaT03  $F_{2:3}$  population in 2013 and 2014 field trials.

<sup>a</sup> Additive or dominant effect of 'Sa03' allele.

<sup>b</sup> Percentage of variance explained at the peak of QTL.

Table	14.	Bolting	time	in	$F_{2:3}$	individuals	categorized	by	the	$F_2$	genotype	at
AFAT1	10E1	2 (Field	trial i	n 20	)13).							

Population	Genotype	Number of F <sub>2:3</sub> lines	Bolting time <sup>a</sup>
F <sub>2</sub> (Sa03 x T03)	T03 homozygous	10	56 a
	heterozygous	48	45 b
	Sa03 homozygous	20	40 c
Т03	T03 homozygous		45
F1 (Sa03 x T03)	heterozygous		12
Sa	Sa03 homozygous		0

<sup>a</sup> Average number of days between bolting dates of Sa03 and  $F_{2:3}$  lines. Different letters indicate significant difference according to Tukey-Kramer HSD test (P = 0.05).

between parental lines differing in bolting time. Because bolting times were highly correlated between trial years in both the KiC and SaT03 populations (Fig. 6 and Fig. 7), bolting time appears to be highly heritable.

In the KiC population, a total of five QTLs could be identified on four chromosomes. Among these, two QTLs, qBlt1a and qBlt2a, were both detected in the field trials in 2009 and 2010 (Table 9). The closest markers to the LOD peaks for these QTLs were the same for qBlt1a (AFB11E05) or proximal for qBlt2a (AFAT04B03 in the 2009 field trial and AFAT01H05 in the 2010 field trial) (Fig. 8). In addition, when the F<sub>2</sub> genotypes were classified according to these linked markers, there were significant differences in bolting time between the homozygous Ki and C genotypes (Table 12). Therefore, these two QTLs will be effective for predicting bolting time under field conditions. qBlt1a was identified in the field trials in 2009 and 2010, but was not identified in the 2008 greenhouse trial. From these results, qBlt1a is likely involved in the low-temperature requirement for vernalization response because plants were grown at temperatures higher than 10 °C in the 2008 greenhouse trial.

In the SaT03 population, only a single QTL was detected during repeated field trials in 2013 and 2014 (Table 13). Comparative mapping of *qBlt2a* on both the KiC and SaT03 linkage maps revealed that *qBlt2a* would be identical in both populations. The parental line T03 that harbors the late-bolting trait requires lower temperatures to induce bolting than does Sa03 (data not shown). These results suggest that *qBlt2a* plays an important role in vernalization leading to bolting and flowering based in bunching onion.

In a previous study, a major QTL for pseudostem pungency was identified on Chr. 2a in the SaT03 population (Tsukazaki et al., 2012). The position of the pungency QTL accorded with *qBlt2a*, the bolting time QTL detected in the present study. Although it is quite unlikely that genes controlling pungency also function in flowering, the genes corresponding to these traits are closely linked in the parental line T03. The T03-type allele at AFAT04B03 closely linked to the QTL for both pungency and bolting time has not been found in Ki or C (data not shown) and has been observed in only a limited number of cultivars belonging to the bunching onion cultivar group 'Kaga' (Tsukazaki et al. 2012). These results suggest that the T03 allele of AFAT04B03 is derived from this

genetic resource; however, no effect of this allele on variation in the late bolting trait was found.

The varietal differences in bolting time of bunching onion have been related to the differences in their low temperature requirements for flower-bud formation and in the size of adult plants (Inden and Asahira, 1990). Yamasaki et al. (2000a) reported that the optimum temperature for vernalization in bunching onion was 7 °C; however, some cultivars could initiate flower buds at 15 °C (Yamasaki et al., 2000b). The very early-flowering cultivar 'Bei-cong' has a low requirement for cold treatment to induce flower initiation (Lin and Chang, 1980), and can thus flower without overwintering in Japan. In order to detect QTL for bolting unrelated to a cold requirement, the KiC population was evaluated under a minimum of 10 °C in the 2008 greenhouse trial. One QTL (*qBlt6a*) was detected only in the 2008 trial and might control a flower induction through a mechanism other than green plant vernalization. Although it is not clear whether Ki requires temperatures higher than 10 °C for flowering, this line might use an atypical pathway for induction of flowering similar to that of 'Bei-cong', which possesses a seed vernalization-like response (Yamasaki et al., 2012).

T03, the late-bolting line, has short plant stature as it is derived from the short-statured cultivar 'Fuyuwarabe' (Wako et al., 2010). Plant hormones of the gibberellin family stimulate shoot elongation and bolting (Davies, 2004). Shiraiwa et al. (2011) reported that the gene coding gibberellin 3-oxidase (AfGA3oxI) catalyzes the conversion of GA9 to GA4 and GA20 to GA1 in bunching onion. AfGA3oxI was highly expressed at early and middle developmental stages in flower stalks, and was mapped to chromosome 7A of shallot (A. cepa Aggregatum group) (Shiraiwa et al., 2011). Although AfGA3oxI has not been mapped in bunching onion, this gene is not predicted to be associated with the QTL for bolting time identified in the present study.

In conclusion, this study found that a major QTL for bolting time was located on Chr. 2a in bunching onion. The T03-derived allele at AFAT10E12, an SSR locus linked to qBlt2a would be a useful marker for selecting bunching onions with late bolting traits.

# Chapter V: Construction of an *Allium cepa* linkage map using doubled haploid technology

## Introduction

*A. cepa* is subdivided into two groups: Common onion, which is referred to as bulb onion; and Aggregatum, which is known as shallot (Fritsch and Friesen, 2002; Hanelt, 1990). Shallot has a close genetic relationship to bulb onion, although it differs from bulb onion in morphological and ecological characters. Shallot is an important genetic resources for the improvement of bulb onion because it carries useful traits such as resistance to *Fusarium oxysporum* (Vu et al., 2012). Genetic studies in *A. cepa* have been limited due to its outcrossing nature, biennial generation time, and inbreeding depression.

Although several polymorphic isozyme loci in bulb onion have been found (Cryder et al., 1991), these were insufficient for construction of a linkage map. Recently, genetic mapping of *A. cepa* using DNA markers has progressed (Wilkie et al., 1993; Bradeen and Havey, 1995; van Heusden et al., 2000a; 2000b; Ipek et al., 2005; Martin et al., 2005; Baldwin et al., 2012; McCallum et al., 2012; Duangjit et al., 2013) and genetic information on various traits has been amassed. For example, QTLs controlling sucrose and fructan contents (McCallum et al., 2006) and genes encoding lachrymatory factor synthase (Masamura et al., 2012) have been reported. SSR markers are ideal DNA markers owing to their simplicity, reproducibility, and codominant inheritance (Jones et al., 1997). Fischer and Bachmann (2000) reported for the first time the development of 30 SSR markers from bulb onion. Araki et al. (2009) analyzed the genetic relationships among *A. cepa*, *A. fistulosum*, *A. vavilovii*, *A. galanthum*, *A. roylei*, and *A. altaicum* using these SSR markers. Tsukazaki et al. (2008; 2011) developed numerous SSR markers from bunching onion (*A. fistulosum*) and used them to construct linkage maps in bunching onion.

Doubled haploid (DH) techniques that use the chromosomal doubling of haploid plants can shorten the time needed, offer homozygous pure lines, and provide valuable materials for genomic analysis (Alan et al., 2003). Because dominant loci could be treated equally to co-dominant loci in segregating DH population, efficient mapping could be conducted in wheat (Jia et al., 2005), Chinese cabbage (Ajisaka et al., 2001) and cabbage (Voorrips et al., 1997) etc. However, a linkage map construction using DH lines has not been reported in *Allium*. Campion and Allon (1990) first reported the haploid plant regeneration from female gametes by *in vitro* culture of unpollinated flowers in *A. cepa*. Although chromosome doubling was induced in small percent of regenerated plants (Jakše et al., 2003; Sulistyaningsih et al., 2006), the colchicine treatment was necessary in almost haploid plants to survive.

In the present study, gynogenic individuals derived from F<sub>1</sub> plants crossed between shallot and bulb onion DH lines were produced. Informative markers polymorphic between the parental DH lines were developed and a linkage map was constructed using this gynogenic population. In addition, several markers closely linked to useful traits were mapped.

### Materials and methods

## Plant materials and unpollinated flower culture

 $F_1$  hybrids from a cross between the shallot DH line DHA as a seed parent and the bulb onion DH line DHC as a pollen parent were used for unpollinated flower culture. DHA was derived from the shallot strain 'Chiang Mai' from Thailand, and DHC was derived from the long-day onion cultivar 'Sapporo-ki' from Japan, as described in Abdelrahman et al. (2015). Unpollinated flower culture of  $F_1$  hybrids was conducted according to the methods of Sulistyaningsih et al. (2006), with some modifications. Flower buds collected 3 days before flowering (6–7 mm in length) were sterilized in 70% ethanol and 0.01% mercury (II) chloride. Twenty-one florets were placed on a Petri dish (9 cm in diameter) containing B5 solid media (pH 5.8) with 4% (w/v) sucrose and 2 mg/l 2,4dichlorophenoxyacetic acid. Plates were incubated in a growth chamber maintained at 25 °C with a day length of 16 h for 3 months. Shoots emerging from florets were transferred to MS solid media with 4% sucrose and cultured for two months. Thereafter, plantlets were acclimatized and grown in greenhouse. The number of somatic chromosomes in each plantlet was determined by the Feulgen staining and squash method after treating seedling root tips with 0.05 % colchicine at 4 °C for 3 h.

# Marker analysis

Genomic DNA was isolated from leaf blades of regenerated plants according to the method described by van Heusden (2000a) and used for marker analysis. To identify female-derived plants, a primer set (see Table 16) was used to amplify a fragment of the GI (GIGANTEA; Gene Bank Accession No. GQ232756) gene that is polymorphic between DHA and DHC. PCR was performed as follows: pre-incubation at 94 °C for 3 min; followed by 35 cycles of denaturation at 94 °C for 1 min, annealing at 60 °C for 1 min, and extension at 94 °C for 1 min. PCR products were digested with TaqI for 3 h and were separated by electrophoresis through 2% (w/v) agarose gel. To detect polymorphisms between DHA and DHC, SSR markers derived from the A. cepa genome-(A. cepa gSSRs, Fischer and Bachmann, 2000), SSR markers derived from the A. fistulosum genome(A. fistulosum gSSRs), SSR markers derived from A. cepa ESTs (Kuhl et al., 2004; Martin et al., 2005; Jakše et al., 2005; Tsukazaki et al., 2008; 2011) and other mainly CAPS and SCAR markers (Kuhl et al., 2004; Masuzaki et al., 2006a; 2006b; 2006c; McCallum et al., 2006; Yaguchi et al., 2008) were screened for polymorphisms between parental lines. In addition, eight markers linked to the Ms locus (Park et al., 2013; Yang et al., 2013) were also screened.

PCR was performed in a 10- $\mu$ L reaction mixture containing 20 ng template, 1  $\mu$ L 10 × PCR buffer, 0.8  $\mu$ L dNTPs (2.5 mM each), 0.8  $\mu$ L forward primer (10  $\mu$ M), 0.8  $\mu$ L reverse primer (10  $\mu$ M), and 0.05  $\mu$ L Ex Taq Polymerase (5 units/ $\mu$ L; Takara Bio Inc.). PCR conditions appropriate for each primer set are described in Table 15, Table 16, and Table 17. PCR products were separated on 3% (w/v) agarose gel or 5% (w/v) denaturing polyacrylamide gel according to the methods of Song et al. (2004), Ohara et al. (2005a), and Tsukazaki et al. (2008). In addition, some forward primers were Xuorescent-labeled with 6-FAM, NED, PET, or VIC dyes (Applied Biosystems, CA, USA) prior to use for PCR. PCR products were loaded onto a capillary DNA sequencer (ABI3730; Applied Biosystems), and analyzed using GeneMapper ver. 3.0 software (Applied Biosystems).

# **Construction of linkage map**

I AUIC 10.	LITTICI SCHUCT	ces and mapping mon	nation of 4.5 Autum cepu and 1.4. Justin	CC MINCON	N IIIAINCIN	n rin nalen	IC IIIIago III	ap.	
				Ð	Annealing	DH	map	Chromosomal location	
Marker name	Genbank Accn.	Core repeats	Forward primer sequence (5-5) control Reverse primer sequence (5'-3') control Reverse	CK ondition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
A. cepa EST-de	rived SSR (32)								
ACE031	CF445265	(CAA)7	ATGATTTTTCGAGACGCCAAAATGAG TTCAAAGAGCGCTACGGGCTATAGTG	·	50.0	Chr.8	0.0	8C	Tsukazaki et al., 2011
ACE052	TC387	(GAA)8	AGTCGTATGAATCCTTTCTTCTCCAT TTCTCTTTGCTTTG	-1	50.0	Chr.1b	31.7	4C	Tsukazaki et al., 2011
ACE063_2	CF438796	(CAG)8	TTTAGAGTCCAAGAATGGGTAGGCAAG AGTTCTCCCTTAAACAGTGCAAAAT	1.	55.0	Chr.1a	24.6		This study
ACE068	CF449065	(GCA)9	AGCTACGAGCACGATCTAAGCCATT CTGCCTTCCTTAATCTTCCCAGGT	1.	50.0	Chr.4	73.2	4C	Tsukazaki et al. 2011
ACE069	TC6947	(GCC)3(GCA)10	GTTATTCAACCAACAAGCACCCC ACTTAGTGCTGTAACTGGCCGGAG	.1	55.0	Chr.2	89.0	2C	Martin et al., 2005 Tsukazaki et al., 2011
ACE072	TC6464	(GGA)6	ACCATCCGCTGGTAATGAGACAAT GTCCTGCATAGCTGATCCCAATTT	·I	50.0	Chr.2	15.2	2C	Tsukazaki et al., 2011
ACE080	TC1121	(TA)22	AGGATTAAAATGGGAGAAAGAGAT TTGAGAATAAGAAGCCGCTTGA		50.0	Chr.1b	0.0	1F	Tsukazaki et al., 2008
ACE092	CF439016	(TA)4tgtacatacgta(TG)5( TA)7g(AC)4	AGATGGATTGTTCCAAAGCCCATAA CTGCACAACGTACCCAATTAAGCA	· <b>-</b>	55.0	Chr.6a	15.5		This study
ACE127	TC1106	(TG)4tatg(TA)6tgtgg(TG )3(TA)6	TCGAGACGAGCACGCATGAAAA AGCTCCATTGTCCGTTCCATT	. <b>.</b>	55.0	Chr.4	55.9	5F	Tsukazaki et al., 2008
ACE163	CF449247	(AGC)6	GGATGGAGGTTTCAGAGACAAG GCTGCAGTATTTCTTGCTGATG	· <b>-</b>	55.0	Chr.6a	2.0		This study
ACE192	CF438037	(TG)8(GT)5	ACACATGCGTCTGTGTGTGTATGT TAGCTCTCGGTATCTCCCCAG		55.0	Chr.6a	22.8		This study
ACE209	CF442000	(ATAC)7(AT)5	CGTACCAGTGGAAGATTTGTCA CAGGAAGCTGCAAATGATAGTG	1.	55.0	Chr.4	70.4		This study
ACE241	CF446529	(CT)5cc(CT)4	ACTCTCTTCTCGAGCTCCCTC CCGTTTTCCGACATTATTTTCT	· <b>-</b>	55.0	Chr.5b	0.0	5F	Tsukazaki et al., 2011
ACE292	TC6067	(A)8c(A)8c(A)10ac(A)1 0	TGGAAGCATCTGATATTAGAAAAAGA TCGAAGAAGCTTCAGAGAGGAT	· <b>-</b>	55.0	Chr.2	130.2		This study
ACE302	TC6971	(A)8g(A)10	CGAACTGCATAAAAGTTAGGGC CTCGAGAAGGCTAAGGCAAATA		55.0	Chr.7	158.0		This study

and 31 A fistulosum SSR markers located on the linlage man and manning information of 43 Allinum con Tahla 15 Primer se

Table 15. (co	ntinued)								
Marbar	GanRant		Forward nrimer securence (51.21)	ЪСР	Annealing	HQ	map	Chromosomal location	
name	Accn.	Core repeats	rouward primer sequence $(5^{-3})$ Reverse primer sequence $(5^{-3})$	condition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
ACE469	CF441873	(T)10	TTGGCATATATGATCCTATGCAG CGAGAAATAAATCAAGCACAGC		55.0	Chr.4	19.7		This study
ACE489	CF444905	(A)10	AAAGATCGATCCCAAGTCTCTG TGGTAGCTCCATGCATAATCAA	i	55.0	Chr.3a	66.1		This study
ACE547	TC4431	(A)8g(A)10g(A)10(GAA A)4	CAAACTTGAATCCCTTCCAAAA GGTATCCAAGTTAGTGCCCAGA	.1	55.0	Chr.4	19.7		This study
ACM009_A	CF436678	(AGC)6	GCAACGGTAGAAGAACCTGC AACCTCTTTTGGTGCCTCCT	.1	50.0	Chr.1a	50.6	1C	Kuhl et al., 2004 Tsukazaki et al., 2011
ACM009_B	Id.	Id.	Id.	.bI	.bI	Chr.3a	0.0		
ACM024	CF446873	(GCA)10	CCCCATTTTCTTCATTTTTCTCA TGCTGTTGCTGTTGTTGTTG	.1	55.0	Chr.2	89.1	2F	Kuhl et al., 2004 Tsukazaki et al., 2008
ACM065	CF449328	(TG)24	TAAGCTCTGATGGAGGATGGTT GTCGGTCACATTGAAAACTGAA	1	50.0	Chr.5b	34.4	5C	Kuhl et al., 2004 Tsukazaki et al., 2011
ACM071	CF449595	(AG)10	TCTCATTTCAACTTTCTACCTATCC CTGACATTTGCTCGACTGGA	·I	50.0	Chr.5a	15.1	5F	Kuhl et al., 2004 Tsukazaki et al., 2008
ACM132	CF447889	(ATAC)14ac(CATG)4	TGCACACCGTTTCCATTTTA ATGGGGCCTGGTAAGTTTTT		55.0	Chr.2	60.3	2F	Martin et al., 2005 Tsukazaki et al., 2011
ACM134	CF449417	(GA)8	CACACACCACACACATCAA ACACACACAAGAGGGAAGGG		58.0	Chr.7	59.1	7C	Martin et al., 2005 Tsukazaki et al., 2011
ACM168_2	CF450904	(TG)6	GGCCATGAGACATACAACTCAA AACTGCAAGAAGAGAAATTGCC		50.0	Chr.3b	40.1	3F	Tsukazaki et al., 2011
ACM169_2	CF451226	(TG)4ta(TG)4, (TG)4ta(TG)6	AACATTCTCCCCCCTTCTTC GAGGAGGAGTTACGTGTTGC		50.0	Chr.3b	32.0		This study
ACM170	CF437581	(TTC)6	TTCTGCAATGAAAACACATTGA ATCCAACTGAGTCGGCAATC	1.	50.0	Chr.8	53.2		Jakše et al., 2005
ACM177	CF440609	(TCA)14	TGAGACTGGTGCAGGAACTG CAAGGAAATACATTTGTGGGAAAA		50.0	Chr.2	148.0	2C	Martin et al., 2005 Tsukazaki et al., 2011
ACM225	CF444176	(A)12	GTGTTTTATACTTGCGCAATCC TCAAATCAAGCTCGAACAAAAA		50.0	Chr.4	81.5		This study
ACM266	CF451915	(TG)6	GGGTTCAACCTCTGAAGTCTTG TAAGTTCACATGAGCATCCCAC		50.0	Chr.2	11.9		This study

					Annealing	DH	map	Chromosomal	
Marker name	GenBank Accn.	Core repeats	Forward primer sequence (5'-3') Reverse primer sequence (5'-3')	PCR condition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
ACM304	DQ273270	(CA)7	GAATTTAGGCCCATTTCAAGG TGATTTGCCTAATGTTTTTACG	:=	64.0	Chr.5b	16.0	5C	McCallum et al., 2012
A. cepa gSSR (11									
AMS02	,	CCACACCACACACA CCACCACACACACA A	GCATTAACTATCTAAAACATTG CCATCAACTCATAACAGGT	:=	54.0	Chr.7	77.1		Fischer and Bachmann, 2000
AMS03-199		(GT)21	TAACCCTAGGATGAGTTGAG GGATTTCCTCTTGAGATGA	ij	56.0	Chr.7	53.7		Fischer and Bachmann, 2000
AMS03-254		Id.	Id.	.bI	.Id.	Chr.4	0.0		
AMS03-293		Id.	Id.	Id.	.Id.	Chr.2	30.7		
AMS03-490		Id.	Id.	Id.	.Id.	Chr.5a	42.0		
AMS03-580		Id.	Id.	.bI	Id.	Chr.8	74.5		
AMS17		(CA)7tg(CA)21(TA)3	AGTGGACTCAAGGCAGATG ATCACCATTCACCGTTTACT	:=	58.0	Chr.1b	49.5		Fischer and Bachmann, 2000
AMS21	ı	(CA)25	GGTTGTTTCCACTACACTTGAG CGTCCTTGGTATTCTTGTGC	:=	56.0	Chr. 1a	63.1		Fischer and Bachmann, 2000
AMS23	·	(AT)5(GT)19	GCTGTTCACTGGTCTATCTGG ATTCGGTGCTGATTTTCG	Ξ	58.0	Chr.4	78.3		Fischer and Bachmann, 2000
AMS29	ı	aag(AAAG)2ggata(GAA )3aagaagaagaagaagaa(C AA)2(CA)2	CATCAGAAAATCGCATCAC TTGAAACTTGGAAGGTTGTC	:=	54.0	Chr.8	30.8		Fischer and Bachmann, 2000
AMS30	1	(CA)8cg(CA)22(TA)4	CACTAATGGGGGTAAATAATGTTCTAC TTGCCTTGAAATCCAGAC	Ξ	57.4	Chr.2	143.8		Fischer and Bachmann, 2000

/	~								
Marken	CanDants			a.v.c	Annealing	HQ	map	Chromosomal location	
Marker name	Genbank Accn.	Core repeats	rorward primer sequence (5'-3') c Reverse primer sequence (5'-3') c	ruk condition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
A. fistulosum gS	SR (31)								
AFA01E09	AB499346	(TA)3tctatatgt(AC)13	CCCAACCTACTTGAGGGATTGCTA GTTCTGTGTATGCAGGCAATTTGG	· =-	50.0	Chr.2	107.1	2F	Tsukazaki et al., 2008
AFA01F12	AB499348	(AC)12(AT)3	TGAAGGGGGACAAAATAAGAAGCA TCTCCCCCACTTAAAAGAATTTCG	-1	50.0	Chr.2	0.0	5F	Tsukazaki et al., 2015
AFA01H12	ı	(TA)6(TG)18	CAGAACAATCAATAGCCACGACCA TCAGGAGACCGCCTACCTTTTGTA		50.0	Chr.2	77.6	3F	Tsukazaki et al., 2008
AFA02C03	ı	(TA)5(TG)17	TTCAGGTTGCATCCCATATTGTT TTCAGGTTTGGAGACCACCAAGTT	-1	50.0	Chr.2	103.7		This study
AFA02E09	·	(AC)4at(AC)11(AT)6	TGCATACACACACGCAGACATA TTCCCCCTTTCTAGTCATCATGAGA	-=	50.0	Chr.6b	0.0	6F	Tsukazaki et al., 2008
AFA03D03	·	(TA)3catgta(TG)8ca(TG) 7ta(TG)3	TGTCCAAAGCAGTGTTAGCTGGAA CCAAGCACATGAAAGTGCACAAG	-11	50.0	Chr.3a	18.8	3F	Tsukazaki et al., 2008
AFA03D07	AB499363	(TG)3(CA)18	TGTCCTCGAAAATGAGATGCTTCA TGAAATGTAACCCGACTGTCCAAA	· <b>—</b>	50.0	Chr.7	11.6	7F	Tsukazaki et al., 2008
AFA03F08	AB499365	(TA)4(TG)7ta(TG)7	TTAGGTAAAGGGACGAAACGACCA TGCCTCCAGGACTGAACAATACAA	-1	50.0	Chr.5b	9.7	1F	Tsukazaki et al., 2008
AFA07G06		(TG)5tc(TA)5(TG)14	TGAACCGTTCCCATCTACCTTCTG ATGAGATCGTGATACAACAGGCG		50.0	Chr.6a	3.0	8F	Tsukazaki et al., 2008
AFA09C08	AB499397	(TA)4c(AT)4(GT)10a(T G)4c(GT)6a	GAAACCGATAATGACTTAATTGCTTGAA	I I	50.0	Chr.2	71.1	5F	Tsukazaki et al., 2007
AFA10A08	AB499401	ta(TG)9 (TG)11	CCTTATCITAATCIGCTTCCAGTCTCTCA GTTTAGGGCGTAAAATCTAAACGCT GTGCTTTTGACTAACCTCGCATCC	. <u> </u>	50.0	Chr.6a	0.0	5F	Tsukazaki et al., 2008
AFA10B06	AB499402	(AC)15(TC)6	CCGCAAACTGGACTAACTGTACAAA GGCAGAATAGCCCAAGTGTTTTCA		50.0	Chr.6a	49.5	1F	Tsukazaki et al., 2007
AFA11G05	AB795053	(TA)5(TG)11	GGTCAGGAGCAACCCAAACATTAC TCCTTTACCACCATACCTGATTTTG		50.0	Chr.8	40.8	5F	Wako et al., 2016
AFA13G12	AB795055	(AC)12	GGATTTCCTTAGGCGATTTGATGAGTC TTACAACTCCATCGCTGTG	.i Q	50.0	Chr.6b	2.0	3F	Tsukazaki et al., 2015

Table 15. (co	ntinued)								
Markar	GanDonb		Economic animar carnianca (51-21)	aJa	Annealing	HQ	map	Chromosomal location	
Ivlarker name	Accn.	Core repeats	roi wate primer sequence (5'-3') Reverse primer sequence (5'-3')	rondition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
AFA16E07	AB564840	(GT)3(ATGT)2a(TG)6tat g(TATATG)2 (TG)4tc(TA)4tccg(TG)3 caoa(TG)4	TATGCGTAGACATGAGGGCACAAT CACAAACATGAGAAATACGTTGGCA		50.0	Chr.7	0.0	3F	Tsukazaki et al., 2007
AFA23G01	AB564861	(TA)8(TG)14cgc(AT)3gt agatcatatgtatg (TA)3tggatgtttatt(TG)4	CATCATTTCCATCATCACCTCCATGTC AAGGGCTAAGGTTTCTCTTCCTACA		55.0	Chr.7	7.76	IF	Tsukazaki et al., 2011
AFAA03F01		(AC)8	CGACTTTGTTCTCGCTCTTGGTT AAATTGCACAAGGCTCTGCGAGAT		50.0	Chr.7	40.9	7F	Tsukazaki et al., 2007
AFAA06C01	AB564881	(AC)5aaaaacatacat(AC)1 1aa(AC)3(AT)3	TGGTACTCACTGATTTAATGGCATAAT CTTTAATGATGATCTGTTGCATACG		50.0	Chr.5a	61.7	5F	Tsukazaki et al., 2011
AFAT00B05	AB499440	(AT)3ctatgctaaccttctctaaa gtacaac (TA)6tttcgtgtatc(TG)12	TGTCCACTATTCACGGCTTACTCA GATGGGTCTGCCTCCTCATTTT		50.0	Chr.3b	0.0	2F	Tsukazaki et al., 2007
AFAT00B07	AB795073	(AC)7gc(AC)6(ATAC)4 ct(AC)9a(TA)8	CCTATACTTGTCATCCAATAGTGAACC ATTAGTAATGGACGGGCCTAGGGA	-	50.0	Chr.3b	53.7	2F	Tsukazaki et al., 2015
AFAT04C08		(AG)14(TG)6tatc(TG)5tc tgtg(TATG)2ta(TATG)2t ag(TG)5	TGGTACCATAAACCTCGAACCAATG CATATTCAAGGGAATGCCACATGC		50.0	Chr.7	95.0	7F	Tsukazaki et al., 2008
AFAT05E04		(TA)3tg(TA)3tctctgtgtctg tg(TC)3(TG)8gggggggtgt (GT)3(TG)5	TGTTATTTGAGGTGAAGCAGCAAGA CCATTTTTGATGTGGGGGGCAAAACC		50.0	Chr.8	22.2	8F	Tsukazaki et al., 2008
AFB08D06_A	AB564956	(AG)18	GGGTGTATGTGTACGGCTATGGGT CTCCGTCTCCTACCTCCTCCAAAT		50.0	Chr.3a	42.4	6F	Tsukazaki et al., 2011
AFB08D06_B	Id.	Id.	Id.	.bI	.bI	Chr.1b	37.3		
AFB20E09	AB795092	(TC)14	AACACATACCGGAGGCAACCTAGA ATGTATGCTTCGGATTGCTGATGA	· <b></b>	50.0	Chr.2	59.6	2F	Tsukazaki et al., 2015
AFB237	AB499495	(CA)3tgt(GA)16	AGGCGTAGCATAAGGGTG GTGTGCGTGAGTGACTGTG		50.0	Chr.2	131.7	7F	Tsukazaki et al., 2008
AFB23C03	AB795095	(TC)3ctgcatctatcc(TC)10	AAGATGGATAGCGGCTTAAACAAG TCAAGCAAATATCGCGTCTGAAAC		55.0	Chr.4	14.8	6F	Wako et al., 2016
AFC09D10	ı	(GAA)8	GATGGGTGTCAGCGTTGTTAAGTG TCATTTCCCCGAGTCTTTGTCTTG		50.0	Chr.3a	11.1	4F	Tsukazaki et al., 2008

_
(d
ō
Ē
n
Ξ.
Ē
0
ଁ
ં
5. C
<b>15.</b> (c
<b>15.</b> (c
e 15. (c
ole 15. (c
able 15. (c
able 15. (c

					Annealing	DH1	nap	Chromosomal location	
Marker name	Genbank Accn.	Core repeats	Forward primer sequence (5'-3') Reverse primer sequence (5'-3')	PCK condition <sup>a</sup>	temperature (°C)	Linkage group <sup>b</sup>	Position (cM)	assigned in previous studies <sup>c</sup>	Reported in
AFRA03E09	1	(CG)4c(AC)9a(TA)5tctat gtgcg(TG)3	TTGCCAGGATGAGAAGAGGT TTGCCAGGATGAGAAGAGGT		50.0	Chr.4	36.3		This study
AFS015		(TA)4tgtatg(TA)6tg(TA) 5tgtg(TA)6tg(TA)3	ATCTCACTGTCCTTGTACCTGAAAG CATCTTGACTTTGTGATATTTGTGC		50.0	Chr.1b	24.7	1F	Ohara et al., 2005 Tsukazaki et al., 2008
AFS111	AB499330	(AT)3gtgtctcta(TG)8	TGTTTAATGGACTTTCAATGCCTGT GCATTAAAATGAAGAAAATCCCGAAG		50.0	Chr.3b	24.0	2F	Song et al., 2004 Tsukazaki et al., 2008
<sup>a</sup> i, an initial de	enaturation ster	to at 94 °C for 2 min; 10 cy	cles at 94 °C for 30 s; at annealing temper	ature + 10 °C	(reducing by	/ 1 °C per cy	cle) for 30 s	and 72 °C for 3(	) s ; 35 cycles at 94

°C for 30 s; at annealing temperature for 30 s and 72 °C for 30 s; and a final 4 min extension at 72 °C. ii, an initial denaturation step at 94 °C for 3 min; 40 cycles at 94 °C for 1 min; at annealing temperature for 1 min and 72 °C for 1 min; and a final 7 min extension at 72 °C. iii, an initial denaturation step at 95 °C for 11 min; 40 cycles at 94 °C for 30 s; at annealing temperature for 30 s and 72 °C for 30 s; and a final 10 min extension at 72 °C.

<sup>b</sup> Linkage groups determined chromosomal identies using monosomic addition lines were indicated in boldface type.

<sup>c</sup> 'C' represents the chromosome number of A. cepa and 'F' indicates that of A. fistulosum.

1 able 10.		ducinces an	u mapping intorma	C-11011 NAATIAN-I CAT NAAD MINING 07 10 110111		VEIS INCALED		map.			
-		- -			PCR	Annealing	F	Map loca	alization	Chromosomal location	
Marker name	Marker type	GenBank Accn.	Annotation	Forward primer sequence (5'-3') Reverse primer sequence (5'-3')	condi- tion <sup>a</sup>	temperature (°C)	Restriction enzyme	Linkage group	position (cM)	assigned in previous studies <sup>b</sup>	Reported in
ACAAX07	CAPS	CF447127	Vitis vinifera Tubulin alpha-1 chain-like	TGAGTTGGTGATTTCAGCCACGGAGA GCATGTGCCCCGTGCTATATTTGTTG		55.0	Taq I	Chr.5b	23.9		Kuhl et al., 2004
ACABE16	CAPS	CF447747	Ricinus communis Serin/threonine protein phosphatase 2a requlatory subvnit A	AAGATTCCGTACGCCTGTTAGCTGTTG CTTGACACGGGAAGAATGTGCTGTAT		55.0	Afa I	Chr.4	76.7		Kuhl et al., 2004
ACABX01	CAPS	CF448816	Oryza sativa OsNHX1	CATTCTCACAGTGTTCTTTTGCGG ATTGGATAAGAAAGAAGGGGAA	·	55.0	Afa I	Chr. 1a	0.0	6C, 6F	Kuhl et al., 2004 Tsukazaki et al., 2011
ACAEJ67	CAPS	CF451546	<i>Ricinus communis</i> Ayxin response facter	GCAGGTATCAGCGTCAACTAATAAGGAA CATGACTGTCTGTGGACGACTTGCAC		55.0	Afa I	Chr.5a	0.0		Kuhl et al., 2004
ACE6067	CAPS	AJ006067	Allium cepa mRNA for invertase	TTGCCAGGATGAGAAGAGGT TTGCCAGGATGAGAAGAGGT		55.0	Taq I	Chr.3b	42.4		This study
AF212154	SCAR	AF212154	<i>Allium cepa</i> ATP- sulfurylase (ATPS)	CCAAAACCGCCTAATCCTCA GGCACCGACATATTTACAAAGG	Ξ	63.0	ı	Chr.3a	70.2		This study
AF458090	CAPS	AF458090	<i>Allium cepa</i> high affinity sulfate transporter (ST)	TACATGGCTGGATGCCAAAC ATTCACAACTCCTGGTACCTTTC	:=	68.0	Alu I	Chr.7	8.1		This study
AY221244	SNP	AY221244	Allium cepa putative chalcone synthase A (CHS- A)	CGATACATGCACGTAAACGAAC ATGCGCTCGACATATTCCC	:=	68.0	ı	Chr.2	97.8	2C	Masuzaki et al., 2006a, 2006b
AY221246	CAPS	AY221246	Allium cepa flavonone 3- hydroxylase (F3H)	AGAGAGGGAAATATGTAGG GGCTCCTCTAATATCGGTT	:=	65.0	Alu I	Chr.3a	50.9	3C	Masuzaki et al., 2006a, 2006b
AY221250	SCAR	AY221250	Allium cepa dihydroflavonol 4- recuctase (DFR)	CAAAAGCCCGAATACGATG CGGTTCATTTGGATGATGG	:=	65.0	ı	Chr.7	106.8	7C	Masuzaki et al., 2006a, 2006b
AY541032	CAPS	AY541032	Allium cepa cinammate 4- hydroxylase (CA4H)	TCGGCAACTGGCTCCAAGTC TGATTGACCAGTTCCGCTATGCC	:=	70.0	Afa I	Chr.1a	43.2		This study

Table 16. Primer sequences and mapping information of 28 Allium cepa EST-derived non-SSR markers located on the linlage map.

		~									
J. T.	M M				PCR	Annealing		Map loca	lization	Chromosomal location	
marker name	type	denbank Accn.	Annotation	r or ward primer sequence ( <i>z</i> - <i>z</i> ) Reverse primer sequence ( <i>s</i> '-3')	condi- tion <sup>a</sup>	temperature (°C)	enzyme	Linkage group	position (cM)	assigned in previous studies <sup>b</sup>	Reported in
AY585677	SCAR	AY585677	Allium cepa anthocyanidine synthase (ANS)	TTTGCTCGATCGTTTAGCRGAAGAAGA TGAGGATGATGACAAAGTTAGCGGAGC A	:=	68.0	1	Chr.4	23.5		Kim et al., 2005
AY700850	CAPS	AY700850	Allium cepa Chalcone isomerase (CHI)	TGCTTTGATTCAGTCATCC AATAATCGACTCCAATACGG	:=	67.0	Fok I	Chr.3a	58.8	3C	Masuzaki et al., 2006a, 2006b
CF434892	CAPS	AY434892	Cinnamomum osmophloem 4- coumarate-CoA ligase (4C1.)	AGTAGCCATGAATCCCAACCTC TTCACGCCCTGTACAGATTCC	:=	67.0	Taq I	Chr.1a	0.0		This study
CF437610	SNP	CF437610	Asparagus officinalis acid inveltase (AIV)	GGTTCAAAAGACGCATCCAA TAATCCTGCCCATTATCAGAAGT	:=	60.0		Chr.2	124.3	2C	Yaguchi et al., 2008
CF443389	CAPS	CF443389	<i>Chrysanthemum</i> x <i>mortfolium</i> geranylgeranyl pyrophosphate svnthase (GGPS)	CCTAAACCTGTTCTAACACCTAC CAGTGGATTCAAGATCCACTAC	:=	60.0	Hae III	Chr.5a	67.6		This study
EU164758	CAPS	EU164758	Allium cepa sucrose- phosphate synthase (SPS)	GAAGGCTGATATTGTTGGTGAAG TGTGTCGTAGGAGCCTGATG	:=	64.4	Afa I	Chr.8	60.2	8C	Yaguchi et al., 2008
FT3	SCAR	ı	Allium cepa flowering locus T-like protein 2 (FT3)	CAATGGTGATGCATGAGCC GGTGCATACACAGTTTGCC	:=	62.0		Chr.2	19.7		This study
GQ232751	CAPS	GQ232751	Allium cepa CONSTANS-like protein (COL)	GGAGAAGGTGACACGTGG GTCTGCTGTACGGGTTGG	:=	66.0	Hae III	Chr.4	8.2		This study
GQ232756	CAPS	GQ232756	Allium cepa GIGANTEA (GI)	GATCTTGCATCTAGACTC ACATAGTTTGCCTCAAG	:=	60.0	Taq I	Chr.7	57.2		This study
TC141	CAPS	TC4475	Silene latifolia endoglucanase	GCAACTGGAAATAACACTTACC GGTAGTGTTTGCCATATCCC	· <b>_</b>	65.0	Mbo I	Chr.2	120.2		This study
TC1698	CAPS	TC5921	<i>Desulfitobacterinm</i> <i>hafniense</i> diguanylate cyclase	CAGATACTGCTCGGCTTACTAAATTCAC TCTATCTGAATTAGGAGGTGCATATCCC	· <b>-</b> -	57.0	Taq I	Chr.1a	68.0		Kuhl et al., 2004

Table 16. (continued)

Table 16.	(continue	(p									
					РСВ	Annealino		Map local	ization	Chromosomal	
Marker name	Marker type	GenBank Accn.	Annotation	Forward primer sequence $(5^{-3})$ Reverse primer sequence $(5^{-3})$	condi- tion <sup>a</sup>	temperature (°C)	Restriction enzyme	Linkage group	position (cM)	assigned in previous studies <sup>b</sup>	Reported in
TC1915	CAPS	BQ580175	Tamarix hispida cationic peroxidase	GGACAAATTTGCTGCTATGGG ACCCTGGGTACCAGTAAGG		57.0	Afa I	Chr.4	40.6		This study
TC5837	CAPS	CF436276	Brachypodium distachyon chlorode channel protein CLC-c- like	CGTGCTGCTCCTATGGACTAC GCTCAGGAGCCACTGAGAAC		65.0	Hha I	Chr.1b	41.9		This study
TC6857	CAPS	CF436951	Triticum aestivum DNA polynerase alpha catalytic subunit	CGTGAGATCACAGCAAAGCATC GATCATGACTTTCTACGGCAAAGG		58.0	Taq I	Chr.2	114.4		This study
TC7174	CAPS	CF435297	Asparagus officinalis early flowering protein 1 (EF1)	GTGGAGTTGGCAGCATCAGG CGGTAGGATTGGCAGCAAGG		70.0	FokI	Chr.8	41.6		This study
TC7745	CAPS	CF445035	Lycoris longituba transcription factor putative late elongated hypocotyl (LHY)	TGGACTGATCAAGAGCAC CACGAGACAACCATCTTCC		62.0	I qsX	Chr.3b	46.0		This study
Y07838	CAPS	Y07838	Allium cepa fructan 6G- fructosyltransferase (6G-FFT)	TCTTTTGCTCGGTTTGGTTC TTGCCCCAATATGCAAAATC		50.0	Taq I	Chr.6a	2.0	6C	McCallum et al., 2006
<sup>a</sup> i, an initial for 3 min: 35	denaturatior -40 cycles a	n step at 94 °C f ut 94 °C for 1 m	for 2 min; $35-40$ cycles at $94$ in: annealing for 1 min and 7	<sup>o</sup> C for 0.5 min; at annealing temperature for 1 2 <sup>o</sup> C for 1 min and a final 7 min extension at	1 min and 7 72 °C.	72 °C for 30 s; a	nd a final 4 min	extension at 72	2 °C. ii, an in	iitial denaturat	ion step at 94 °C

-a

<sup>b</sup> Linkage groups determined chromosomal identies using monosomic addition lines were indicated in boldface type.

° 'C' represents the chromosome number of A. cepa and 'F' indicates that of A. fistulosum.

	-			Annealing		Map loci	alization	
Marker name	Marker type	Forward primer sequence (5'-3') Reverse primer sequence (5'-3')	PCK condition <sup>a</sup>	temperature (°C)	Restriction enzyme	Linkage group	position (cM)	Reported in
jnurf05	SCAR	AACAAATCAATCGCCTGAAAA ATTATGGCCGATTTCTCAGC	· <del></del>	57.0		Chr.2	55.5	Park et al., 2013
jnurf17	CAPS	GAGCCCACTGCTATTGAGGA CCTGGAATGTATAGCAAGCCTAA	-1	59.0	Xho I	Chr.2	60.7	Park et al., 2013
jnurf20	SCAR	GGTGAAGATGGCATGTGGGGGGGGAGATCAAA TAGATGATGTTCTCAGGCGGCGGGGGGTT	:11	65.0		Chr.2	38.3	Park et al., 2013
DNF-566	SCAR	TACAGATTTGTTATCTTCTTCTTCT TTCATTTGTTAGGATGTACTCTTACC	Ш	58.0		Chr.2	58.9	Yang et al., 2013
RNS-357	SCAR	TCAGTATCAATAGAAGGAATCAC GTATACCATTGGTACTTGATGCA	Ξ	59.0	ı	Chr.2	58.9	Yang et al., 2013
<sup>a</sup> i, an initia <sup>oC for 30 c.</sup>	ul denatura	tion step at 95 °C for 5 min; 10 cycles at 95 °C for 3 in terms for 20 °C for 3 fin	10 s; 67 °C (0.	8 °C decremen	ts in each cycl	e) for 30 s a	und 72 °C fo	r 30 s ; 30 cycles at 95

Table 17. Primer sequences and mapping information of five Allium cepa Ms linked markers located on the linlage map.

<sup>o</sup>C for 30 s; at annealing temperature for 30 s and 72 °C for 30 s; and a final 7 min extension at 72 °C. ii, an initial denaturation step at 95 °C for 5 min; 40 cycles at 94 °C for 30 s; at annealing temperature for 1 min and 72 °C for 1 min; and a final 10 min extension at 72 °C. iii, an initial denaturation step at 95 °C for 6 min; 35 cycles at 94 °C for 30 s; at annealing temperature for 45 s and 72 °C for 45 s; and a final 5 min extension at 72 °C. iii, an initial denaturation step at 95 °C for 6 min; 35 cycles at 94 °C for 30 s; at annealing temperature for 45 s and 72 °C for 45 s; and a final 5 min extension at 72 °C.
A total of 107 markers polymorphic between the parental lines and an additional phenotypic marker, a locus controlling bulb pigmentation, were used to construct a linkage map. Standard  $\chi^2$  analysis was used to test the segregation pattern at each marker locus for deviations from the expected Mendelian ratio of 1:1 in the population. Linkage analysis was performed using the computer program JoinMap ver. 4.0 (van Ooijen, 2006). The Kosambi function was used to obtain cM values (Kosambi, 1944). A log of the odds threshold of 3.0 was used to group and order the markers. The chromosomal locations of each linkage group were determined by comparison to previously developed maps (McCallum et al., 2012; Tsukazaki et al., 2008, 2011, 2015). Additionally, a complete set of bunching onion-shallot MAALs (2n = 17; FF + 1A to +8A, where F stands for the basic chromosome set of *A. fistulosum*, and A represents chromosome identity of the detected linkage groups. Shallot 'Chiang Mai' and *A. fistulosum* 'Kujo-hoso' and monosomic addition lines were used as DNA sources. PCR was conducted as described above and amplified products were separated on 2–3% (w/v) agarose gel.

## Results

## Production of gynogenic mapping population

Florets from F<sub>1</sub> plants flowered 5–7 days after initiating culture. Ovaries enlarged and shoots emerged from them after 2–3 months. A total of 291 shoots were obtained from 10,604 florets and 100 shoots regenerated into plants. Observation of somatic chromosomes indicated that these regenerated plants included 46 haploids (2n = 8), 40 diploids (2n = 16), three triploids (2n = 24), three tetraploids (2n = 32), and eight chimeric plants (Table 18). Their genotypes at the co-dominant marker locus GI were either shallot-type or bulb onion-type, indicating that all of the regenerated plants were derived from female gametes of F<sub>1</sub> plants (Fig. 10).

### Linkage map construction

Of the 666 markers tested, polymorphisms between DHA and DHC were detected with 235 markers (35.3%) (Table 19). The efficiency of polymorphism detection was highest

No. of plants tested	Frequency distribution of plants that showed different chromosome numbers							
	2n = 8	8, 16	16	24	16, 32	32		
100	46	7	40	3	1	3		

**Table 18.** Chromosome numbers of plants from gynogenesis in unpollinated flower culture of  $F_1$  hybrids.

Table 19. Numbers of polymorphic markers between DHA and DHC.

Origin	Marker type	No. of tested markers	No. of polymorphic markers	Percentage of polymorphic markers
Allium cepa genome <sup>a</sup>	SSR	32	9	28.1
Allium cepa EST	SSR	196	96	49.0
Allium cepa genome <sup>b</sup>	SSR	22	5	22.7
Allium cepa EST	Non SSR	102	32	31.4
Allium cepa genome <sup>c</sup>	SCAR	8	5	62.5
Allium fistulosum genome	SSR	306	88	28.8
Total		666	235	35.3

<sup>a</sup> Fischer and Bachmann (2000).

<sup>b</sup> BAC library

<sup>c</sup> Yang et al. (2012) and Park et al. (2012)



Fig. 10. DNA polymorphisms of CAPS (GI) markers between genotypes.

(a) M, 100bp ladder; S1 and S2, DHA; C, DHC; H<sub>1</sub> and H<sub>2</sub>; F<sub>1</sub> crossed between DHA and DHC. (b) S, DHA; C: DHC; 0–39, regenerated plants from unpollinated flower culture of F<sub>1</sub> hybrids.

for A. cepa EST-derived SSRs (49.0%), followed by A. cepa EST-derived InDels or SNPs (31.4%). A. fistulosum gSSRs showed polymorphisms at the same frequency as A. cepa gSSRs. A linkage map was constructed with 108 markers (32 A. cepa EST-SSRs, 31 A. fistulosum gSSRs, 28 A. cepa EST non-SSRs, 11 A. cepa gSSRs, five A. cepa SCARs, and the phenotypic bulb color marker) (Fig. 11). The resulting map contains 12 linkage groups spanning 799 cM. Of 107 mapped markers, 25 deviated from the expected 1:1 segregation ratio (P < 0.05). Eighteen markers were previously assigned to A. cepa chromosomes (Masuzaki et al., 2006a, 2006b, 2006c; Yaguchi et al., 2008; Tsukazaki et al., 2011; Table 15 and Table 16). In the present study, the chromosomal locations of 10 markers were newly identified using MAALs (Fig. 12). From these results, all 12 linkage groups were assigned to eight chromosomes. The bulb skin color phenotypic marker was mapped to the terminal part of chromosome 7 and located in close proximity to the AY221250 marker that was derived from a gene coding dihydroflavonol 4-reductase (DFR) (Fig. 11). Twenty-eight A. fistulosum gSSR markers mapped on A. fistulosum genetic maps (Tsukazaki et al., 2008; 2011; 2015; Wako et al., 2016) were located on an A. cepa linkage map in the present study (Table 15). Of these, only 10 markers shared identical chromosomal locations in two species. Five SCAR markers, jnurf05, jnurf07, jnurf20 (Park et al., 2012), RNS-357, and DNF-566 (Yang et al., 2012), which were reported to be closely linked to the Ms nuclear male-fertility restoration locus of cytoplasmic male-sterility (CMS) were located on chromosome 2. Four markers except jnurf20 were mapped in close proximity within 5 cM of each other (Fig. 11).

#### Discussion

This is the first report of linkage map based on a gynogenic segregating population in *A. cepa*. Compared with previously reported bulb onion maps (e.g., the 1907 cM map reported by Martin et al. 2005), the map constructed in the present study covering 799 cM does not cover the entire genome, therefore, gaps in this map need to be filled in by identifying and mapping more markers. Nevertheless, this linkage map contains useful genetic information for agronomic traits because of the close linkage between the bulb pigmentation and the gene encoding DFR, and assignment of several markers relevant to



Fig. 11. Linkage map constructed using a gynogenic population derived from F1 crossed between DHA and DHC. This map consists of 12 linkage groups with 108 markers covering 799 cM. All the linkage groups were assigned to eight chromosomes of Allium cepa (Chr. 1–8). Five flanking markers to Ms locus located on Chr. 2 developed by Yang et al. (2013) and Park et al (2013) are underlined.



**Fig. 12.** Identification of relevant *Allium cepa* chromosomes for markers GQ232756 digested with *Fok* I (a), GQ232751 (b), TC7174 digested with *Fok* I (c) and TC7745 digested with *Taq* I (d) by using MAALs. Arrows indicate shallot-specific DNA fragments.

*Ms* on chromosome 2.

Shallot has a close genetic relationship with bulb onion and is an economically important crop in low-latitude regions (Shigyo and Kik, 2008). Using MAALs, loci affecting contents of a major metabolite that interacts with abiotic stress (Masuzaki et al. 2006c), amino acids, cysteine sulfoxide (Masamura et al. 2011), sucrose, and fructans (Yaguchi et al. 2008) in shallot have been revealed. Specific saponin compounds conferring resistance to *Fusarium* basal rot have been identified in shallot (Vu et al. 2012). These reports demonstrated the potential of shallots as a useful resource for the genetic study and future breeding of bulb onions.

Of the markers used in the present study, A. cepa EST-derived SSRs were most informative. SSRs comprised the largest category of polymorphic markers (96) identified here, with a high (49.0%) frequency of marker polymorphism between parental lines. Thirty-two markers were located on the linkage map and the chromosomal assignments of nine of 10 markers were consistent with previous studies (Table 15). Furthermore, seven markers were located on the A. fistulosum genetic map and six of these were assigned to the same chromosome (Table 15). Synteny comparisons of the genomes of these species could now be performed using linkage maps with common markers. Many A. fistulosum-derived gSSR markers have been useful for mapping in A. cepa. Thirty-one A. fistulosum-derived gSSR markers were assigned to this linkage map. The frequency of polymorphism in markers derived from A. fistulosum gSSRs (28.8%) was equal to those of A. cepa derived gSSRs (28.1%). About two-thirds of A. fistulosum-derived gSSR markers located on the A. cepa map did not show accordant chromosomal assignment. The chromosomal locations of 3 markers (AFA13G12, AFAT00B7, and AFB08D06 A) determined using MAALs differed from those of A. fistulosum. Nevertheless, these markers could be useful as common markers in both species, as fewer gSSR markers derived from A. fistulosum have been thus far assigned to A. cepa genetic maps (Tsukazaki et al., 2011). Additionally, SSR markers allowing high-throughput genotyping using a DNA sequencer can enable efficient and low-cost genetic analyses. Recently, Tsukazaki et al. (2015) isolated more than 2300 di- to penta-nucleotide SSRs from *de novo* assembly of transcripts from A. fistulosum. These EST-derived SSRs would be useful for

constructing a linkage map in A. cepa.

Molecular markers linked to the male-fertility restoration locus relevant to CMS-S have been developed in onions. Gökçe et al. (2002) identified two RFLP markers flanking the *Ms* locus at distances of 0.9 and 8.6 cM, and Bang et al. (2011) converted these two RFLP markers into PCR markers. Yang et al. (2013) developed SCAR markers (DNF-566 and RNS-357) tightly linked to both alleles at *Ms* locus. Park et al. (2013) developed six SCAR markers linked to the *Ms* locus and mapped these markers onto chromosome 2. In the present study, five of these eight markers were located on linkage group Chr. 2. Two markers, jnurf05 and jnurf17, were closely linked at a distance of 5.2 cM. Park et al. (2013) reported that no recombinant was found between jnurf05 and the *Ms* locus, and jnurf17 also tightly linked to these loci. The genotype of DHA shallot at the *Ms* locus would be *msms* (data not shown). These results suggest that higher resolution mapping of *Ms* locus is possible using a gynogenic segregating population such as the one produced in this study.

These chromosome-specific markers will be useful for intensive mapping of other markers on specific chromosomes to obtain markers closely linked to desirable QTL for agricultural traits, as well as for comparing genomic synteny between *A. cepa* and *A. fistulosum* using linkage maps.

## **Chapter VI: General discussion**

Bunching onion is one of the most important vegetable crops in East Asian countries and represents the highest annual production value of all *Allium* crops in Japan. Since the latter 1980s, F1 hybrid cultivars of bunching onion have been released quickly to the market. Growers desire cultivars with high yield, quality, and adaptability to a wide range of environmental conditions. However, published studies on the genetics of bunching onion have been very limited. Moue and Uehara (1985) have described the mode of inheritance of cytoplasmic male sterility (CMS) in bunching onion. Yamashita et al. (2010) have discovered male sterile plants from several bunching onion accessions and verified the inheritance mode of the male sterility via intraspecific crossings in order to identify CMS resources. CMS is an indispensable trait for F1 hybrid seed production in bunching onion. Ohara et al. (2004; 2005b; 2009) examined the seedling growth of F1 hybrids between various inbred lines or cultivars in bunching onion and demonstrated the mode of inheritance and heterosis in seedling growth by diallel analysis and QTL analysis using genetic maps. These results suggest that the breeding of cultivars with vigorous seedling growth is possible by exploiting heterosis, and the resulting cultivars might be suitable for plug nursery systems and machine-assisted transplanting. Tsukazaki et al. (2012) revealed that a major QTL for pungency was located on chromosome 2 in bunching onion. However, genetic studies of disease resistance and bolting time in bunching onion had been lacking despite the importance of these agronomic traits. Firstly, the present study showed the potential of A. cepa as a useful resource for breeding rust resistance bunching onion. A. cepa cultivars were highly resistant to rust fungus P. allii in controlled environment and field tests. The gene(s) for rust resistance located on chromosome 1A were found to act mainly during the seedling stage. Secondly, QTL analysis of rust resistance in the bunching onion parental line developed by recurrent selection was conducted. Three QTLs with minor effects were detected, although no major QTL was detected. Thirdly, bolting time was evaluated in two bunching onion populations under different environmental conditions in order to understand the genetic relationships between late bolting and vernalization. A major QTL was detected on the same linkage group Chr. 2a in the two populations and their maps were compared using the same set of markers. Fourthly, a linkage map for *A. cepa* was constructed using doubled haploid technology to obtain information for marker-assisted selection in *Allium* crop species. This general discussion refers to the advanced study of breeding for disease resistance and late bolting, and offers suggestions regarding development of markerassisted selection using genomic information for *Allium* vegetable breeding.

#### 1) Genetic studies of disease resistance in bunching onion

In a previous study, 133 cultivars of bunching onion were evaluated for differences in susceptibility to rust under artificial inoculation in an experimental field; however, no completely resistant cultivar was found in bunching onion (Yamashita et al., 2005). In the present study, A. cepa cultivars including bulb onions and shallots showed high levels of resistance at different ages to P. allii isolates derived from A. fistulosum. These results indicate that A. cepa could serve as a useful resource for breeding rust resistance in bunching onions. Screening of a set of bunching onion-shallot monosomic addition lines revealed a high level of resistance only in FF+1A during the seedling stage, suggesting that the gene(s) controlling rust resistance could be located on chromosome 1A of shallot. At the adult plant stage, the degree of rust resistance exhibited by MAALs, multichromosome addition lines, and hypoallotriploids was not consistent under controlledenvironment and field conditions, even though shallot showed distinct rust resistance. The effects of plant age and position of the inoculated leaf could be considerable, as discussed by Jennings et al. (1990b). Further, environmental conditions influencing infection could be variable, particularly during field tests. From the results of the present study, we can conclude that A. cepa possesses much greater rust resistance than does A. *fistulosum* and that multiple genes quantitatively control this trait. Further investigation for more accurate assignment of rust resistance genes to shallot chromosomes is necessary. Introgression of rust resistance of shallot into bunching onion lines could be achieved using MAALs. Recombination between homoeologous chromosomes during meiosis has not been induced by gamma-ray irradiation. However, a minimum of two heteromorphic bivalents has been observed in meiotic analysis of interspecific hybrids between bunching

onion and bulb onion (Peffley, 1986). Complete resistance to downy mildew (Peronospora destructor) in onion was found in the wild relative A. roylei Stearn, and was introgressed into A. cepa using interspecific hybridization and backcross breeding (Kofoet et al., 1990; van der Meer and de Vries, 1990). Although A. roylei was taxonomically assigned to the section Schoenoprasum, its nuclear DNA and cpDNA are closely related to A. cepa (Havey, 1992; van Raamsdonk et al., 2000; 2003). Successful hybridization between A. roylei and A. cepa resulted in interspecific hybrid plants and backcross progenies. The downy mildew resistance locus was located on the distal end of chromosome 3 via GISH (genomic in situ hybridization) (Scholten et al., 2007) and mapping study (van Heusden et al., 2000b). Although a large introgressed A. roylei fragment harbors a recessive lethal factor located proximal to the downy mildew resistant gene, a recombinant containing a crossover between the lethal factor and the resistance gene was identified, and a homozygous introgression line was successfully produced (Scholten et al., 2007). A. roylei could also serve as a bridging species between A. cepa and A. fistulosum (Khrustaleva and Kik, 1998; 2000). However, it took about 20 years to introduce the downy mildew resistant gene successfully into the cultivated onion (Scholten et al., 2007), so interspecies introgression of genes is a difficult task.

Among bunching onion cultivars, considerable variation has been observed in the degree of rust severity. This suggests that rust resistance still persists in bunching onion and is a quantitative trait. QTL analysis is effective for revealing the mode of inheritance of the traits and is useful for developing selection markers. In the current study, three QTLs (*qRst1a*, *qRst3a*, and *qRst8a*) were detected in different trials and were validated using a population derived from residual heterozygous lines. Inconsistent results between trials are partly due to differences in evaluation methods despite a relatively high correlation coefficient between two experiments (Fig. 3). Fukino et al. (2013) and Yoshioka et al. (2014) reported a number of QTLs for powdery mildew resistance and downy mildew resistance in cucumber, the identity of which depend on tests conducted under different in temperatures or inoculation methods. The rust resistance in bunching onion is considered to be adult-plant resistance (APR), such as that reported for stripe rust resistance in wheat (Chen, 2005), because few varietal differences in disease severity are

observed at the seedling stage in bunching onion (Table 1). APR is more likely conferred by minor genes that are typically race non-specific, inherited quantitatively, and have greater potential for durability (Chen, 2005). An APR gene usually contributes partial resistance and combinations of 4-5 APR genes can act additively to confer adequate levels of durable resistance (Singh et al., 2011). In wheat, APR genes for stripe rust at 13 loci have been cataloged and more than 160 QTLs that reduce stripe rust severity have been identified (Rosewarne et al., 2013). Combinations of several such QTLs (genes) are required to obtain sufficiently high levels of resistance (Singh et al., 2011). Many studies have shown that selection for resistance can be performed visually in disease nurseries, but clearly such selection is greatly aided by use of molecular markers. Bunching onion lines resistant to rust disease have been selected in the field by inoculation or spontaneous infection. However, disease severity depends on environmental factors and plant conditions. The use of DNA markers enables reliable selection of resistant plants even at the seedling stage, especially because selection for disease resistance controlled by multiple genes requires many plants in a large field, more time, and higher breeding costs than selection for resistance controlled by a single gene.

## 2) Genetic studies of late bolting in bunching onion

The bolting time of bunching onion cultivars depends on their low-temperature requirements (Inden and Asahira, 1990). In the present study, QTL for bolting time were identified using two populations derived from crosses between parental lines differing in bolting time. In one population KiC, a parental line with the early-bolting trait, was derived from the Taiwanese ever-flowering cultivar 'Bei-cong', which has minimal requirements for low-temperature to induce flower initiation. Two QTLs, *qBlt1a* and *qBlt2a* were consistently detected in the two-year field trials. Because markers linked to each QTL have been validated using populations, these two QTLs will be effective for predicting bolting time under field conditions. The late-bolting parental line for the population SaT03 was derived from the recently released cultivar 'Fuyuwarabe'. 'Fuyuwarabe' has novel bunching onion characteristics such as low pungency, short thick leaves, and a pseudostem (Wako et al., 2010). The late-bolting trait is desirable for

extending the growing season of this novel type of bunching onion. A single QTL for late bolting was detected in the SaT03 population during repeated field trials. Comparative mapping showed that *qBlt2a* was identical on the KiC and SaT03 linkage maps. These results suggest that *qBlt2a* plays an important role in vernalization leading to bolting and flowering in bunching onion. In order to remove the effect of a low-temperature requirement, The KiC population was evaluated under a minimum of 10 °C in the heated greenhouse. *qBlt2a* was detected, although another QTL, *qBlt6a*, was observed only in this trial. This unique QTL could control a flower-induction mechanism different from vernalization.

Many genes controlling the induction and timing of flowering have been isolated and characterized in Arabidopsis (Crevillén and Dean, 2010; Dennis and Peacock, 2007). In Arabidopsis, vernalization results in the epigenetic silencing of the floral repressor FLOWERING LOCUS C (FLC) (Song et al., 2012). FLC encodes a MADS-box transcription factor and has been identified as a repressor of the floral transition. The silencing of FLC allows the photoperiodic induction of FLOWERING LOCUS T (FT), which encodes a mobile signaling protein involved in regulating flowering (Andrés and Coupland, 2012). Lee et al. (2013) reported functional characterization of the FT-like family in onion (A. cepa) and indicated that FT-like genes (AcFT2) were involved in the vernalization-responsive initiation of flowering, while other FT genes (AcFT1 and AcFT4) were involved in the photoperiodic induction of bulb formation in onion. Baldwin et al. (2014) revealed significant population differentiation in AcFT2 and AcSOC1 in bulb onion. They also identified a QTL for bolting, designated as AcBlt1 on chromosome 1, using the 'Nasik Red' × DH2150 population and discussed the association of these candidate genes with flowering (Baldwin et al., 2014). AcFT2 was mapped to chromosome 5, and AcFT1 was mapped to chromosome 1 using the W202A  $\times$  'Texas Grano' mapping population. However, upon comparative mapping between two populations, AcFT1 was not closely linked with AcBlt1 (Baldwin et al., 2014). Although AcSOC1, the homolog of Suppressor of overexpression of constants 1 (SOC1) was revealed on chromosome 1 using a set of MAALs (Shigyo et al., 1996), this gene could not be mapped in either population.

Mapping studies in bunching onion have implied a high degree of genome synteny with bulb onion (Tsukazaki et al., 2008), but comparative mapping has not yet been conducted, because the number of markers common between bunching onion and bulb onion have been limited. Recently, more than 50,000 unigenes for bunching onion were obtained from transcriptome shotgun assembly of next-generation sequencing data, and numerous SSR, SNP, and InDel markers have been developed (Tsukazaki et al., 2015). These unigene collections should be searched for orthologous candidate genes involved in flower formation in the future. These informative studies will facilitate the understanding of genes related to the induction of flowering and comparative mapping of the genomes of bunching onion and bulb onion.

# 3) Toward application of marker-assisted selection to the breeding of *Allium* vegetables

The efficiency of marker-assisted selection (MAS) using molecular markers closely linked to genes controlling agronomic traits for decreasing breeding costs and time has been demonstrated (Tanksley, 1993; Young, 1999; Kumar, 1999; Ohsawa, 2003). The genetic characteristics of bunching onion and bulb onion had not previously been investigated using molecular approaches. Molecular genetic analyses will allow comparative genomics approaches between bunching onion and bulb onion to facilitate molecular breeding for both crops.

The *A. cepa* linkage map constructed in the present study consists of 12 linkage groups with 108 markers including 32 *A. cepa* EST-SSRs, 31 *A. fistulosum* gSSRs, 28 *A. cepa* non-SSR ESTs, 11 *A. cepa* gSSRs, five *A. cepa* SCARs, and a phenotypic marker for bulb pigmentation, covering 799 cM. This is the first linkage map based on a gynogenic segregating population in *A. cepa*. Although this map has smaller coverage compared with previously reported bulb onion maps (Martin et al., 2005; McCallum et al., 2012; Duangjit et al., 2013), the substantial marker resources accumulated in recent years will enable additional mapping. Many bulb onion-derived markers have been used for genetic linkage mapping in bunching onion (Tsukazaki et al., 2008). In the present study, seven SSR markers derived from *A. cepa* EST that had previously been assigned to *A. fistulosum* 

genetic maps (Tsukazaki et al., 2008, 2011) were located on the *A. cepa* linkage map. Additionally, a considerable number of SSR markers derived from bunching onion were allocated to the *A. cepa* linkage map. These results permit comparison of genomes of these species using linkage maps. This map would be useful for molecular mapping and QTL analysis of other agronomic characters of bulb onion. Furthermore, comparative mapping between *A. cepa* and *A. fistulosum* using common markers would facilitate the analysis of the genetics of *Allium* crops.

The availability of genomic information, such as whole-genome sequences, expressed sequence tag sequences, and high-density genetic linkage maps, facilitates transition from traditional to molecular breeding. For example, the whole-genome sequencing and construction of draft genomes have been achieved in tomato, eggplant, and radish (The Tomato Genome Consortium, 2012; Hirakawa et al., 2014; Kitashiba et al., 2014). In bulb onion, Baldwin et al. (2012) performed skim sequencing of 6.6 Mbp genomic DNA using a next-generation sequencer. De novo assembly of transcriptome sequence data has been used for construction of high-resolution maps (Duangjit et al., 2013; Kim et al., 2015). Tsukazaki et al. (2015) and Sun et al. (2016) reported more than 50,000 unigenes obtained from transcriptome shotgun assembly, respectively. However, the genus Allium includes species with some of the largest nuclear genomes among cultivated plants (12–16 Gbp) with very low gene density of 1 per 168 kb (Jakše et al., 2008). Several molecular markers linked to characteristics have been developed in A. cepa (Gökçe and Havey, 2002; Gökçe et al., 2002; McCallum et al., 2006; 2007; Yaguchi et al., 2008; Baldwin et al., 2014) and A. fistulosum (Ohara et al., 2009; Tsukazaki et al., 2012). Molecular markers for rust resistance and bolting time QTLs in bunching onion were identified in the present study. The linkage map developed here will enable assembly of genomic sequence data into chromosome models. Such genomic information will accelerate the development of DNA markers linked to important gene functions relevant to agronomic and other traits. It is expected that these research outcomes will facilitate the breeding of Allium crops suitable for the needs of growers and consumers in the future.

#### Acknowledgements

I am most grateful to Dr. Masayoshi Shigyo, my major supervisor (Professor at the Graduate School of Science and Technology for Innovation, Yamaguchi University), for his able guidance, invaluable suggestions, advice, encouragement, and insightful criticisms throughout this study.

I would like to express my deep gratitude to Dr. Shin-ichi Ito (Professor at the Graduate School of Science and Technology for Innovation, Yamaguchi University) for his generous assistance, and kind support and encouragement of this work.

I wish to express my deep gratitude to Dr. Naoki Yamauchi (Professor at the Department of Biological and Environmental Science, Faculty of Agriculture, Yamaguchi University) for his generous assistance, and kind support and encouragement of this work.

I wish to sincerely thank Dr. Akihiro Itai (Professor at Kyoto Prefectural University) for his advice and constructive criticisms in this study.

I would like to express my deep gratitude to Dr. Akio Kojima (former Director General of the Institute of Vegetable and Tea Science, National Agriculture and Food Research Organization (NIVTS, NARO)) for his able guidance, invaluable suggestions, advice, encouragement, and insightful criticisms throughout this study.

I wish to give special thanks to Drs. Hikaru Tsukazaki (Tohoku Agricultural Research Center, NARO), Ken-ichiro Yamashita (Western Region Agricultural Research Center, NARO), Shigenori Yaguchi (National Fisheries University), and Takayoshi Ohara (NIVTS, NARO) for their generous assistance, kind support, invaluable suggestions and advice throughout this study.

I would like to express my deep gratitude to Drs. Satoshi Shimazaki, Naoko Midorikawa and Takako Sakai (Mikado Kyowa Seed Co. Ltd.) for their generous assistance, and kind support and encouragement of this work.

I wish to sincerely thank Mr. Satoshi Iwata (Department of Biological and Environmental Science, Faculty of Agriculture, Yamaguchi University) for his generous assistance and kind support of this work.

I am also grateful to Mss. Kimiko Tanaka, Keiko Wakita, Mayumi Ikeyama, Teruko Kihira, Hiroko Tanoda, and Satomi Negoro (NIVTS, NARO) for their technical

assistance.

I would also like to thank the staff of the Research Support Center of NIVTS, NARO for management of the field sites.

This work was supported by a Grant-in-Aid for Young Scientists (B) (19780010) from the Japan Society for the Promotion of Science, and the NARO Research Project No. 211 "Establishment of Integrated Basis for Development and Application of Advanced Tools for DNA Marker-Assisted Selection in Horticultural Crops."

## Literature cited

- Abdelrahman, M., Y. Sawada, R. Nakabayashi, S. Sato, H. Hirakawa, M. El-Sayed, M.Y. Hirai, K. Saito, N. Yamauchi, and M. Shigyo. 2015. Integrating transcriptome and target metabolome variability in doubled haploids of *Allium cepa* for abiotic stress protection. Mol. Breeding 35: 195.
- Abe, T. and H. Nakazumi. 2004. Varietal differences based on low temperature requirements and the optimum temperatures for vernalization in Japanese bunching onion. Bull. Hokkaido Pref. Agric. Exp. Stn. 86: 11–17.
- Ajisaka, H., Y. Kuginuki, S. Yui, S. Enomoto, and M. Hirai. 2001. Identification and mapping of a quantitative trait locus controlling extreme late bolting in Chinese cabbage (*Brassica rapa* L. ssp. *pekinensis* syn. *campestris* L.) using bulked segregant analysis. Euphytica 118: 75–81.
- Akaba, M., Y. Kaneko, K. Hatakeyama, M. Ishida, S. W. Bang, and Y. Matsuzawa. 2009. Identification and evaluation of clubroot resistance of radish chromosome using a *Brassica napus – Raphanus sativus* monosomic addition line. Breeding Sci. 59: 203– 206.
- Alan, A. R., M. A. Mutschler, A. Brant, E. Cobb, and E. D. Earle. 2003. Production of gynogenic plants from hybrids of *Allium cepa* L. and *A. roylei* Stearn. Plant Sci. 165: 1201–1211.
- Albini, S.M., and G.H. Jones. 1988. Synaptonemal complex spreading in *Allium cepa* and *Allium fistulosum*. II. Pachytene observations: the SC karyotype and the correspondence of late recombination nodules and chiasma. Genome 30: 399–410.

- Andrés, F. and G. Coupland. 2012. The genetic basis of flowering responses to seasonal cues. Nat. Rev. Genet. 13: 627–639.
- Araki, N., S. Masuzaki, H. Tsukazaki, S. Yaguchi, T. Wako, Y. Tashiro, N. Yamauchi, and M. Shigyo. 2010. Development of microsatellite markers in cultivated and wild species of sections *Cepa* and Phyllodolon in *Allium*. Euphytica 173: 321–328.
- Baldwin, S., R. Revanna, M. Pither-Joyce, M. Shaw, K. Wright, S. Thomson, L. Moya, R. Lee, R. Macknight, and J. McCallum. 2014. Genetic analyses of bolting in bulb onion (*Allium cepa* L.). Theor. Appl. Genet. 127: 535–547.
- Baldwin, S., R. Revanna, S. Thomson, M. Pither-Joyce, K. Wright, R. Crowhurst, M. Fiers, L. Chen, R. Macknight, and J. A. McCallum. 2012. Toolkit for bulk PCR-based marker design from next-generation sequence data: application for development of a framework linkage map in bulb onion (*Allium cepa* L.). BMC Genome 13: 637.
- Bang H, D. Y. Cho, K. Yoo, M. Yoon, B. S. Patil, and S. Kim. 2011. Development of simple PCR-based markers linked to the *Ms* locus, a restorer-of-fertility gene in onion (*Allium cepa* L.). Euphytica 179: 439–449.
- Bradeen, J. M. and M. J. Havey. 1995. Restriction fragment length polymorphisms reveal considerable nuclear divergence within a well-supported maternal clade in *Allium* Section *Cepa* (Alliaceae). Amer. J. bot. 82: 1455–1462.
- Brewbaker, J. L. 1983. Breeding for disease resistance. In: T. Kommedahl and P. H. Williams (eds.). Challenging problems in plant health. American Phytopathological Society, St. Paul, MN. pp 441–449.
- Brewster, J. 2008. Onions and other vegetable alliums. CABI Wallingford, UK
- Brueggeman, R., N. Rostoks, D. Kudrna, A. Kilian, F. Han, J. Chen, A. Druka, and B. Steffenson. 2002. The barley stem rust-resistance gene Rpg1 is a novel disease-resistance gene with homology to receptor kinases. Proc. Natl. Acad. Sci. USA 99: 9328–9333.
- Campion, B. and C. Allon. 1990. Induction of haploid plants in onion (*Allium cepa* L.) by in vitro culture of unpollinated ovules. Plant Cell Tiss. Organ Cult. 20: 1–6.
- Chen, X. M. 2005. Epidemiology and control of stripe rust (Puccinia striiformis f. sp.

tritici) on wheat. Can. J. Plant Pathol. 27: 314–337.

- Crevillén, P. and C. Dean. 2011. Regulation of the floral repressor gene FLC: the complexity of transcription in a chromatin context. Curr. Opin. Plant Biol. 14: 38–44.
- Cryder, C. M., J. N. Corgan, N. S. Urquhart, and D. Clason. 1991. Isozyme analysis of progeny derived from (*Allium fistulosum ×Allium cepa*) ×*Allium cepa*. Theor. Appl. Genet. 82: 337–345.
- Davies, P. J. 2004. The plant hormones: their nature, occurrence, and function. In: Davies,
  P. J. (ed) Plant hormones biosynthesis, signal transduction, action. Kluwer Academic
  Publishers, Dordrecht, The Netherlands, pp 1–15.
- Dedryver, F., M. F. Jubier, J. Thouvenin, and H. Goyeau. 1996. Molecular markers linked to the leaf rust resistance gene *Lr24* in different wheat cultivars. Genome 39: 830– 835.
- de Jong, P. D. 1995. Growth of leek rust epidemics in time in three cultivars during the early stage of the epidemic. Eur. J. Plant Pathol. 101: 139–148.
- Dennis, E. and W. J. Peacock. 2007. Epigenetic regulation of flowering. Curr. Opin. Plant Biol. 10: 520–527.
- Dong, Y. X., Z. H. Cheng, H. W. Meng, H. Q. Liu, C. N. Wu, and A. R. Khan. 2013. The effect of cultivar, sowing date and transplant location in field on bolting of Welsh onion (*Allium fistulosum* L.). BMC Plant Biol. 13: 154.
- Dreiseitl, A. and B. J. Steffenson. 2000. Postulation of leaf resistance gene in Czech and Slovak barley cultivars and breeding lines. Plant Breed. 119: 211–214.
- Duangjit, J., B. Bohanec, and A. P. Chan, C. D. Town, and M. J. Havey. 2013. Transcriptome sequencing to produce SNP-based genetic maps of onion. Theor. Appl. Genet. 126: 2093–2101.
- Emsweller, S. L. and H. A. Jones. 1935. An interspecific hybrid in *Allium*. Hilgardia 9: 265–273.
- FAO. 2013. FAOSTAT. http://faostat3.fao.org/browse/Q/QC/E.
- Feuerstein, U., A. H. D. Brown and J. J. Burdon. 1990. Linkage of rust resistance genes from wild barley (*Hordeum spontaneum*) with isozyme markers. Plant Breed. 104:

318-324.

- Feuillet, C., S. Travella, N. Stein, L. Albar, A. Nublat and B. Keller. 2003. Map-based isolation of the leaf rust disease resistance gene Lr10 from the hexaploid wheat (*Triticum aestivum* L.) genome. Proc Natl. Acad. Sci. USA 100: 15253–15258.
- Fischer, D. and K. Bachmann. 2000. Onion microsatellites for germplasm analysis and their use in assessing intra- and interspecific relatedness within the subgenus Rhizirideum. Theor. Appl. Genet. 101: 153–164.
- Ford-Lloyd, B.V. and S. J. Armstrong. 1993. Welsh onion *Allium fistulosum* L. In: Kalloo G, Bergh BO (eds) Genetic improvement of vegetable crops, Pergamon Press, pp 51–58.
- Fritsch, R. M., and N. Friesen. 2002. Evolution, domestication and taxonomy, p. 5–30. In: H.D. Rabinowitch and Currah (eds.). Allium Crop Science: Recent Advances. CABI Publishing, Wallingford, U. K.
- Fukino, N., Y. Yoshioka, M. Sugiyama, Y. Sakata, and S. Matsumoto. 2013. Identification and validation of powdery mildew (*Podosphaera xanthii*)-resistant loci in recombinant inbred lines of cucumber (*Cucumis sativus* L.). Mol. Breed. 32: 267– 277.
- Gökçe, A. F. and M. J. Havey. 2002. Linkage equilibrium among tightly linked RFLPs and *Ms* locus in open-pollinated onion populations. J. Amer. Soc. Hort. Sci. 127: 944–946.
- Gökçe, A. F., J. McCallum, Y. Sato, and M. J. Havey. 2002. Molecular tagging of the *Ms* locus in onion. J. Amer. Soc. Hort. Sci. 127: 576–582.
- Haishima, M., J. Kato, and H. Ikehashi. 1993. Isozyme polymorphism in native cultivars of Japanese bunching onion (*Allium fistulosum* L.). Jpn. J. Breed. 43:537–547.
- Hanelt, P. 1990. Taxonomy, evolution, and history. In: Rabinowitch, H. D. and L. L. Brewster (eds.). Onions and Allied Crops. Vol 1. CRC Press, Inc., Boca Raton, Florida, USA. pp 1–26.
- Hang, T.T. M., M. Shigyo, N. Yamauchi, and Y. Tashiro. 2004. Production and characterization of alien chromosome additions in shallot (*Allium cepa* L. Aggregatum group) carrying extra chromosome(s) of Japanese bunching onion (*A*.

fistulosum L.). Genes Genet. Syst. 79: 263–269.

- Havey, M. 1992. Restriction enzyme analysis of the chloroplast and nuclear 45s ribosomal DNA of *Allium* sections *Cepa* and *Phyllodolon*. Plant Syst. Evol. 183: 1731.
- Hirakawa, H., K. Shirasawa, K. Miyatake, T. Nunome, S. Negoro, A. Ohyama, H. Yamaguchi, S. Sato, S. Isobe, S. Tabata, and H. Fukuoka. 2014. Draft Genome Sequence of Eggplant (*Solanum melongena* L.): the Representative Solanum Species Indigenous to the Old World. DNA Res. 21: 649–660.
- Huang, L., S. A. Brooks, W. Li, J. P. Fellers, H. N. Trick, and B. S. Gill. 2003. Map-based cloning of leaf rust resistance gene *Lr21* from the large and polyploid genome of bread wheat. Genetics 164: 655–664.
- Huang, L. and B. S. Gill. 2001. An RGA-like marker detects all known *Lr21* leaf rustresistance gene family members in *Aegilops tauschii* and wheat. Theor. Appl. Genet. 103: 1007–1013.
- Inden, H. and T. Asahina. 1990. Japanese bunching onion (*Allium fistulosum* L.). In Brewster JL and Rabinowitch, H. D. (eds) Onion and allied crops. Vol III. CRC Press, Boca Raton, Florida, pp 159–178.
- Ipek, M., A. Ipek, S. G. Almquist, and P. W. Simon. 2005. Demonstration of linkage and development of the first low-density genetic map of garlic, based on AFLP markers. Theor. Appl. Genet. 110: 228–236.
- Jagathpriya, S. W., J. S. Brian, and A. B. Falk. 2003. Conversion of an amplified fragment length polymorphism marker into a co-dominant marker in the mapping of the *Rph15* gene conferring resistance to barley leaf rust, *Puccinia hordei* Otth. Theor. Appl. Genet. 108: 712–719.
- Jakše, M., M. J. Havey, B. Bohanec. 2003. Chromosome doubling procedures of onion (*Allium cepa* L.) gynogenic embryos. Plant Cell Reports 21: 905–910.
- Jakše, J., W. Martin, J. McCallum, and M. J. Havey. 2005. Single nucleotide polymorphisms, indels, and simple sequence repeats for onion cultivar identification. J. Am. Soc. Hortic. Sci. 130: 912–917.
- Jakše, J., J. D. F. Meyer, G. Suzuki, J. McCallum, F. Cheung, C. D. Town, and M. J. Havey. 2008. Pilot sequencing of onion genomic DNA reveals fragments of

transposable elements, low gene densities, and significant gene enrichment after methyl filtration. Mol. Genet. Genom. 280: 287–292.

- Jennings, D.M., B. V. Ford-Lloyd, and G. M. Butler. 1990a. Rust infections of some *Allium* species: An assessment of germplasm for utilizable rust resistance. Euphytica 49: 99–109.
- Jennings, D.M., B. V. Ford-Lloyd, and G. M. Butler. 1990b. Effect of plant age, leaf position and leaf segment on infection of leek by leek rust, *Puccinia allii*. Plant Pathol. 39: 591–597.
- Jia, G., P. Chen, G. Qin, G. Bai, X. Wang, S. Wang, B. Zhou, S. Zhang, and D. Liu D. 2005. QTLs for Fusarium head blight response in a wheat DH population of Wangshuibai/Alondra's'. Euphytica 146: 183–191.
- Jones, C. J., K. J. Edwards, S. Castaglione, M. O. Winfield, F. Sala, C. van de Wiel, G. Bredemeijer, B. Vosman, M. Matthes, A. Daly, R. Brettschneider, R. Bettini, M. Buiatti, E. Maestri, A. Malcevschi, N. Marmiroli, R. Aert, G. Volckaert, J. Rueda, R. Linacero, A. Vazquez, and A. Karp. 1997. Reproducibility testing of RAPD, AFLP and SSR markers in plants by a network of European laboratories. Mol. Breed. 3: 381–390.
- Jones, R.N., and H. Rees. 1968. Nuclear DNA variation in Allium. Heredity 23: 591-605.
- Kakizaki, T., T. Kato, N. Fukino, M. Ishida, K. Hatakeyama, and S. Matsumoto. 2011. Identification of quantitative trait loci controlling late bolting in Chinese cabbage (*Brassica rapa* L.) parental line Nou 6 gou. Breed. Sci. 159: 151–159.
- Kaneko, Y., T. Natsuaki, S. W. Bang, and Y. Matsuzawa. 1996. Identification and evaluation of Turnip Mosaic Virus (TuMV) resistance gene in kale monosomic addition lines of radish. Breeding Sci. 46: 117–124.
- Khrustaleva, L.I., and C. Kik. 1998. Cytogenetical studies in the bridge cross *Allium cepa* × (*A. fistulosum* × *A. roylei*). Theor. Appl. Genet. 96: 8–14.
- Khrustaleva, L.I., and C. Kik. 2000. Introgression of *Allium fistulosum* into *A. cepa* mediated by *A. roylei*. Theor. Appl. Genet. 100: 17–26.
- Kim, S., M. Kim, Y. Kim, S. Yeom, K. Cheong, K. Kim, J. Jeon, S. Kim, D. Kim, S. Sohn,Y. Lee, and D. Choi. 2015. Integrative structural annotation of *de novo* RNA-Seq

provides an accurate reference gene set of the enormous genome of the onion (*Allium cepa* L.). DNA Res. 22: 19–27.

- Kim, S.Y., B. S. Park, S. J. Kwon, J. Kim, M. H. Lim, Y. D. Park, D. Y. Kim, S. C. Suh, Y. M. Jin, J. H. Ahn, and Y. H. Le. 2007. Delayed flowering time in *Arabidopsis* and *Brassica rapa* by the overexpression of *FLOWERING LOCUS C (FLC)* homologs isolated from Chinese cabbage (*Brassica rapa* L. ssp. *pekinensis*). Plant Cell Rep. 26: 327–336.
- King J.J., J. M. Bradeen, O. Bark, J. A. McCallum, and M. J. Havey. 1998. A low-density map of onion reveals a role for tandem duplication in the evolution of an extremely large diploid genome. Theor. Appl. Genet. 96: 52–62.
- Kitamoto, N., S. Yui, K. Nishikawa, Y. Takahata, and S. Yokoi. 2014. A naturally occurring long insertion in the first intron in the *Brassica rapa* FLC2 gene causes delayed bolting. Euphytica 196: 213–223.
- Kitashiba, H., F. Li, H. Hirakawa, T. Kawanabe, Z. Zou, Y. Hasegawa, K. Tonosaki, S. Shirasawa, A. Fukushima, S. Yokoi, Y. Takahata, T. Kakizaki, M. Ishida, S. Okamoto, K. Sakamoto, K. Shirasawa, S. Tabata, and T. Nishio. 2014. Draft sequences of the radish (*Raphanus sativus* L.) genome.DNA Res. 21 :481–490.
- Kofoet, A., C. Kik, W. A. Wietsma, and J. N. de Vries. 1990. Inheritance of resistance to downy mildew (*Peronospora destructor* [Berk.] Casp.) from *Allium roylei* Stearn in the backcross *Allium cepa* L. × (*A. roylei* × *A. cepa*). Plant Breed. 105: 144–149.
- Kosambi, D. 1944. The estimation of map distance from recombination values. Ann. Eugen. 12: 172–175.
- Kuhl, J. C., F. Cheung, Q. Yuan, W. Martin, Y. Zewdie, J. MaCallum, A. Catanach, P. Rutherford, K. C. Sink, M. Jenderk, J. P. Prince, C. D. Town, and M. J. Havey. 2004.
  A unique set of 11,008 onion expressed sequence tags reveals expressed sequence and genomic differences between the monocot orders *Asparagales* and *Poales*. Plant Cell 16: 114–125.
- Kumar, L.S. 1999. DNA markers in plant improvement: An overview. Biotech. Advan. 17: 143–182.
- Kumazawa, S. and H. Katsumata. 1965. Negi (Japanese bunching onion). p. 280-289. In:

S. Kumazawa (ed.). Sosai-engei Kakuron (Vegetable Crops). Yokendo, Tokyo. [In Japanese].

- Labani, R., and T. Elkington. 1987. Nuclear DNA variation in the genus *Allium* L. (Liliaceae). Heredity 59: 119–128.
- Lander, E., P. Green, J. Abrahamson, A. Barlow, M. Daly, S. Lincoln, and L. Newburg. 1987. MAPMAKER: an interactive computer package for constructing primary genetic linkage maps of experimental and natural populations. Genomics 1: 174– 181.
- Lanzotti, V., A. Romano, S. Lanzuise, G. Bonanomi, and F. Scala. 2012. Antifungal saponins from bulbs of white onion, *Allium cepa* L. Phytochemistry, 74: 133–139.
- Lee, R., S. Baldwin, F. Kenel, J. McCallum, and R. Macknight. 2013. *FLOWERING LOCUS T* genes control onion bulb formation and flowering. Nat. Comm. 4: 2884.
- Li, Q. Q., S. D. Zhou, X. J. He, Y. Yu, Y. C. Zhang, and X. Q. Wei. 2010. Phylogeny and biogeography of *Allium* (Amaryllidaceae: Allieae) based on nuclear ribosomal internal transcribed spacer and chloroplast rps16 sequences, focusing on the inclusion of species endemic to China. Ann. Bot. 106: 709–733.
- Lin, M. W. and W. L. Chang. 1980. Interspecific hybridization in genus *Allium*. I. Effect of different temperatures on bolting of Japanese bunching onion (*Allium fistulosum* L.). Chinese Hortic. 26: 173 [in Chinese with English summary].
- Mago, R., W. Spielmeyer, J. Lawrence, S. Lagudah, G. Ellis and A. Pryor. 2002. Identification and mapping of molecular markers linked to rust resistance gene located on chromosome 1RS of rye using wheat-rye translocation lines. Theor. Appl. Genet. 104: 1317–1324.
- Martin, W. J., J. McCallum, M. Shigyo, J. Jakše, J. C. Kuhl, N. Yamane, M. Pither-Joyce, A. F. Gokce, K. C. Sink, C. D. Town, and M. J. Havey. 2005. Genetic mapping of expressed sequences in onion and in silico comparisons with rice show scant colinearity. Mol. Genet. Genomics 274: 197–204.
- Masamura, N., J. McCallum, L. Khrustaleva, F. Kenel, M. Pither-Joyce, J. Shono, G. Suzuki, Y. Mukai, N. Yamauchi, and M. Shigyo. 2012. Chromosomal organization and sequence diversity of genes encoding lachrymatory factor synthase in *Allium*

cepa L. Genes Genomes Genetics 2: 643-651.

- Masamura, N., S. Yaguchi, Y. Ono, T. Nakajima, S. Masuzaki, S. Imai, N. Yamauchi, and M. Shigyo. 2011. Characterization of Amino Acid and S-alk(en)yl-L-cysteine Sulfoxide Production in Japanese Bunching Onion Carrying an Extra Chromosome of Shallot. J. Japan Soc. Horti. Sci. 80: 322–333.
- Masuzaki, S., N. Araki, N. Yamauchi, N. Yamane, T. Wako, A. Kojima, and M. Shigyo. 2006a. Chromosomal locations of microsatellites in onion. Hort. Science 41: 315– 318.
- Masuzaki, S., M. Shigyo, and N. Yamauchi. 2006b. Complete assignment of structural genes involved flavonoid biosynthesis influencing bulb color to individual chromosomes of the shallot (*Allium cepa* L.). Genes Genet. Syst. 81: 255–263.
- Masuzaki, S., M. Shigyo, and N. Yamauchi. 2006c. Direct comparison between genomic constitution and flavonoid contents in *Allium* multiple alien addition lines reveals chromosomal locations of genes related to biosynthesis from dihydrokaempferol to quercetin glucosides in scaly leaf of shallot (*A. cepa* L). Theor. Appl. Genet. 112: 607–617.
- McCallum, J., S. Baldwin, M. Shigyo, Y. Deng, S. van Heusden, M. Pither-Joyce, and F. Kenel. 2012. AlliumMap-A comparative genomics resource for cultivated *Allium* vegetables. BMC Genomics 13: 168.
- McCallum, J., A. Clarke, M. Pither-Joyce, M. Shaw, R. Butler, D. Brash, J. Scheffer, I. Sims, S. van Heusden, M. Shigyo, and M. J. Havey. 2006. Genetic mapping of a major gene affecting onion bulb fructan content. Theor. Appl. Genet. 112: 958–967.
- McCallum, J., M. Pither-Joyce, M. Shaw, F. Kanel, S. Davis, R. Butler, J. Scheffer, J. Jakse, and M. J. Havey. 2007. Genetic mapping of sulfur assimilation genes a QTL for bulb onion pungency. Theor. Appl. Genet. 114: 815–822.
- Mesbah, M., O. E. Scholten, Th. S. M. de Bock, and W. Lange. 1997. Chromosome location of genes for resistance to *Heterodera schachtii*, *Cercospora beticola* and *Polymyxa betae* using sets of *Beta procumbens* and *B. patellaris* derived monosomic additions in *B. vulgaris*. Euphytica 97: 117–127.

Ministry of Agriculture, Forestry and Fisheries, Japan. 2014. The ranking and

composition ratio of major crops for annual output in Japan 2012. http://www.e-stat.go.jp/SG1/estat/List.do?lid=000001117298. [In Japanese].

- Morinaka, T. 1985. Parasitism of rust fungi of welsh onion and Chinese chive to Allium species. Proceedings of the Kanto – Tosan Plant Protection Society 32: 115–116. [In Japanese].
- Moue, T. and T. Uehara. 1985. Inheritance of cytoplasmic male sterility in *Allium fistulosum* L. (welsh onion). J. Jpn. Soc. Hortic. Sci. 53:432–437. [in Japanese].
- Ohara, T., Y. S. Song, H. Tsukazaki, T. Wako, T. Nunome, and A. Kojima. 2005a. Construction of a genetic linkage map of Japanese bunching onion (*Allium fistulosum*) based on AFLP and SSR markers. Euphytica 144: 255–263.
- Ohara, T., H. Tsukazaki, Y. S. Song, T. Wako, K. Yamashita, and A. Kojima. 2009. Mapping of Quantitative Trait Loci Controlling Seedling Growth in Bunching Onion (*Allium fistulosum* L.). J. Japan. Soc. Hort. Sci. 78: 436–442.
- Ohara, T., T. Wako and A. Kojima. 2004. Diallel analysis of seedling growth in Japanese bunching onion (*Allium fistulosum* L.). Breed. Res. 6: 11–19. [In Japanese with English abstract].
- Ohara, T., T. Wako, T. Nunome and A. Kojima. 2005b. Relationship between heterosis and genetic distances as revealed by AFLP markers in intervarietal F<sub>1</sub> hybrids of Japanese bunching onion (*Allium fistulosum*). J. Japan. Soc. Hort. Sci. 74: 68–77. [In Japanese with English abstract].
- Ohsawa, R. 2004. Use of molecular markers in vegetable breeding. Hort. Res. 3: 1–6. [in Japanese].
- Palloix, A., E. Pochard, T. Phaly and A. M. Daubeze. 1990. Recurrent selection for resistance to *Verticillium dahliae* in pepper. Euphytica 47: 79–89.
- Park, J., H. Bang, D. Y. Cho, M.K. Yoon, B. S. Patil, and S. Kim. 2013. Construction of high-resolution linkage map of the *Ms* locus, a restorer-of-fertility gene in onion (*Allium cepa* L.). Euphytica 192: 267–278.
- Paul, H., B. Henken, Th. S. M. de Bock, and W. Lange. 1992. Resistance to *Polymyxa* betae in Beta species of the section *Procumbentes*, in hybrids with *B. vulgaris* and in monosomic chromosome additions of *B. procumbens* in *B. vulgaris*. Plant

Breeding 109: 265-273.

- Peffley, E.B. 1986. Evidence for chromosome differentiation of *Allium fistulosum* and *A. cepa*. J. Am. Soc. Hort. Sci. 111: 126–129.
- Peffley, E.B. and A. Hou. 2000. Bulb-type onion introgressants possessing *Allium fistulosum* L. genes recovered from interspecific hybrid backcrosses between *A. cepa* L. and *A. fistulosum* L. Theor. Appl. Genet. 100: 528–534.
- Peterka, H., H. Budahn, O. Schrader, R. Ahne, and W. Schütze, 2004. Transfer of resistance against the beet cyst nematode from radish (*Raphanus sativus*) to rape (*Brassica napus*) by monosomic chromosome addition. Theor. Appl. Genet. 109: 30–41.
- Raupp, W. J., L. E. Browder and B. S. Gill. 1983. Leaf rust resistance in *Aegilops squarrosa*: its transfer and expression in common wheat (*Triticum aestivum* L.). Phytopathology 73: 818.
- Rosewarne, G. M., S. A. Herrera-Foessel, R. P. Singh, J. Huerta-Espino, C. X. Lan, and Z. H. He. 2013. Quantitative trait loci of stripe rust resistance in wheat. Theor. Appl. Genet. 126: 2427–2449.
- Savitsky, H. 1978. Nematode (*Heterodera schachtii*) resistance and meiosis in diploid plants from interspecific *Beta vulgaris* x *B. procumbens* hybrids. Can. J. Genet. Cytol. 20: 177–186.
- Scholten, O. E., A. W. van Heusden, L. I. Khrustaleva, K. Burger-Meijer, R. A. Mank, R.
  G. C. Antonise, J. L. Harrewijn, W. Van haecke, E. H. Oost, R. J. Peters, C. Kik.
  2007. The long and winding road leading to the successful introgression of downy mildew resistance into onion. Euphytica 156: 345–353.
- Schranz, M. E., P. Quijada, S. B. Sung, L. Lukens, R. Amasino, and T. C. Osborn. 2002. Characterization and effects of the replicated flowering time gene *FLC* in *Brassica rapa*. Genetics 162: 1457–1468.
- Shaner, G. and R. E. Finney. 1977. The effect of nitrogen fertilization on the expression of slow mildewing resistance in knox wheat. Phytopathology 67: 1051–1056.
- Shigyo, M., M. Iino, H. Ino, and Y. Tashiro. 1999a. Transmission rates of extra chromosomes in alien monosomic addition lines of Japanese bunching onion with

extra chromosomes from shallot. J. Japan Soc. Hort. Sci. 68: 18-22.

- Shigyo, M., M. Iino, and Y. Tashiro. 1999b. Fertility of alien monosomic addition lines of Japanese bunching onion (*Allium fistulosum* L.) with extra chromosomes from shallot (*A. cepa* L. Aggregatum group). J. Japan Soc. Hort. Sci. 68: 494–498.
- Shigyo, M. and C. Kik. (2008) Onion. In Prohens, J. and F. Nuez (eds.) Vegetables: handbook of plant breeding. Vol. 2, Springer Verlag, Berlin, p.121–162.
- Shigyo, M., Y. Tashiro, M. Iino, N. Terahara, K. Ishimaru, and S. Isshiki. 1997. Chromosomal locations of genes related to flavonoid and anthocyanin production in leaf sheath of shallot (*Allium cepa* L. Aggregatum group). Genes Genet. Syst. 72: 149–152.
- Shigyo, M., Y. Tashiro, S. Isshiki, and S. Miyazaki. 1995. Chromosomal locations of five isozyme gene loci (*Lap-1, Got-1, 6-Pgdh-2, Adh-1 and Gdh-1*) in shallot (*Allium cepa* L. Aggregatum group). Jpn. J. Genet. 70: 399–407.
- Shigyo, M., Y. Tashiro, S. Isshiki, and S. Miyazaki. 1996. Establishment of a series of alien monosomic addition lines of Japanese bunching onion (*Allium fistulosum* L.) with extra chromosomes from shallot (*A. cepa* L. Aggregatum group). Genes Genet. Syst. 71: 363–371.
- Shigyo, M., Y. Tashiro, and S. Miyazaki. 1994. Chromosomal locations of glutamate oxaloacetate transaminase gene loci in Japanese bunching onion (*Allium fistulosum* L.) and shallot (*A. cepa* L. Aggregatum group). Jpn. J. Genet. 69: 417–424.
- Shigyo, M., T. Wako, A. Kojima, N. Yamauchi, and Y. Tashiro. 2003. Transmission of alien chromosomes from selfed progenies of a complete set of *Allium* monosomic additions: the development of a reliable method for the maintenance of a monosomic addition set. Genome 46: 1098–1103.
- Shiraiwa, N., K. Kikuchi, I. Honda, M. Shigyo, H. Yamazaki, D. Tanaka, K. Tanabe, and A. Itai. 2011. Characterization of endogenous gibberellins and molecular cloning of a putative gibberellin 3-oxidase gene in bunching onion. J. Amer. Soc. Hort. Sci. 136: 382–388.
- Singh, R.P., J. Huerta-Espino, S. Bhavani, S. A. Herrera-Foessel, D. Singh, P. K. Singh,G. Velu, R. E. Mason, Y. Jin, P. Njau, and J. Crossa. 2011. Race-non-specific

resistance to rust diseases in CIMMYT spring wheats. Euphytica 179: 175–186.

- Song, J., A. Angel, M. Howard, and C. Dean. 2012. Vernalization a cold-induced epigenetic switch. J. Cell Sci. 125: 3723–3731.
- Song, Y. S., K. Suwabe, T. Wako, T. Ohara, T. Nunome, and A. Kojima. 2004. Development of microsatellite markers in Japanese bunching onion (*Allium fistulosum* L.). Breed. Sci. 54: 361–365.
- Stein, N., C. Feuillet, T. Wicker, E. Schlagenhauf, and B. Keller. 2000. Subgenome chromosome walking in wheat: A 450-kb physical contig in *Triticum monococcum*L. spans the *Lr10* resistance locus in hexaploid wheat (*Triticum aestivum* L.). Proc. Natl. Acad. Sci. USA 97: 13436–13441.
- Sulistyaningsih, E., Y. Aoyagi, and Y. Tashiro. 2006. Flower bud culture of shallot (*Allium cepa* L. Aggregatum group) with cytogenetic analysis of resulting gynogenic plants and somaclones. Plant Cell Tiss. Organ Cult. 86: 249–255.
- Sun, X. D., X. H. Yu, S. M. Zhou, and S. Q. Liu. 2016. De novo assembly and characterization of the Welsh onion (*Allium fistulosum* L.) transcriptome using Illumina technology. Mol. Genet. Genomics 291: 647–659.
- Takeuchi, T. 1990. Forecasting of occurrence of rust of welsh onion. Bull. Chiba Agric. Exp. Stn. 31: 73–84. [In Japanese].
- Tanksley, S. D. 1993. Mapping polygenes. Annu. Rev. Genet. 27: 205–233.
- The Angiosperm Phylogeny Group. 2009. An update of the Angiosperm Phylogeny Group classification for the orders and families of flowering plants: APG III. Botanical Journal of the Linnean Society 161: 105–121.
- The Tomato Genome Consortium. 2012. The tomato genome sequence provides insights into fleshy fruit evolution. Nature. 485: 635–641.
- Tsukazaki, H., H. Fukuoka, Y. S. Song, K. Yamashita, T. Wako, and A. Kojima. 2006. Considerable heterogeneity in commercial F1 varieties of bunching onion (*Allium fistulosum*) and proposal of breeding scheme for conferring variety traceability using SSR markers. Breed. Sci. 56: 321–326.
- Tsukazaki, H., T. Nunome, H. Fukuoka, H. Kanamori, I. Kono, K. Yamashita, T. Wako, and A. Kojima. 2007. Isolation of 1,796 SSR clones from SSR-enriched DNA

libraries of bunching onion (Allium fistulosum). Euphytica 157: 83-94.

- Tsukazaki, H., S. Yaguchi, S. Sato, H. Hirakawa, Y. Katayose, H. Kanamori, K. Kurita, T. Itoh, M. Kumagai, S. Mizuno, M. Hamada, H. Fukuoka, K. Yamashita, J. A. McCallum, M. Shigyo, and T. Wako. 2015. Development of transcriptome shotgun assembly-derived markers in bunching onion (*Allium fistulosum*). Mol. Breed. 35: 55.
- Tsukazaki, H., S. Yaguchi, K. Yamashita, M. Shigyo, A. Kojima, and T. Wako. 2012. QTL analysis for pseudostem pungency in bunching onion (*Allium fistulosum*). Mol. Breed. 30: 1689–1698.
- Tsukazaki, H., K. Yamashita, S. Yaguchi, S. Masuzaki, H. Fukuoka, J. Yonemaru, H. Kanamori, I. Kono, H. Tran Thi Minh, M. Shigyo, A. Kojima, and T. Wako. 2008. Construction of SSR-based chromosome map in bunching onion (*Allium fistulosum*). Theor. Appl. Genet. 117: 1213–1223.
- Tsukazaki, H., K. Yamashita, S. Yaguchi, K. Yamashita, T. Hagihara, M. Shigyo, A. Kojima, and T. Wako. 2011. Direct determination of the chromosomal location of bunching onion and bulb onion markers using bunching onion-shallot monosomic additions and allotriploid-bunching onion single alien deletions. Theor. Appl. Genet. 122: 501–510.
- Ulloa, M., J. N. Corgan, and M. Dunford. 1995. Evidence for nuclear-cytoplasmic incompatibility between *Allium fistulosum* and *A. cepa*. Theor. Appl. Genet. 90: 746–754.
- van der Meer Q.P. and J. N. de Vries. 1990. An interspecific cross between *Allium roylei* Stearn and *Allium cepa* L., and its backcross to *A. cepa*. Euphytica 47: 29–31.
- van Heusden A.W., J. W. van Ooijen, R. Vrielink-van Ginkel, W. H. J. Verbeek, W. A. Wietsma, and C. Kik. 2000a. A genetic map of an interspecific cross in Allium based on amplified fragment length polymorphisms (AFLP<sup>TM</sup>) markers. Theor. Appl. Genet. 100: 118–126.
- van Heusden, A.W., M. Shigyo, Y. Tashiro, R. Vrielink-van Glinkel, and C. Kik. 2000b. AFLP linkage group assignment to the chromosomes of *Allium cepa* L. *via* monosomic addition lines. Theor. Appl. Genet. 100: 480–486.

- van Ooijen, J. W. 2006. JoinMap 4.0, Software for the calculation of genetic linkage map in experimental population. Kyazma BV, Wageningen, Netherlands.
- van Raamsdonk, L. W. D., W. Ensink, A. W. van Heusden, M. Vrielink-van-Ginkel, and C. Kik. 2003. Biodiversity assessment based on cpDNA and crossability analysis in selected species of *Allium* subgenus Rhizirideum. Theor. Appl. Genet. 107: 1048– 1058.
- van Raamsdonk, L. W. D., M. Vrielink-van-Ginkel M, and C. Kik. 2000. Phylogeny reconstruction and hybrid analysis in *Allium* subgenus Rhizirideum. Theor. Appl. Genet. 100: 1000–1009.
- Voorrips, R. E., M. C. Jongerius, and H. J. Kanne. 1997. Mapping of two genes for resistance to clubroot (*Plasmodiophora brassicae*) in a population of doubled haploid lines of *Brassica oleracea* by means of RFLP and AFLP markers. Theor. Appl. Genet. 94: 75–82.
- Vu, H.Q., M.A. El-Sayed, S. Ito, N. Yamauchi, and M. Shigyo. 2012. Discovery of a new source of resistance to *Fusarium oxysporum*, cause of *Fusarium* wilt in *Allium fistulosum*, located on chromosome 2 of *Allium cepa* Aggregatum group. Genome 55: 797–807.
- Wako, T., A. Kojima, K. Yamashita, H. Tsukazaki, T. Ohara, and Y. Sakata. 2010. Breeding and characteristics of a bunching onion (*Allium fistulosum* L.) cultivar 'Fuyuwarabe' with shortleaf sheath. Hort. Res. (Jpn.) 9: 273–285. [in Japanese with English summary].
- Wako, T., H. Tsukazaki, S. Yaguchi, K. Yamashita, S. Ito, and M. Shigyo. 2016. Mapping of quantitative trait loci for bolting time in bunching onion (*Allium fistulosum* L.). Euphytica 209:537–546.
- Wako, T., K. Yamashita, H. Tsukazaki, T. Ohara, A. Kojima, and Y. Noguchi. 2012.
  Development of 'Negi Chuukanbohon Nou 1' a bunching onion (*Allium fistulosum* L.) parental line with rust resistance. Bull. Natl. Inst. Veg. & Tea Sci. 11: 55–62. [In Japanese].
- Wako, T., K. Yamashita, H. Tsukazaki, T. Ohara, A. Kojima, S. Yaguchi, S. Shimazaki, N. Midorikawa, T. Sakai, N. Yamauchi, and M. Shigyo. 2015. Screening and

incorporation of rust resistance from *Allium cepa* into bunching onion (*Allium fistulosum*) via alien chromosome addition. Genome 58: 135–142.

- Wang, S., C. J. Basten, and Z. B. Zeng. 2007. Windows QTL Cartographer 2.5. Department of Statistics, North Carolina State University, Raleigh, NC. http://statgen.ncsu.edu/qtlcart/WQTLCart.htm.
- Wehner, T. C. and C. S. Cramer. 1996. Ten cycles of recurrent selection for fruit yield, earliness, and quality in three slicing cucumber populations. J. Amer. Soc. Hort. Sci. 121: 362–366.
- Wilkie, S. E., P. G. Isaac, and R. J. Slater. 1993. Random amplified polymorphic DNA (RAPD) markers for genetic analysis in *Allium* Theor. Appl. Genet. 86: 497–504.
- Yaguchi S., T. T. M. Hang, H. Tsukazaki, Q. H. Vu, S. Masuzaki, T. Wako, N. Masamura,
  S. Onodera, N. Shiomi, N. Yamauchi, and M. Shigyo. 2009. Molecular and
  biochemical identification of alien chromosome additions in shallot (*Allium cepa* L.
  Aggregatum group) carrying extra chromosome(s) of bunching onion (*A. fistulosum*L.). Genes Genet. Syst. 84: 43–55.
- Yaguchi, S., J. McCallum, M. Shaw, M. Pither-Joyce, S. Onodera, N. Shimoi, N. Yamauchi, and M. Shigyo. 2008. Chromosomal localization and mapping of genes affecting carbohydrate accumulation in *Allium cepa* L. Plant Cell Physiol. 49: 730–739.
- Yamasaki, A., H. Miura, and K. Tanaka. 2000a. Effect of photoperiods before, during and after vernalization on flower initiation and development and its varietal difference in Japanese bunching onion (*Allium fistulosum* L.). J. Hort. Sci. Biotechnol. 75: 645– 650.
- Yamasaki, A., K. Tanaka, and H. Inden. 2012. Seed vernalization-like response in Japanese bunching onion (*Allium fistulosum* L.). Acta. Hort. 969: 133–134.
- Yamasaki, A., K. Tanaka, M. Yoshida, and H. Miura. 2000b. Effects of day and night temperatures on flower-bud formation and bolting of Japanese bunching onion (*Allium fistulosum* L.). J. Jpn. Soc. Hortic. Sci. 69: 40–46.
- Yamashita, K., H. Tsukazaki, A. Kojima, T. Ohara, and T. Wako. 2010. Inheritance mode of male sterility in bunching onion (*Allium fistulosum* L.) accessions. Euphytica

173:357-367.

- Yamashita, K., T. Wako, T. Ohara, H. Tsukazaki, and A. Kojima. 2005. Improvement of rust resistance in bunching onion (*Allium fistulosum* L.) by recurrent selection. J. Japan Soc. Horti. Sci. 74: 444–450.
- Yang, Y. Y., Y. M. Huo, J. Miao, B. J. Liu, S. P. Kong, L. M. Gao, C. Liu, Z. B. Wang, Y. Tahara, H. Kitano, and X. Wu. 2013. Identification of two SCAR markers co-segregated with the dominant *Ms* and recessive *ms* alleles in onion (*Allium cepa* L.). Euphytica 190: 267–277.
- Yoshioka, Y., Y. Sakata, M. Sugiyama, and N. Fukino. 2014. Identification of quantitative trait loci for downy mildew resistance in cucumber (*Cucumis sativus* L.). Euphytica 198: 265–276.
- Young, N. D. 1999. A cautiously optimistic vision for marker-assisted breeding. Mol. Breed. 5: 505–510.

## **SUMMARY**

In Japan, bunching onion (*Allium fistulosum* L.) is an indispensable vegetable crop for its economic value, familiarity in cuisines, and health benefits. Little genetic information is available about disease resistance and late bolting, which are essential traits in bunting onion cultivars. Rust caused by *Puccinia allii* Rudolphi is a serious foliar disease of bunching onion. To improve rust resistance of bunching onion, it is necessary to identify genetic resources with high resistance and understand the mode of inheritance of resistance genes. Late bolting is also one of the most important agronomic traits for year-round production in bunching onion because it affects the yield and quality of the harvested products during the spring and early summer. However, genetic and molecular studies of bolting time in bunching onion have not been reported thus far. The combined use of doubled haploid (DH) lines and molecular markers in *Allium* genetic studies can provide essential information for breeding programs.

The present studies were conducted to reveal such useful genetic resources and the genetic basis of rust resistance and late bolting in bunching onion. Further, informative molecular markers were developed and a linkage map was constructed using a DH population of *A. cepa* to improve selection methods for *Allium* vegetable breeding.

## (1) Screening and incorporation of rust resistance from *Allium cepa* into bunching onion (*Allium fistulosum*) via alien chromosome addition

Bunching onion (*A. fistulosum* L.; 2n = 16; genomes FF), bulb onion (*A. cepa* L. Common onion group), and shallot (*A. cepa* L. Aggregatum group; genomes AA) cultivars were inoculated with rust fungus *Puccinia allii* isolated from bunching onion. Bulb onions and shallots are highly resistant to rust, suggesting that they would serve as useful resources for breeding rust resistant bunching onions. To identify the *A. cepa* chromosome(s) carrying rust resistance, a complete set of eight *A. fistulosum* – shallot monosomic alien addition lines (MAALs) were inoculated with *P. allii*. At the seedling stage, FF+1A showed a high level of resistance in controlled-environment experiments, suggesting that the genes related to rust resistance could be located on shallot chromosome 1A. However, no adult plants from the MAALs, multi-chromosome addition

lines, or hypoallotriploids exhibited strong resistance comparable to shallot. But the addition line FF+1A+5A showed reproducibly high levels of rust resistance.

## (2) Mapping of quantitative trait loci for rust resistance in bunching onion

Disease severity was evaluated in an F<sub>2:3</sub> population derived from crosses between a resistant line Sa03 and a susceptible line T03, and quantitative trait loci (QTL) analysis was conducted using the bunching onion genetic linkage map. In two inoculation trials in the greenhouse, three QTLs related to rust resistance were detected on the linkage groups 1a on chromosome 1 (Chr. 1a), 3a on chromosome 3 (Chr. 3a), and 8a on chromosome 8 (Chr. 8a). Although these QTLs were detected in some trials but not others, they were validated using a population derived from residual heterozygous lines. Variances in rust resistance explained by those QTLs were relatively low, suggesting that other loci are involved in rust resistance in line Sa03.

## (3) Mapping of quantitative trait loci for bolting time in bunching onion

The bolting times of two F<sub>2:3</sub> populations derived from crosses between cultivars differing in bolting time were evaluated and QTL analysis was conducted. When the KiC population, which was derived from a cross between the ever-flowering line Ki and the late-bolting line C, was grown under field conditions, two QTLs associated with bolting time were repeatedly detected on the linkage groups 1a on chromosome 1 (Chr. 1a) and 2a on chromosome 2 (Chr. 2a). However, the QTL on Chr. 1a was not detected when the KiC population was grown in a heated greenhouse under unvernalized conditions. A single QTL with major effect was identified exclusively on the linkage group Chr. 2a in the SaT03 population derived from a cross between early-bolting line Sa03 and late-bolting T03 evaluated under field conditions. QTL located on Chr. 2a in both populations were linked to the same marker loci, suggesting that these regions were strongly related. Simple sequence repeat loci linked to these QTLs had significant effects on bolting time in both populations.

## (4) Construction of an Allium cepa linkage map using doubled haploid technology

A gynogenic mapping population for constructing the linkage map of *A. cepa* was produced from F<sub>1</sub> plants of a cross between DH shallot and bulb onion lines. This population of 100 plants included 46 haploids (2n = 8), 40 diploids (2n = 16), and 14

near-polyploids that were obtained from 10,604 unpollinated flower cultures. All of these were confirmed to have been derived from female gametes by analyzing co-dominant markers. Using this population, we constructed a map consisting of 12 linkage groups with 108 markers covering 799 cM. Chromosome assignments of 12 linkage groups were identified using MAALs. Bulb pigmentation, which was used as a phenotypic marker, was closely linked to the DFR gene of on chromosome 7. Five molecular markers were located in close proximity to *Ms*, the fertility restoration locus for cytoplasmic male sterility (CMS) on chromosome 2.
#### 摘要

ネギ (Allium fistulosum L.) は日本人の食によく親しまれ,健康増進に役立ち, わが国における生産額も多い不可欠な野菜である.しかし,品種育成を行う上で 有益な病害抵抗性や晩抽性に関する遺伝学的知見の蓄積はなされておらず,こ れらの形質に着目した育種素材や選抜手法に関する研究開発に支障をきたして いる.本研究では,ネギの重要病害であるさび病に抵抗性をもつ品種育成を進展 させるため,高度抵抗性を有する遺伝資源の同定とそれらの抵抗性の遺伝様式 の解明に取り組んだ.また,晩抽性は春~初夏の生産における収量と品質の安定 化に関係する農業形質であり,ネギ抽苔メカニズムの分子遺伝学的解明の端緒 として同形質に関与する遺伝領域の解明を試みた.一方で,ネギ属の遺伝学的研 究において倍加半数体 (DH)系統と分子マーカーを併用することは,育種計画 に有益な知見を与える.そこで,ネギ属野菜の育種における効率的な選抜法開発 に資するため, Allium cepa L.のDH集団を用いて分子マーカーおよび連鎖地図の 構築を行った.

# (1) Allium cepaのさび病抵抗性系統の検索と異種染色体添加による抵抗性のネ ギへの導入

ネギ (2n=16, ゲノム構成FF) から分離したさび病菌 Puccinia allii をネギ, タ マネギ (A. cepa Common onion group) およびシャロット (A. cepa Aggregatum group, AA) に接種したところ, タマネギとシャロットは強い抵抗性を示し, ネギのさ び病抵抗性育種に有用な遺伝資源であることが示唆された. 抵抗性に関与する A. cepa の染色体を明らかにするために, シャロットの8本の染色体を1本ずつ添 加した8種類のネギ単一異種染色体添加系統 (2n=2x+1=17, FF+1A-FF+8A, 以 下添加系統) にさび病菌を接種した. 環境制御装置を用いた幼苗期の接種試験で は, FF+1Aが強度抵抗性を示し, シャロットの第1染色体に抵抗性に関与する遺 伝子が存在する可能性が示唆された. 一方で, 単一染色体添加系統, 複数染色体 添加系統および低三倍体の成植物を用いた接種試験では, シャロットに匹敵す る強い抵抗性は発揮されなかったが, FF+1A+5Aにおいて比較的強い抵抗性が再 現良く認められた.

## (2) ネギのさび病抵抗性のQTL解析

さび病抵抗性系統 'Sa03' と罹病性系統 'T03' との交雑F<sub>2:3</sub>集団におけるさび 病発病程度を調査し,同集団を用いて作成されたネギ連鎖地図を用いて量的遺 伝子座の特定 (QTL解析) を行った.ビニールハウスを用いた2回の接種検定に おいて,さび病抵抗性に関するQTLが第1,3,8染色体にそれぞれ対応する連鎖 群Chr.1a, Chr.3a, Chr.8aに検出された.検定により異なるQTLが検出されたもの の,それぞれのQTLに近傍のマーカーの遺伝子型によりそれらの抵抗性に及ぼ す効果が立証された.これらのQTLにより説明される分散は比較的小さいこと から, 'Sa03' には抵抗性に関与する他の遺伝子座も存在すると考えられた.

#### (3) ネギの抽苔性に関するQTL解析

抽苔期の異なるネギ系統間交雑によるF2:3集団について抽苔期を調査し,QTL 解析を行った.台湾から導入した不時抽苔性を有する系統'Ki'と晩抽性系統'C' との交雑による KiC集団について,露地圃場で越冬栽培後の抽苔期を調査した ところ,2個のQTLが第1および第2染色体にそれぞれ対応する連鎖群Chr.1aおよ びChr.2aに再現性よく検出された.春化を抑制させるために,同集団を最低気温 10°C以上の加温温室内で栽培したところ,連鎖群Chr.1a上のQTLは検出されなか った.一方で,早期抽苔性系統の'SaO3'と晩抽性系統'TO3'の交雑による SaTO3 集団について露地越冬栽培で調査した結果,連鎖群Chr.2a上に単一の主要QTLが 認められた.両集団の連鎖群Chr.2a上のQTLは同じマーカー座上に位置すること から,両者は密接に関連していると考えられた.これらのQTLに連鎖したSSRマ ーカーは、両集団において抽苔性に関して有意な効果を示した.

## (4) 倍加半数体技術を用いた Allium cepa 連鎖地図の構築

病害抵抗性の程度や抽苔期が異なるシャロットとタマネギのDH系統を交雑 したFi植物から雌性発生分離集団を育成し、これらを用いて A. cepa の連鎖地図 を作成した. 10,604個の未受粉小花を培養して再生した100個体の植物体 [うち 半数体 (2n=8) 46個体,正二倍体 (2n=16) 40個体,高次倍数体14個体]を用 い、12連鎖群,108マーカーからなる全長799cMの連鎖地図が構築された. さら に添加系統を用いて12連鎖群を8本の染色体に振り分けた. 鱗茎部の着色性に関 する形態マーカーは,第7染色体連鎖地図のDFR遺伝子と密接に連鎖していた. また,第2染色体連鎖地図上に座乗する細胞質雄性不稔性(CMS)の稔性回復を 支配する *Ms* 遺伝子座の近傍に5個の分子マーカーを集中マッピングした.

本研究の結果,ネギのさび病抵抗性および晩抽性に関して有効な育種素材を 明らかにするとともに,これらの形質に連鎖するDNAマーカーが開発されたこ とから,生産者や消費者のニーズに応じた品種育成において効率的な選抜に利 用することが可能になる.また,ネギと共通のDNAマーカーをもつ *A. cepa* 連 鎖地図が作成されたことにより,今後,両種の比較ゲノム研究を進展させ,ネギ 属野菜における有用形質の遺伝系解明に向けた取り組みをより一層加速させる 基盤を構築することができた.

## List of papers related to the thesis

Screening and incorporation of rust resistance from *Allium cepa* into bunching onion (*Allium fistulosum*) via alien chromosome addition

Tadayuki Wako, Ken-ichiro Yamashita, Hikaru Tsukazaki, Takayoshi Ohara, Akio Kojima, Shigenori Yaguchi, Satoshi Shimazaki, Naoko Midorikawa, Takako Sakai, Naoki Yamauchi, and Masayoshi Shigyo

Genome 58(4): 135–142. 2015

(In relation to Chapter II)

Mapping of quantitative trait loci for bolting time in bunching onion (*Allium fistulosum* L.)

Tadayuki Wako, Hikaru Tsukazaki, Shigenori Yaguchi, Ken-ichiro Yamashita, Shin-ichi Ito, and Masayoshi Shigyo

Euphytica 209(2): 537-546. 2016

(In relation to Chapter IV)